Obstetrical interventions during labour and birth: an examination of effects on breastfeeding, neonatal mortality and children's educational outcomes

Engida Yisma Derbie

BSc, MSc



Submitted in fulfilment of the requirements for the degree of Doctor of

Philosophy

School of Public Health

Faculty of Health and Medical Sciences

The University of Adelaide

Australia

December, 2020

Declaration

I certify that this work contains no material which has been accepted for the award of any other degree or diploma in my name, in any university or other tertiary institution and, to the best of my knowledge and belief, contains no material previously published or written by another person, except where due reference has been made in the text. In addition, I certify that no part of this work will, in the future, be used in a submission in my name, for any other degree or diploma in any university or other tertiary institution without the prior approval of the University of Adelaide and where applicable, any partner institution responsible for the joint-award of this degree.

I acknowledge that copyright of published works contained within this thesis resides with the copyright holder(s) of those works.

I also give permission for the digital version of my thesis to be made available on the web, via the University's digital research repository, the Library Search and also through web search engines, unless permission has been granted by the University to restrict access for a period of time.

I acknowledge the support I have received for my research through the provision of an Australian Government Research Training Program Scholarship.

Signed:

Engida Yisma Derbie (Candidate)

Table of contents

Declaration	i
Table of contents	ii
List of tables	iv
List of figures	v
List of abbreviations and acronyms	vi
Abstract	viii
Acknowledgements	xiii
Dedication	xv
Publications contributing to this thesis	xvi
Presentations arising out of this thesis	xvii
Media coverage of findings arising from this thesis	xviii
Chapter 1: Introduction	1
1.1 Background	1
1.2 Thesis aim and research questions	8
1.3 Thesis outline	9
Chapter 2: Literature review	11
2.1 Induction of labour	12
2.2 Caesarean section	21
2.3 Apgar scores	
2.4 Research justification	42
Chapter 3: Methodology	44
3.1 Data sources	44
3.2 Exposures: - measures of obstetrical interventions during labour and birth	49
3.3 Outcomes	51
3.4 Confounding	54
3.5 Statistical Analysis	55
Chapter 4: Caesarean section in Ethiopia: prevalence and sociodemographic charac	cteristics 61
4.1 Preface	61
4.2 Publication	
Chapter 5: The changing temporal association between cases room birth and poonate	al dooth in
Ethiopia: socondary analysis of nationally representative surveys	
5 1 Draface	
5.2 Publication	
Chapter 6: Impact of caesarean section on breastfeeding indicators: within-country	and meta-
analyses of nationally representative data from 33 countries in sub-Saharan Africa	
6.1 Preface	87
6.2 Publication	89
Chapter 7: Associations between Apgar scores and children's educational outcome	s at eight
years of age	103
7.1 Preface	103

7.2 Publication	105
Chapter 8: Elective labour induction versus expectant management in women and educational outcomes at eight years of age	children's 114 114
8.2 Publication	115
Chapter 9: Discussion and conclusions 9.1 Key findings and contributions	141 141
9.2 Strengths and limitations	147
9.3 Implications	150
9.4 Concluding remarks	152
Appendices Appendix A: Online Supplementary Material for Study 1	153 153
Appendix B: Online Supplementary Material for Study 2	158
Appendix C. Online Supplementary Material for Study 3	168
Appendix D: Online Supplementary Material for Study 4	
Appendix E: Online Supplementary Material for Study 5	190
References	191

List of tables

Table 2.1 Indications to perform caesarean section	25
Table 3.1 Data sources used to examine each research aim	46
Table 3.2 Studies in the thesis and their statistical analysis approaches	55

List of figures

Figure 1.1 Mechanisms by which obstetrical interventions during labour and birth may affect children's health and neurodevelopment
Figure 2.1 Caesarean section rates by regions according to UNICEF region 2 classification scheme
Figure 2.2 Caesarean section rates based on most recently available data across 164 countries.
Figure 2.3 A conceptual model that demonstrate where Apgar score falls in series of pregnancy and perinatal events
Figure 3.1 Countries (highlighted in blue) where the data for studies included in the thesis were drawn
Figure 3.2 DAG for the association between caesarean section and breastfeeding indicators.
Figure 3.3 The 'Three Delays Model'60

List of abbreviations and acronyms

ACOG	American College of Obstetricians and Gynaecologists
AIPW	Augmented Inverse Probability Weighting
aPR	Adjusted Prevalence Ratio
ATE	Average Treatment Effect
CI	Confidence Interval
DAG	Directed Acyclic Graph
DHS	Demographic and Health Surveys
GPS	Global Positioning System
HIV	The human immunodeficiency virus
IQ	Intelligence quotient
IRB	Institutional Review Board
IRSAD	Index of Relative Socio-economic Advantage and Disadvantage
IUGR	Intrauterine Growth Restriction
LMICs	Low and Middle Income Countries
NAPLAN	National Assessment Program – Literacy and Numeracy
NICE	National Institute for Health and Clinical Excellence
NICU	Neonatal Intensive Care Unit
NMS	National Minimal Standard
OR	Odds Ratio
PGE1	Prostaglandin E1
PGE2	Prostaglandin E2
PROM	Pre-labour rupture of membranes
PSU	Primary Sampling Unit

RR	Relative Risk
SAECDP	South Australian Early Childhood Data Project
SD	Standard deviations
SEP	Socioeconomic position
SOGC	Society of Obstetricians and Gynaecologists of Canada
UNICEF	United Nations International Children's Emergency Fund
USAID	United States Agency for International Development
WHO	World Health Organization

Abstract

Background

Obstetrical interventions during labour and birth are essential for perinatal care as part of any contemporary obstetric practice. Various underlying biological mechanisms have been proposed in linking obstetrical interventions during labour and birth with breastfeeding, infant's health and children's neurodevelopmental outcomes at later life. These mechanisms include changes in the gut microbiota composition, exposure to different levels of physical stress and stress hormone surges during labour and delivery, as well as epigenetic alteration of gene expression. The available evidence regarding the effect of obstetrical interventions during labour and birth on short-and long-term outcomes is limited. For instance, much of the available evidence was generated from high-income countries. Moreover, many of the previous studies were hampered by non-longitudinal study designs, small sample sizes and inconsistent findings, which may be due to suboptimal control of confounders and other biases. This thesis addresses these issues and utilises data from low-, middle- and high-income country settings.

Aims

The overarching aim of this thesis is to examine the effect of obstetrical interventions during labour and birth on breastfeeding indicators (early initiation of breastfeeding, exclusive breastfeeding under 6 months, and children ever breastfed), neonatal mortality and children's educational outcomes at eight years of age.

Specifically, the aims include:

viii

- To estimate the prevalence and examine sociodemographic factors associated with caesarean section in Ethiopia.
- To examine the changing temporal association between caesarean birth and neonatal death in Ethiopia from 2000 to 2016 as well as to provide an interpretation of the associations using the 'Three Delays Model' in the context of Ethiopia.
- To investigate the effect of caesarean section on breastfeeding indicators—early initiation of breastfeeding (within 1 hour), exclusive breastfeeding under 6 months and children ever breastfed (at least once)—in each of the 33 countries in sub-Saharan Africa, as well as to summarise the magnitude of these within-country effects in an overall estimate using random-effects meta-analyses.
- To examine the effect of Apgar scores of 0-5, 6, 7, 8 and 9 (compared with 10) on children's educational outcomes at eight years of age.
- To estimate the effect of elective induction of labour at 39 weeks of gestation as compared with expectant management on children's educational outcomes at eight years of age.

Methods

Data for this thesis were drawn from the Demographic and Health Surveys (DHS) and the South Australian Early Childhood Data Project (SAECDP). The DHS are widely available high-quality data sources from low- and middle-income countries. The SAECDP is an established project that encompasses high-quality whole-of-population linked administrative data from state and federal sources in South Australia. The DHS data from 33 low- and middle-income countries in sub-Saharan Africa were used for the first three studies while the

ix

SAECDP data from South Australia were used for the final two studies in this thesis. The use of these two different data sources allowed this thesis to capture the effects of obstetrical interventions during labour and birth on women's breastfeeding practices, neonatal health and children's educational outcomes across diverse health system resource settings. For each study, the potential confounding was identified based on *a priori* subject matter and expert knowledge as well as through the use of the Directed Acyclic Graphs (DAGs). The analytic approaches to answer the aims of this thesis included the modified Poisson regression (Log-Poisson regression), augmented inverse probability weighed (AIPW) estimator, negative control outcome (a tool for detecting confounding and bias), random-effects meta-analysis as well as an application of the 'Three Delays Model'.

Results

In the first study, the national caesarean section rate increased from 0.7% in 2000 to 1.9% in 2016, with increases across 7 of the 11 administrative regions in Ethiopia. In the adjusted analysis, women who gave birth in a private health facility had a 78.0% higher risk of caesarean section (adjusted prevalence ratio (aPR) (95% CI) 1.78 (1.22 to 2.58) when compared to women who gave birth in public health facility. Having four or more births was associated with a lower risk of caesarean section compared to first births (aPR (95% CI) 0.36 (0.16 to 0.79)).

In the second study, in Ethiopia, the adjusted prevalence ratios (aPR) for neonatal death among neonates born via caesarean section versus vaginal birth increased over time, from 0.95 (95% CI, 0.29 to 3.19) in 2000 to 2.81 (95% CI, 1.11 to 7.13) in 2016. The association between caesarean birth and neonatal death was stronger among rural women (aPR (95% CI)

3.43 (1.22 to 9.67)) and among women from the lowest quintile of household wealth (aPR (95% CI) 7.01 (0.92 to 53.36)) in 2016. On the other hand, the aggregate-level analysis revealed that increased caesarean section rates were correlated with a decreased proportion of neonatal deaths.

In the third study, the within-country analyses in sub-Saharan Africa showed, compared with vaginal birth, caesarean section was associated with aPR for early initiation of breastfeeding that ranged from 0.24 (95% CI, 0.17 to 0.33) in Tanzania to 0.89 (95% CI, 0.78 to 1.00) in South Africa. The aPR for exclusive breastfeeding under 6 months ranged from 0.58 (95% CI; 0.34 to 0.98) in Angola to 1.93 (95% CI; 0.46 to 8.10) in Cote d'Ivoire, while the aPR for children ever breastfed ranged from 0.91 (95% CI, 0.82 to 1.02) in Gabon to 1.02 (95% CI, 0.99 to 1.04) in Gambia. The meta-analysis combining effect estimates from 33 countries in sub-Saharan Africa showed caesarean section was associated with a 46% lower prevalence of early initiation of breastfeeding (pooled aPR, 0.54 (95% CI, 0.48 to 0.60))). However, the pooled effects indicated there was little association with exclusive breastfeeding under 6 months (pooled aPR, 0.94 (95% CI; 0.88 to 1.01) and children ever breastfeed (pooled aPR, 0.98 (95% CI; 0.98 to 0.99) among caesarean versus vaginally born children.

In the fourth study, after adjusting for confounding, the risk differences comparing fiveminute Apgar scores of 0-5 with Apgar score of 10 for children scoring at/below the national minimum standard (NMS) on the National Assessment Program—Literacy and Numeracy (NAPLAN) tests for each domain were: reading (0.07 (95% CI -0.16 to 0.29)), writing (0.27 (95% CI -0.14 to 0.68)), spelling (0.15 (95% CI -0.10 to 0.40)), grammar (0.04 (95% CI -0.21 to 0.29)) and numeracy (0.21 (95% CI -0.04 to 0.45)). Risk differences for children

xi

performing at/below the NMS were also evident when Apgar score of 6 were compared with Apgar score of 10.

In the fifth (last) study, after adjusting for confounding, the average treatment effects (ATEs) comparing elective induction of labour at 39 weeks of gestation with expectant management for children scoring at/below the NMS on each domain were: reading (0.01 (95% CI -0.02 to 0.03)), writing (0.02 (95% CI -0.00 to 0.04)), spelling (0.01 (95% CI -0.01 to 0.04)), grammar (0.02 (95% CI -0.01 to 0.04)) and numeracy (0.03 (95% CI 0.00 to 0.05)).

Conclusions

The findings from this thesis present a comprehensive analyses of the effect of obstetrical interventions during labour and birth on breastfeeding, neonatal mortality and children's educational outcomes at eight years of age by utilising data from low-, middle-, and high-income countries. The findings of Study 1 highlighted that there were large disparities in caesarean section use in Ethiopia, demonstrating unequal access. The results from Studies 2, 3 and 4 suggest that obstetrical interventions during labour and birth (caesarean section and Apgar score) have an influence on neonatal mortality, breastfeeding and children's educational outcomes at later age. However, the findings of Study 5 suggest that elective induction of labour at 39 weeks of gestation as compared with expectant management did not affect children's educational outcomes at eight years of age.

Acknowledgements

First and foremost, I would like to say 'thanks be to God for his indescribable gift'.

To my supervisors, Associate Professor Lisa Smithers, Professor John Lynch and Professor Ben Mol, I sincerely thank you for your guidance, unreserved support and encouragement throughout the PhD. As supervisors, you have imparted scientific rigour and have improved my research and writing over the years, which was motivating.

I would also like to acknowledge members of the *BetterStart* Child Health and Development Research Group who have contributed to the work in the thesis. In particular, Dr Helena Schuch who helped me to understand the SAECDP datasets. On numerous occasions, I have sought the advice of Dr Murthy Mittinty on statistical methods and he has always been most willing.

I was lucky to have supportive fellow PhD students and staff—Dr Razlyn Rahim, Mumtaz Begum, Alexandra Procter, Mi Du, Cherise Fletcher, Yonatal Tefera, Hassen Mohammed and Dr Blesson Varghese. I also want to thank Dr Sean O'Leary who has been very supportive during the first year of my candidature and for making himself easily accessible every time I need his help.

Special thanks to Adam Salleh—General Manager of Adelaide International Village—for his assistance in setting up the home office for free, when working in my regular office was very difficult because of the Coronavirus disease (COVID-19) pandemic. Adam, I can never thank you enough for your generosity.

I would like to acknowledge my previous mentor Associate Professor Ayalew Astatkie, who introduced me the path to research. My friend and colleague—Dr Stephany Ly, thank you for

endorsing my PhD application and providing encouragement throughout my candidature. Further thanks go to all colleagues and the management of School of Nursing and Midwifery at Addis Ababa University in Ethiopia for your support one way or another.

My best friends were with me in all my ups and downs during this time. Berhanu Yazew, Dr Gizachew Tessema and Dr Biruck Yirsaw—I am very grateful for your support, advice and friendship. Yonas Abebe—thank you for your unwavering support, advice and encouragement—I am most grateful for our brotherhood. I also want to acknowledge for the support I received from many friends in so many ways: Dr Zelalem Addis and his wife Belaynesh Asrat, Ketema Bizuwork, Amare Tigabu, Muktar Beshir Ahmed, Seifu Nigussie, and other associates in Australia and Ethiopia.

My family—you have been there all along. Mom and Dad, I am deeply indebted for your sacrifices and prayers. My brothers and sisters, I extend my gratitude to all of you for your encouragement and interest. In particular, Engineer Nigussie Yisma Derbie—thank you for your support and advice in all aspects of my endeavours.

And finally, I am very grateful for the excellent opportunity given to me through the provision of an Australian Government Research Training Program Scholarship.

Dedication

To my parents

Publications contributing to this thesis

- Yisma E, Smithers LG, Lynch JW, Mol BW. Cesarean section in Ethiopia: prevalence and sociodemographic characteristics. *J Matern Fetal Neonatal Med* 2019; **32**(7): 1130-5. https://doi.org/10.1080/14767058.2017.1401606
- Yisma E, Mol BW, Lynch JW, Smithers LG. The changing temporal association between caesarean birth and neonatal death in Ethiopia: secondary analysis of nationally representative surveys. *BMJ Open* 2019; 9(10):e027235.
 http://dx.doi.org/10.1136/bmjopen-2018-027235
- Yisma E, Mol BW, Lynch JW, Smithers LG. Impact of caesarean section on breastfeeding indicators: within-country and meta-analyses of nationally representative data from 33 countries in sub-Saharan Africa. *BMJ Open* 2019; 9(9): e027497. <u>http://dx.doi.org/10.1136/bmjopen-2018-027497</u>
- Yisma E, Mol BW, Lynch JW, Murthy NM, Smithers LG. Associations between Apgar scores and children's educational outcomes at eight years of age. *Aust N Z J Obstet Gynaecol* 2020; 1–7. <u>https://doi.org/10.1111/ajo.13220</u>
- Yisma E, Mol BW, Lynch JW, Murthy NM, Smithers LG. Elective labour induction vs expectant management in women and children's educational outcomes at 8 years of age. Ultrasound Obstet Gynecol 2020. <u>https://doi.org/10.1002/uog.23141</u>

Presentations arising out of this thesis

- Yisma E, Mol BW, Lynch JW, Smithers LG. The temporal association between caesarean birth and neonatal death in Ethiopia over time. *Australian Public Health Conference*, Adelaide, Australia, September 2019.
- Yisma E, Mol BW, Lynch JW, Smithers LG. The impact of caesarean section on breastfeeding indicators: within-country and meta-analyses of nationally representative data from 33 countries in sub-Saharan Africa. 13th Annual Florey Postgraduate Research Conference, Adelaide, Australia, September 2019.
- Yisma E, Mol BW, Lynch JW, Smithers LG. The changing temporal association between caesarean birth and neonatal death in Ethiopia. *Australasian Epidemiological Association* 2019 Annual Scientific Meeting, Brisbane, Australia, October 2019.
- 4. Yisma E, Mol BW, Lynch JW, Smithers LG. The impact of caesarean section on breastfeeding indicators: within-country and meta-analyses of nationally representative data from 33 countries in sub-Saharan Africa. *Australasian Epidemiological Association* 2019 Annual Scientific Meeting, Brisbane, Australia, October 2019.
- 5. Yisma E, Mol BW, Lynch JW, Smithers LG. The impact of caesarean section on breastfeeding indicators: within-country and meta-analyses of nationally representative data from 33 countries in sub-Saharan Africa. The American Public Health Association, Annual Meeting and Expo, Philadelphia, Pennsylvania, USA, November 2019.

Media coverage of findings arising from this thesis

1. 'World-first research links newborn test with NAPLAN score', The Australian, Sydney,

24 August 2020.

2. 'Low newborn health score linked to poor year 3 NAPLAN results', *Scimex (the Science Media Exchange)*, 24 August 2020.

Chapter 1: Introduction

1.1 Background

Obstetrical interventions during labour and delivery are essential for perinatal care as part of any contemporary obstetric practice. Even though the majority of women give birth following spontaneous labour and vaginal delivery, labour induction and caesarean section are the most frequently applied obstetrical interventions during labour and delivery. The rates of obstetrical interventions during labour and birth continue to rise worldwide.¹⁻⁴ About 1 in 4 pregnant women had their labour induced in high-income countries while the rate is generally lower in low-and middle-income countries.¹ On the other hand, births that occurred by caesarean section increased from 12.1% in 2000 to 21.1% in 2015 globally.⁵ There are also substantial global and regional disparities in the use of obstetric interventions during labour and birth.⁵⁻⁷ For instance, in low-and middle-income countries,^{8.9} labour induction and caesarean section are generally less common due to lack of access and unequal access. However, in high income countries such interventions could be overused without medical indications.^{5,10} Although it is essential to increase the obstetric interventions such as caesarean section for women and babies who are not medically eligible for the procedures.

When medically indicated, it is clear that obstetrical interventions during labour and childbirth are effective in preventing adverse outcomes. For instance, while labour induction and caesarean section are effective in improving perinatal outcomes by avoiding obstetric complications, caesarean section alone was associated with reduced risk of urogenital prolapse and urinary incontinence for mothers.^{11,12} Moreover, obstetrical interventions during labour and birth reduce maternal and neonatal mortality due to specific conditions

(indications). For instance, both induction of labour and caesarean section reduce maternal morbidity and mortality for women suffering from hypertensive disorders of pregnancy after 34 weeks of gestation.¹³ Conversely, both labour induction and caesarean section have risks exceeding the risks of spontaneous labour and spontaneous vaginal delivery outcomes, respectively. Induction of labour may cause uterine hyperstimulation and this may increase the risk of uterine rupture.¹⁴ Labour induction could also affect women's overall experience of childbirth because some methods of labour induction have side-effects and/or are painful or unpleasant.¹⁵ Likewise, caesarean birth may have numerous adverse health consequences such as increased risk of asthma, allergies, type 1 diabetes for the child and increased risk of cardiac arrest, hysterectomy and puerperal infection for mothers when compared to vaginal birth.¹⁶⁻¹⁸

Evidence from recent systematic reviews and meta-analyses, randomised control trials and observational studies of labour induction at or beyond term as compared with expectant management suggest that induction of labour was associated with decreased risk of caesarean deliveries, perinatal deaths and meconium-stained amniotic fluid.¹⁹⁻²⁵ Nevertheless, older observational studies²⁶⁻³¹ that compared women who were induced with women who gave birth following spontaneous onset of labour at the same gestational age demonstrated that labour induction increases caesarean section rates and adverse perinatal outcomes. However, it has been noted that the comparison of induction of labour and spontaneous labour at same gestational age is incorrect.³² This is because spontaneous labour is not a definite alternative to induction of labour. In clinical practice, women and obstetric care providers choose either induction of labour or expectant management, but not spontaneous labour or induction of labour at the same gestational age. Although there is a lot of evidence regarding the beneficial short-term perinatal outcomes of induction of labour at or beyond term, there is little

evidence of the effect of induction of labour versus expectant management on longer-term neurodevelopmental outcomes in children.

Similarly, evidence shows that caesarean section is linked with a number of short-term and long-term health consequences. The adverse health consequences of caesarean section are likely to be driven by the underlying comorbid conditions for which the procedure is warranted. Sandall and colleagues¹² in their 2018 review of studies that evaluated the shortterm and long-term effects of caesarean section revealed that, compared with vaginal birth, caesarean section was associated with increased risks of severe acute maternal morbidity, maternal mortality and adverse outcomes in subsequent pregnancies. In another 2018 systematic review and meta-analysis on long-term benefits and risks of caesarean birth, it was found that caesarean section was associated with increased odds of asthma, future subfertility and a number of subsequent pregnancy risks such as placenta previa, uterine rupture and stillbirth, when compared to vaginal birth.¹¹ Furthermore, in a 2012 systematic review and meta-analysis involving 53 studies from 33 low-, middle- and high-income countries (with no data from African countries),³³ it was found that early breastfeeding (that is, any initiation or breastfeeding at hospital discharge) was lower among infants delivered by caesarean section (OR, 0.57; 95% CI, 0.50 to 0.64) when compared to infants born vaginally. Longer periods taken to initiate breastfeeding are associated with greater difficulty in establishing breastfeeding and increases in neonatal mortality.³⁴ The effect of caesarean section on breastfeeding indicators from the perspective of low-and middle-income countries, particularly in sub-Saharan Africa, remains to be elucidated.

There are a number of studies³⁵⁻⁵¹ that have been conducted to examine the relationship between caesarean section and neonatal mortality, but the results are inconsistent. The

majority of the previous studies are at the aggregate-level so that they are comparing changes acting on the population as a whole (i.e., a change in caesarean section rates and neonatal mortality among geographic areas).^{35,36,44-51} Studies based on individual-level data³⁸⁻⁴¹ are more likely to report a considerably increased risk of neonatal mortality associated with caesarean versus vaginally born infants without providing an interpretation of the associations accounting for contextual factors. There seems to be no exploration of why the risk of neonatal mortality following caesarean delivery was increased in studies using individual-level data. Moreover, the previous studies are limited by either inconsistent results or failure to account for contextual factors such as unequal access, infrastructural, structural and health workforce constraints that could play a role in the association between caesarean section and neonatal mortality.

There are three underlying biological mechanisms that provide a possible explanation for how obstetrical interventions during labour and birth (more specifically mode of delivery) could affect children's health and development. The first proposed mechanism is altered bacterial (microbial) colonisation of the infant gut.⁵² The microbial colonisation of the infant gut starts during the intrauterine period when the foetus is exposed to maternal microbiota from the maternal gut and/or blood stream that enter the amniotic fluid crossing the placenta.⁵³ Exposure to the gut colonising bacteria continues at birth as the foetus passes through the birth canal and during the first year of life when the newborn is introduced to oral liquid and solid foods.⁵³ It has been demonstrated that the colonising microbiota are critical for normal development of host defence and are associated with a lifelong health effects in children, including brain development.^{53,54} However, evidence shows that interventions during labour and birth such as caesarean section result in altered gut microbiota patterns, leading to increased susceptibility to disease in children's later life.^{52,53} Moreover, it has also

been revealed that obstetric interventions such as caesarean section can affect the onset of lactation (i.e., lactogenesis), which results in physiological delay in milk secretion and shortening the duration of breastfeeding.⁵² This, in turn, has direct influence on microbiota development because human milk oligosaccharides are thought to shape the diversification of the infant intestinal microbiota (i.e., bifidobacteria colonisation).⁵⁵

The second hypothesis is "exposure to different levels of physical stress and stress hormone surges during delivery."⁵² During spontaneous vaginal delivery, the foetus is exposed to maternal stress hormones and physical forces of labour (uterine contraction), as well as a hypoxic journey through the birth canal. Uterine contraction together with the 'hypoxic journey through the birth canal' (that is, the stress of being born) triggers the release of catecholamine and this was found to be important for successful physiological foetal-toneonatal transition and survival.⁵⁶ Moreover, it has been suggested that increased concentrations of stress hormones are important "signals in the infants for development of the hypothalamic-pituitary-adrenal axis, maturation of the immune system, lung and organ maturation, and neurogenesis."¹² However, the timing and degree of the stress are altered when the infant is born following obstetric interventions because there is too little stress in pre-labour caesarean section or too much stress in case of labour induction/augmentation.⁵⁷ These conditions are contrary to the normal physiology of parturition. Unnecessary obstetrical interventions during labour and birth, which is often compounded by 'early term' delivery (e.g., planned caesarean section is usually performed from 37-39 weeks while elective induction of labour is offered at 39 weeks of gestation), may influence infants' health and neurodevelopment.

The final hypothesis is epigenetic alteration of gene expression.⁵² It has been suggested that adverse prenatal and perinatal stress (i.e., too high and too low stress) may permanently alter neuroendocrine and behavioural responses in later life.⁵⁸ Epigenetic alterations, which is alterations that induce heritable changes in gene expression without changes in DNA sequence, was proposed as the possible mechanism for such adaptive responses.⁵⁹ The most studied epigenetic control mechanism is DNA-methylation and it was shown that hypermethylation is generally associated with gene silencing while hypomethylation make genes remain switched on.⁵² Using umbilical cord blood sample from infants delivered by pre-labour caesarean section and vaginal births to analyse DNA-methylation, Schlinzig and colleagues found that infants delivered by prelabour caesarean section had higher global DNA-methylation in white blood cells than infants born vaginally.⁶⁰

Moreover, in one study that analysed the global DNA-methylation of the hematopoietic stem cell (CD34+) using cord blood from Swedish women, it was found that there were differences in the epigenetic mark-up between caesarean and vaginal birth groups in 343 DNA regions.⁶¹ Furthermore, in addition to caesarean section, EPIgenetic Impact of Childbirth (EPIIC) hypothesis suggested that use of synthetic oxytocin and antibiotics during intrapartum period have effects on neonatal epigenome remodelling processes.⁵⁷ According to the EPIIC hypothesis, foetal epigenomic remodelling anomalies, which have an influence on abnormal gene expression, may result from the reduced or elevated levels of adrenalin, oxytocin and cortisol produced during labour.⁵⁷ It was thought that this reprogramming could manifest in biobehavioral problems and diseases in infants and in children's later life, which is also accompanied by environmental exposures that may alter DNA-methylation.^{57,62} The pictorial presentation of the three underlying biological mechanisms that provide a possible

explanation for how obstetrical interventions during labour and birth could affect children's health and development is provided in **Figure 1.1**.



Figure 1.1 Mechanisms by which obstetrical interventions during labour and birth may affect children's health and neurodevelopment.

On the other hand, because of obstetrical interventions during labour and childbirth such as caesarean section and use of anaesthetics and analgesics, there are increased difficulties in successful foetal-to-neonatal transition.⁶³ For example, it was shown that pre-labour caesarean section reduces the postnatal rise in cortisol, which is a hormone that prepares the foetus for birth and supports the multi-organ extra-uterine transition.⁶³

1.2 Thesis aim and research questions

The overall aim of this thesis is to examine the effect of obstetrical interventions during labour and birth on breastfeeding, neonatal mortality and children's educational outcomes. The specific research questions to be addressed in this thesis are as follows.

- 1. What is the prevalence of caesarean section in Ethiopia and how are sociodemographic characteristics and caesarean section associated in Ethiopia?
- 2. How is caesarean delivery and neonatal death associated in Ethiopia from 2000 to 2016 and does the application of the 'Three Delays Model' facilitate the interpretation of the association between caesarean birth and neonatal death in Ethiopia?
- 3. What impact could caesarean section have on breastfeeding indicators—early initiation of breastfeeding, exclusive breastfeeding under 6 months and children ever breastfed—in each 33 country in sub-Saharan Africa and how big would this effect be when the available evidence (estimate) from each of the 33 countries is summarised in random-effects meta-analysis?
- 4. Do Apgar scores of 0-5, 6, 7, 8 and 9 (compared with 10) associated with children's educational outcomes at eight years of age?
- 5. Does elective induction of labour at 39 weeks of gestation as compared with expectant management affect children's educational outcomes at eight years of age?

1.3 Thesis outline

This thesis—which is presented as a 'thesis by publication'—is organised into nine chapters. Chapter 1, this chapter, presents the introduction, including the description of background, the problem being addressed, the overall aim of the thesis, and the five research questions examined.

Chapter 2 provides a comprehensive literature review of studies related to obstetrical interventions during labour and birth such as induction of labour and caesarean section, and their short- and long-term outcomes. This chapter delivers the foundation of knowledge regarding the effect of obstetrical interventions during labour and birth on maternal and child outcomes, including breastfeeding, maternal and neonatal mortality and children's educational outcomes from the existing literature. Chapter 2 also reviews the gaps in the research literature that motivated the five research questions examined in this thesis.

Chapter 3 describes the various research methods that are used by studies in the thesis. This chapter begins with the descriptions of the two data sources—the Demographic and Health Surveys (DHS) and the South Australian Early Childhood Data Project (SAECDP). Then, the chapter describes how the exposures (measures of obstetrical interventions during labour and birth), the outcomes (measures of breastfeeding indicators, neonatal mortality and children's educational outcomes), and the potential confounders were formulated for each study. Finally, Chapter 3 discusses the statistical analysis approaches used to answer the five research questions.

Chapters 4-8 deliver the five studies arising from this thesis. All of these studies are already published in esteemed peer-reviewed journals. Chapter 4 presents the result of the

comprehensive analysis concerning the prevalence and sociodemographic characteristics associated with caesarean section in Ethiopia from 2000 to 2016 (Study 1). Chapter 5 examines the temporal association between caesarean birth and neonatal death in Ethiopia from 2000 to 2016, as well as it applies the 'Three Delays Model' to facilitate the interpretation of the association between caesarean birth and neonatal mortality in view of contextual factors in Ethiopia (Study 2). Chapter 6 presents the results of novel analysis examining the impact of caesarean section on breastfeeding indicators in 33 countries in sub-Saharan Africa, as well as it summarises the evidence (estimate) from each of the 33 countries in sub-Saharan Africa regarding the association between caesarean section and breastfeeding indicators using random-effects meta-analysis (Study 3). Chapter 7 presents the results of novel analysis examining the associations between both one- and five-minute Apgar scores of 0-5, 6, 7, 8 and 9 (compared with 10) and children's educational outcomes at eight years of age in South Australia (Study 4). Chapter 8 presents the design and results of the final research article in this thesis (Study 5), which investigates the effect of elective induction of labour at 39 weeks of gestation as compared with expectant management on children's educational outcomes at eight years of age.

Chapter 9, last chapter, summarises the overall research by presenting the key findings and contributions of each study, the strengths and limitations, as well as implications and conclusions.

Chapter 2: Literature review

The purpose of this literature review is to provide a general overview of obstetrical interventions during labour and birth such as induction of labour and caesarean section. Recently, obstetrical interventions during labour and birth have become a growing subject area of research interest. This is largely because the rates of obstetrical interventions during labour and birth have been substantially increasing globally. The first part of this chapter gives a brief description about induction of labour and its effects on perinatal outcomes, including caesarean delivery. Next, this chapter reviews evidence for caesarean section and its effects on maternal and child outcomes, including maternal and neonatal morbidity and mortality, breastfeeding, and children's educational outcomes. Finally, a brief overview of the Apgar scores and its effects on short-term perinatal outcomes, as well as its effects on longer-term neurodevelopmental outcomes in children's later life is given. The research justification for the studies examined in this thesis is also provided at the end of this chapter.

2.1 Induction of labour

Induction of labour is the most frequently applied obstetric intervention to artificially stimulate the start of labour. It is thought that when medically indicated, induction of labour is effective to improve perinatal outcomes by avoiding the effects of prolonged pregnancy. When labour induction is performed without any maternal or foetal indication, including for matters of convenience or preference, it is termed as non-medically indicated induction of labour (also known as elective induction of labour).

Evidence shows that the rate of induction of labour is generally high (more than 1 in 4) in high-income countries.¹⁴ Data from government reports indicate a proportion of induction of labour up to 43.1% (for selected women giving birth for the first time) in Australia in 2017,⁶⁴ 27.1% in the United States in 2018,⁶⁵ and from 6.8% to 33% in European countries in 2010.⁶⁶ In low-and middle-income countries, the rates of labour induction are generally low when compared to the rates in high-income countries. However, high rates have been reported in countries such as Sri Lanka (35.5%)⁸ and Cuba (20.1%),⁶⁷ where rates of induction of labour are similar to the rates in high-income countries. In African countries, the average rate of induction of labour was 4.4%, ranging from 1.4% in Niger to 6.8% in Algeria.⁸

It is clear that induction of labour is recommended when maternal and neonatal risks are believed to be lower with induction of labour rather than with expectant management. Expectant management is waiting for onset of spontaneous labour or waiting until the occurrence of a medical condition that indicate induction or caesarean delivery. Furthermore, the procedure of labour induction should only be conducted when the vaginal birth is considered to be the most definite method of delivery and/or there is no contraindication to vaginal delivery (e.g., foetus in transverse lie). It also recommended that induction of labour

should be undertaken after informed consent.⁶⁸ The consent should be accompanied by explanations regarding the medical indications for induction of labour and the benefits, risks and the choice of methods to be used.

2.1.1 Indications for induction of labour and reasons for elective labour induction

Indications for induction of labour vary according to the different guidelines available. There are five widely known guidelines regarding induction of labour. These include the World Health Organization (WHO) recommendations for induction of labour;¹⁴ the Queensland Clinical Guideline;⁶⁹ the Society of Obstetricians and Gynaecologists of Canada (SOGC) clinical practice guideline;⁶⁸ the American College of Obstetricians and Gynaecologists (ACOG) clinical management guidelines for obstetrician–gynaecologists;⁷⁰ and the National Institute for Health and Clinical Excellence (NICE) clinical guidelines.⁷¹ Unlike the other guidelines, the ACOG clinical management guideline provide specific clinical situations (indications) in a series of publications.⁷²⁻⁷⁴

There is a general agreement among these guidelines in recommending induction of labour for some indications. For instance, for near or beyond term pregnancy, all of the guidelines recommend induction of labour when gestation is between 41 and 42 weeks. Likewise, all guidelines agree that induction of labour should be offered to women with pre-labour rupture of membranes (PROM). Conversely, for suspected foetal macrosomia alone, all guidelines suggest that induction of labour at term is not recommended. Some other most common indications for induction of labour recommended by at least one guideline include abruptio placentae, chorioamnionitis, pregnancy hypertension (pre-eclampsia/eclampsia), preterm prelabour rupture of membranes (pPROM), intrauterine foetal growth restriction (IUGR) and oligohydramnios.

In addition to the medical reasons (indications) for labour induction, there are others nonmedical reasons why women and obstetric care providers choose induction of labour. For instance, because of physical discomfort, scheduling issues, or concerns for maternal, foetal or neonatal complications as the pregnancy continues, women may request induction of labour. Similarly, due to the concern that the quickly progressing labour would prevent timely access to hospital or receiving interventions (such as pain relief), women may want to end their pregnancy.⁷⁵ Obstetric care providers may also suggest labour induction not only due to concern about complications that may arise as the pregnancy continues, they may also want to manage women's pregnancy-related discomfort.⁷⁵ Some obstetric providers may recommend labour induction because they may be incentivised financially or via their own scheduling preferences.^{3,75}

2.1.2 Methods of induction of labour

A variety of methods for induction of labour are discussed in the existing literature.⁷⁶⁻⁸⁰ These include pharmacological, mechanical, and complementary and alternative methods. Oxytocin and prostaglandins (i.e., prostaglandin E1 (PGE1)—misoprostol and PGE2—dinoprostone) are the pharmacological methods. While oxytocin is available in tablet, gel or insert forms, misoprostol is administered via different routes (oral, oral titrated solution, buccal/sublingual or vaginal).^{80,81} It has been demonstrated that PGEs are mainly used when the cervix is unfavorable while oxytocin is often administered when the cervix is favourable.⁸¹

Some of the mechanical methods of labour induction include Foley catheter and membrane sweeping. Foley catheter is the most common mechanical method of labour induction. The Foley catheter alone⁸² or in combination with oxytocin⁸³⁻⁸⁵ and misoprostol^{86,87} could be used for pre-induction cervical ripening or labour induction. Evidence shows that when Foley

catheter is used together with oxytocin, it increases the rate of delivery within 24 hours in nulliparas.⁸⁵ Furthermore, when Foley catheter is used with misoprostol, it reduces the intervention-to-delivery time interval.⁸⁸ In a 2020 Cochrane systematic review of 44 randomised studies⁸⁹ conducted in high-, middle- and low-income countries, it was suggested that membrane sweeping as a method of labour induction at or above 36 weeks of gestation was probably effective in increasing the chance of achieving spontaneous onset of labour. However, this evidence was of low certainty.

The complementary and alternative methods of labour induction include acupuncture or acupressure,⁹⁰ castor oil,⁹¹ breast/nipple stimulation,⁹² homeopathy,⁹³ hypnosis⁹⁴ and sexual intercourse.⁹⁵ Because of limited evidence available regarding all these complementary and alternative methods, their role for labour induction is less clear.

2.1.3 Induction of labour and caesarean section

Historically, it has been considered that elective induction of labour was unsafe because it increases the risk of caesarean delivery and is associated with poorer perinatal outcomes when compared to spontaneous labour. For example, in their cohort study on 14,409 women between 36 and 42 weeks of gestation attending two large hospitals in Boston, USA from 1998-1999, Heffner and colleagues⁹⁶ found that, compared with spontaneous onset of labour, induction of labour was associated with a 70% increase in risk of caesarean delivery in nulliparous (aOR, 1.70; 95% CI, 1.48 to 1.95) and a 49% increase in risk of caesarean delivery in studies^{26,29,97-100} have demonstrated similar increased risk of caesarean delivery following induction of labour compared with spontaneous onset of labour.

The results of several previous studies, which compared women who were induced (either electively or medically) and women who gave birth following spontaneous onset of labour, have provided a consistent story that induction of labour was associated with increased risk of caesarean delivery. However, over time, it was noted that using spontaneous labour as a control group was inappropriate as a counterfactual for induced labour. In clinical practice, women and obstetric care providers choose either induction of labour or expectant management, but not spontaneous labour or induction of labour at the same gestational age. The more accurate comparison group is expectant management. Women carry inherent risks of developing obstetric complications when they are expectantly managed. As the pregnancy continues, spontaneous labour onset may occur or women may develop complications (such as post-term pregnancy, oligohydramnios or foetal death), or require emergency interventions (i.e., medically-indicated induction or caesarean section).¹⁰¹ Therefore, it is most plausible to compare the risks of induction of labour with the risks of continuing the pregnancy (expectant management).

Several recent systematic reviews and meta-analyses, randomised control trials and observational studies that evaluated labour induction at or beyond term, compared with expectant management, have been conducted. These studies have found that induction of labour at or beyond term reduces the rate of caesarean delivery. For instance, in 2019, Grobman and colleagues conducted a systematic review and meta-analysis of 6 cohort studies involving 66,019 women who were electively induced at 39 weeks and 584,390 women who were expectantly managed.²⁰ They found that elective induction of labour at 39 weeks was associated with a 17% decrease in the rate of caesarean delivery compared with expectant management (RR, 0.83; 95% CI, 0.74 to 0.93). Furthermore, a 2014 systematic and meta-analysis by Mishanina *et al.*¹⁰² found, compared with expectant management, lower risk of

caesarean delivery among women who were induced even when subgroup analysis by method of induction (e.g., prostaglandin E2, misoprostol, alternative method and mixed method) was performed. Lower rates of caesarean delivery among women who gave birth following labour induction compared with expectant management were also demonstrated by another three systematic reviews and meta-analyses.^{19,101,103}

In a 2018 large multicenter randomised control trial involving 6,106 low-risk nulliparous women randomised to elective labour induction at 39 weeks of gestation (n=3,062) or expectant management (n=3,044) groups found that the rates of caesarean delivery were lower among the elective labour induction group (RR, 0.84; 95% CI, 0.76 to 0.93).²¹ Evidence from several recent observational studies²²⁻²⁵ also showed that labour induction at 39 weeks resulted in lower rates of caesarean delivery when women who were induced were compared with women who were expectantly managed. Therefore, it has become clearer that when a more appropriate comparison group (i.e., expectant management) was used, induction of labour was associated with lower rates of caesarean delivery.

2.1.4 Induction of labour and maternal and newborn outcomes

There is evidence regarding whether the policy of induction of labour in women at or beyond term as compared with expectant management has effects on maternal and perinatal health outcomes. For example, in 2018, Middleton and colleagues¹⁰⁴ included 30 randomised control trials from 14 countries in their Cochrane systematic review to evaluate the effect of labour induction at or beyond term compared with expectant management on pregnancy outcomes for the infant and mother. They found that, compared with expectant management, induction of labour was associated with lower perinatal deaths (RR, 0.33; 95% CI, 0.14 to 0.78), fewer stillbirths (RR, 0.33; 95% CI, 0.11 to 0.96), lower rates of neonatal intensive
care unit (NICU) admission (RR, 0.88; 95% CI, 0.77 to 1.01) and fewer newborns had fiveminute Apgar scores <7 (RR, 0.70; 95% CI, 0.50 to 0.98). For the mother, this review also found that induction of labour was associated with lower rates of caesarean birth (RR, 0.92; 95% CI, 0.85 to 0.99), but an increase in operative vaginal delivery (RR, 1.07; 95% CI, 0.99 to 1.16) when compared with expectant management.

Saccone *et al.* ¹⁰⁵ in their 2019 systematic review and meta-analysis involving 7 randomised clinical trials (n= 7,598) demonstrated that induction of labour at full-term, compared with expectant management, was associated with a 68% decrease in the risk of meconium-stained amniotic fluid (RR, 0.32; 95% CI, 0.18 to 0.57) and a 98.96 g lower mean birthweight (mean difference -98.96 g; 95% CI, -126.29 to -71.63). These findings may mean that labour induction at full-term was associated with better meconium outcomes, but lower birth weight. However, Saccone *et al.* suggested that, at full term, a mean difference of about 100 g may not be significant clinically.¹⁰⁵

Moreover, in other systematic reviews and meta-analyses, randomised control trials and observational studies, induction of labour at or beyond term was associated with reduced risk of perinatal mortality,^{21,106} lower rates of maternal and neonatal morbidity,^{22,24} reduced risk of hypertensive disease of pregnancy (i.e., hypertension that would manifest later),^{19,21,25} reduced risk of meconium aspiration syndrome,^{22,107} lower rates of admission to NICU unit,^{22,107} and reduced need for neonatal respiratory support^{19,23} compared with expectant management.

These previous studies focused on the effect of labour induction from 39-41 weeks of gestation versus expectant management on short-term maternal and perinatal outcomes,

including caesarean delivery. The previous studies have demonstrated that labour induction at or beyond term, compared with expectant management, had lower rates of maternal and neonatal morbidity, lower rates of caesarean section, and fewer stillbirths. These findings would appear to be intuitive, given the fact that with advancing gestational age, the rate of stillbirth and neonatal morbidity also increases.¹⁰⁸ This is because as gestational age increases, the placenta may not provide adequate supply of nutrition to the foetus, resulting in placental insufficiency. Moreover, foetal weight also increases as the gestational age increases and may results in foetal macrosomia (i.e., birth weight \geq 4500 g).

2.1.5 Elective induction of labour and long-term neurodevelopmental outcomes

Because of the risk of adverse short- and long-term outcomes associated with non-medically indicated deliveries, it has been recommended that deliveries following elective induction of labour for women who are below 39 weeks of gestation should be avoided.¹⁰⁹ However, due to the increased risk of perinatal morbidity and mortality associated with late-term and post-term pregnancies, deliveries following induction of labour should be considered when the gestation is at/above 41 weeks.^{14,72} Therefore, the appropriate time when elective induction of labour should be offered to women should be at 39 or 40 weeks of gestation.

Several studies¹⁹⁻²⁵ have reported the beneficial effects of elective labour induction at 39 weeks on short-term perinatal outcomes, including reduction in caesarean delivery rates. However, there has been limited research reported on the link between labour induction at 39 weeks of gestation and children's longer-term neurodevelopmental outcomes at later life.¹¹⁰ It would be important to investigate the effect of elective labour induction at 39 weeks of gestation versus expectant management on children's cognitive outcomes because a 2010 study by MacKay and colleagues¹¹¹ found that the risk of special educational need increased

among children born from 37–39 weeks gestation when compared to children born at 40 weeks. Moreover, it is thought that labour induction increases stress and the presence of adverse prenatal and perinatal stress may result in epigenetic alteration of gene expression, which may permanently change neuroendocrine and behavioural responses in children's later life.⁵² However, this mechanism remains to be fully investigated.

2.2 Caesarean section

It is known that obstetric complications such as antepartum haemorrhage, foetal distress, abnormal foetal presentation and hypertensive disease of pregnancy could be life-threatening for mothers and newborns. When such complications occur, caesarean section is the most common obstetric intervention to save the life of mothers and infants, and should be accessible universally.

2.2.1 Caesarean section rates and disparities in caesarean section uses

For the last three decades, the frequency of caesarean section has increased beyond the proportion that was considered to be optimal (10–15% of births) by agencies such as the WHO.^{46,112-114} For example, in their 2018 study on the use of and disparities in caesarean sections, Boerma *et al.*⁵ estimated that, globally, the total number of births that occurred following caesarean section increased from about 16.0 million (12.1%) in 2000 to about 29.7 million (21.1%) in 2015. Non-medically indicated reasons have been thought to be responsible for this increase in use of caesarean section in many middle- and high-income countries.^{44,46,112} In 2015, the WHO released a declaration on caesarean section stating that rates exceeding 10% were not associated with reductions in maternal and neonatal mortality.¹¹³ Molina *et al.* (2015) in their ecological study, however, estimated that the national caesarean section rates up to 19% were associated with lower maternal and neonatal mortality.³⁵

Even though there are modest increase in caesarean section use in several low- and middleincome countries (LMICs) such as in sub-Saharan African countries, the national rate of caesarean birth is still less than 10%.⁵ This may suggest inadequate access to emergency caesarean section. In sub-Saharan Africa, one of the reasons for the modest increase in the caesarean section rate was an increase in the proportion of deliveries occurring within health facilities. This is because caesarean sections are only performed in health facilities. **Figure 2.1** shows, at a global regional level (using the UNICEF region 2 classification scheme), that caesarean section rates varied from 4.1% in West and Central Africa to 44.3% in Latin America and Caribbean region in 2015.⁵



Figure 2.1 Caesarean section rates by regions according to UNICEF region 2 classification scheme.

Source: Adapted from CNN. https://edition.cnn.com/2018/10/11/health/c-section-rates-study-parenting-without-borders-intl/index.html.

As shown in **Figure 2.2**, at the country level, caesarean section rates varied widely. For example, the national rate of caesarean section was lowest in South Sudan (0.6%) and highest in Dominican Republic (58.1%) in 2015.⁵



Figure 2.2 Caesarean section rates based on most recently available data across 164 countries.

Source: Adapted from CNN. https://edition.cnn.com/2018/10/11/health/c-section-rates-study-parenting-without-borders-intl/index.html.

Moreover, there are a substantial within-country socioeconomic and geographic disparities in caesarean section rates. In LMICs, the rate of caesarean section among births in the richest quintiles of household wealth was about five times more than births from women in the poorest household quintiles.⁵ These substantial economic inequalities in caesarean section

may suggest inadequate access to emergency caesarean section among the poorest subgroups and increased use without medical indication in the richest subgroups, especially in middleincome countries.¹¹⁵

The rates of caesarean section according to within-country geographic differences were also found to be large. For instance, while the national caesarean section rate in Ethiopia is about 2% in 2016, the reported rate for Addis Ababa—capital city of Ethiopia—was 21.4%.¹¹⁶ Although the national caesarean section rates were 18.5% and 32.7% in China and India, respectively, the inter-state rate varied from 7% to 49% in India and the provincial rate ranged from 4% to 62% in China.⁵

Hoxha and colleagues¹¹⁷ in their 2017 systematic review and meta-analysis of 11 studies conducted in upper middle income and high income countries, found that the odds of caesarean delivery was about 41% higher for women who gave birth in private (for-profit) hospitals (OR, 1.41; 95% CI, 1.24 to 1.60) than non-profit hospitals. This may suggest that the increase in caesarean section use might be related to an increase in privatisation of obstetric care services.

2.2.2 Indications and drivers of overuse of caesarean section

In clinical practice, the decision to perform caesarean section is based on the presence of lifethreatening causes (indications) for the mothers and newborns. **Table 2.1** shows the indications (medical causes) leading to caesarean delivery.
 Table 2.1 Indications to perform caesarean section.

Indication*	Description	
Cephalo-pelvic disproportion	Small maternal pelvis, making vaginal birth	
	impossible	
Chorioamnionitis (amniotic infection syndrome)	Infection of the placenta, necessitating	
	urgent delivery	
Maternal pelvic deformity	Anatomical deformity, making vaginal birth	
	complicated/impossible	
Eclampsia and HELLP syndrome	Life-threatening problems of pregnancy	
Foetal asphyxia or foetal acidosis	Life-threatening circumstances for the	
	foetus, leading to foetal hypoxia	
Umbilical cord prolapse	Prolapse of the umbilical cord, resulting in	
	foetal asphyxia	
Placenta previa	Abnormal placental position, hindering	
	vaginal delivery	
Abnormal lie and presentation	Abnormality of foetal position, making	
	vaginal delivery very difficult/impossible	
Uterine rupture	e rupture Acute condition threatening the life of both	
	mother and foetus	
Failure to progress in labour (prolonged labour,	Delayed delivery or cessation of labour,	
secondary arrest)	resulting in bad outcome for the foetus or	
	newborn	
Previous caesarean section	It is generally anticipated that having had	
	one caesarean section makes it difficult to	
	have a vaginal delivery in succeeding	
	pregnancies.	

*Source: Mylonas and Friese, 2015.¹¹⁸

When caesarean section is performed without any medical indications, it is termed as caesarean delivery on maternal request. And, this may lead to overuse (excess use) and increasing use of caesarean section.¹¹⁸ Although, it is difficult to determine the optimal

caesarean section rate, a 2018 paper in the *Lancet Series*⁵ on optimising the use of caesarean section defined population level caesarean section use thresholds of 15% as an indicator of overuse and 10% as an indicator of poor access (underuse).

It has been illustrated that, when the rate of caesarean section is higher than what is needed $(\geq 15\%)$, the drivers of excess use and increasing use of caesarean section could be described in three broader 'categories of reasons.'¹¹⁹ The first category of reasons is related to the different factors linked to childbearing women, families, communities and the broader society.¹¹⁹ Women may request caesarean birth due to fears related to labour pain, pelvic floor damage, urinary incontinence and fear of negative effects on their sexuality or sexual relationships.^{52,118,119} Other reasons could include perceived belief that caesarean section is safe for the newborn and mother as well as some women also consider caesarean birth as a convenient method of delivery (e.g., along with caesarean section, some women also undergo tubal ligation).^{119,120} Moreover, spouses' preference and influence related to convenience, previous negative experience of a partner's labour or birth and a range of other social and cultural pressures are also some of the reasons for increased request for caesarean section.^{52,119}

The second category is regarding factors related to health professionals.¹¹⁹ In most countries, obstetric care provides play a role in the choice of mode of delivery. In addition to women's requests, health-care providers are more likely to make decisions to use caesarean section for financial incentives, convenience and fear of litigation.^{119,121} For example, it is believed that caesarean section is a protective procedure, although can be contrary to the scientific evidence.¹²² As a consequence of this, obstetric care providers could be less likely to be

litigated for performing unnecessary caesarean section rather than for complications that occur during spontaneous vaginal delivery.¹²²

The final category of reasons regarding excess and increasing use of caesarean section is related to factors associated with health-care systems, financial reimbursements, and organisational design and cultures.¹¹⁹ In most settings, caesarean delivery rates are higher in private sectors than in the public health facilities. For instance, the caesarean section rates in private sector in Brazil is about 80–90% while it is about 35–45% in public sector.¹²³ Such large differences may suggest that there are financial incentives to convince women that caesarean section is the best method of childbirth in private practice. In addition to the hospital infrastructure in the private sector, which may favour caesarean delivery (e.g., due to inadequate delivery rooms or birth centres), it has been suggest that the financial reimbursement offered by private health insurance for vaginal delivery is also low.¹²³

Furthermore, inadequate skill or inexperience to perform assisted vaginal delivery (e.g., forceps) may also increase the use of caesarean section.^{119,124} Women's mistrust in the health system and its staff because of their experiences of poor-quality antenatal care, equipment and interactions with obstetric care providers make women opt to have caesarean section.¹²⁵ This, in turn, results in increased use of caesarean section. In resource-constrained settings, the use of caesarean section has been increasing in referral (tertiary) hospitals. This increased use is partly attributable to unnecessary delays in referring women in labour by unskilled obstetric caregivers in primary health facilities.¹²⁴ It is clear that the obstetric complications encountered by mothers in primary health facilities will be compounded by delays when they are referred. The referred women are more likely to arrive late at the tertiary hospital in

critical condition and they often undergo emergency caesarean section as it is the only available option.

2.2.3 Caesarean section and maternal morbidity and mortality

Timely caesarean intervention can successfully decrease maternal and perinatal morbidity as well as mortality when the mothers and/or foetus encounter obstetric complications. However, it was demonstrated that the proportion of maternal morbidity and mortality was higher following caesarean than spontaneous vaginal delivery.¹² For example, a 2018 study on short-term and long-term effects of caesarean section found that, compared with vaginal birth, caesarean section is associated with increased risk of maternal severe acute morbidity and mortality, including an increased risk of adverse outcomes in subsequent pregnancy.¹² Furthermore, a large study conducted by Villar *et al.*³⁸ in 410 health facilities in 24 areas in eight randomly selected Latin American countries found that, compared with 2.0 (95% CI, 1.6 to 2.5) and 2.3 (95% CI, 1.7 to 3.1) times higher odds of severe maternal morbidity, respectively.

As maternal mortality rate is higher in low- and middle-income countries when compared to high income countries, it was also found that the maternal mortality rate following caesarean birth is higher in low-resource settings than in high-resource settings. For instance, a 2019 systematic review and meta-analysis by Sobhy *et al.*¹²⁶ which included 196 studies, covering 12 million pregnancies in 67 LMICs, found that the risk of maternal death following caesarean section was 7.6 per 1000 procedures (95% CI, 6.6 to 8.6). This figure is about 100 times higher than the risk in high-income countries (e.g., the risk in UK is eight deaths per 100,000 caesarean sections).^{126,127} They also found that a quarter of all maternal deaths in

LMICs were related to caesarean section (23.8%; 95% CI, 21.0 to 26.7), and women who underwent emergency caesarean section were twofold more likely to die than women who underwent elective caesarean section.¹²⁶

Bishop *et al.*¹²⁸ in their 2019 African Surgical Outcomes Study followed more than 3,500 women having caesarean section in 183 hospitals across 22 countries in Africa. They found that in-hospital maternal mortality rate following caesarean delivery was 5.43 per 1000 operations (95% CI, 3.1 to 7.8). This figure shows that African women are may be 50 times more likely to die following caesarean delivery when compared to maternal mortality rates in high-income countries.^{129,130} Moreover, about 1 in 6 women developed complications after caesarean delivery in Africa, which is approximately about three times the rate in the USA (a high income country).¹³⁰

In high income countries, it is more likely that caesarean section is undertaken for reasons other than medical indications but it leads to lower risks of mortality and morbidity. However, the increased risks of maternal morbidity and mortality following caesarean delivery in LMICs when compared with high-income countries is probably due to the limited access to comprehensive obstetric care in LMICs.^{37,38,131} For example, in Africa, the caesarean section rate is low and this may mean that several women who would benefit from caesarean section do not access the procedure. This situation is often accompanied by lack of skilled birth attendants and poor health infrastructure (e.g., shortage of medical care institutions, deficiencies in surgical facilities, surgical and anaesthesia personnel and equipment, and blood transfusion capacity). Further study is needed to inform the public health policy regarding the benefit of increasing timely access to caesarean section and timely

decision for caesarean delivery in LMICs. This may help decrease maternal morbidity and mortality rates in LMICs.

2.2.4 Caesarean section and neonatal mortality

As any surgical procedure, caesarean section has risks of complications. However, it may be overused for non-medical reasons in high-resource settings. Caesarean section is also underused in resource constrained settings due to low rates of facility births, lack of health infrastructure and shortage of skilled birth attendants.¹³² Both low and high rates of caesarean section have been associated with increased risks of maternal and newborn adverse health outcomes.¹³²

Given this concern, several studies³⁵⁻⁵¹ have been conducted to examine the relationship between caesarean section rates and neonatal mortality. However, the results of these studies conducted using aggregate- and individual-level data are inconsistent. For example, Molina *et al.*³⁵ in their ecological study conducted using worldwide country-level data found that caesarean section was associated with lower neonatal mortality. However, in another ecological study using data from 159 countries, it was shown that there is no association between caesarean section rates and neonatal mortality, where caesarean section rates were higher than 10%.⁴⁴ This finding may indicate that the benefit of caesarean delivery to lower maternal and neonatal mortality was confined to those who actually require the procedure medically.

Inconsistent results for the association between caesarean birth and neonatal mortality were also reported by different studies based on individual-level data conducted in Africa, Latin America, Asia and United States of America.³⁷⁻⁴² For instance, Villar *et al.*³⁸ in their large

prospective cohort study based on WHO global survey conducted in 410 health facilities in 24 areas in eight Latin American countries found that, with cephalic presentation, both intrapartum and elective caesarean section were associated with 1.66 (95% CI, 1.26 to 2.20) and 1.99 (95% CI, 1.51 to 2.63) times higher odds of neonatal mortality up to hospital discharge, respectively. Conversely, another study using data from the same WHO global survey completed in nine countries in Asia found that both pre-labour (aOR, 0.2; 95% CI, 0.1 to 0.3) and intrapartum caesarean sections (aOR, 0.3; 95% CI, 0.2 to 0.4) were associated with improved perinatal outcomes following breech presentation.³⁷ The findings from these two studies may suggest that the relationship between caesarean section and perinatal outcomes is inconsistent due to the possible regional differences.

It is clear from the existing literature that most of the previous studies regarding the association between caesarean section rate and neonatal mortality rate are at the aggregatelevel so that they are comparing changes acting on the population as a whole (i.e., a change in caesarean section rates and neonatal mortality among geographic areas).^{35,36,44-51} Many of these studies,^{45,47-50} however, failed to adjust for socioeconomic position, which can affect the likelihood of having caesarean section and neonatal mortality.

In contrast, studies based on individual-level data are more likely to report a considerably increased risk of neonatal mortality associated with caesarean versus vaginally born infants. There seems to be no study that explored why the risk of neonatal mortality was increased following caesarean delivery using individual-level data. Moreover, the previous studies are limited by either inconsistent results or failure to account for contextual factors such as unequal access, infrastructural, structural and health workforce constraints that could play a role in the association between caesarean section and neonatal mortality, particularly in

LMICs. A further study is needed to understand the association between caesarean section and neonatal mortality by taking into account contextual factors from the perspective of developing country settings to better inform the public health policy for intervention.

2.2.5 Caesarean section and breastfeeding

Given the rising proportions of labour induction and caesarean section, there is a concern that interventions during labour and childbirth may influence breastfeeding. It is well known that the first hours after birth are critical for establishing mother-infant interaction and breastfeeding.^{133,134} The timing of the first breastfeeding is also a key determinant for child and maternal health. For example, in a large 2016 study (n= 99,938) using pooled data from three randomised trials conducted in Ghana, India and Tanzania, it was found that initiation of breastfeeding within one hour after birth reduces neonatal and early infant mortality.¹³⁵ The study suggested that early initiation of breastfeeding, it is but also by additional mechanisms. The additional mechanisms are probably related to some of the biologically plausible explanations such that early initiation of breastfeeding directly reduces the use of prelacteal feeds, guarantees that the newborn receive 'colostrum' (the first breastmilk) and promotes intestinal maturation as well as some immunological protection.¹³⁶⁻¹³⁸

However, breastfeeding experiences of women who underwent caesarean section can be influenced by several factors, which include timing of initiation of breastfeeding (due to delays in maternal-infant interaction), mothers' emotional reactions to the surgery as well as infants' health and behavior. For instance, as the postoperative pain may be severe, particularly in the first 24 hours, it may restrict maternal mobility and influence breastfeeding.^{139,140} Likewise, due to the potential for physical separation of the mother and

infant after caesarean delivery, given the higher risk of infant admission to NICU as a consequence of respiratory problems, there might be a lesser likelihood of timely initiation of breastfeeding.¹⁴¹

It is hypothesised that the hormonal pathway that stimulates the onset of lactation (lactogenesis) may be disturbed by caesarean section due to maternal stress and/or decreased oxytocin secretion.¹⁴²⁻¹⁴⁵ This, in turn, may delay milk production. It has been supposed that maternal stress may interfere with the release of oxytocin, a hormone that results in milk ejection (let-down reflex). Maternal stress may also affect the level of prolactin, a hormone involved in lactation.¹⁴⁵ Similarly, newborns who were exposed to different level of stress due to obstetrical interventions during labour and birth, such as caesarean section may have disorganised sucking skills and low level of alertness, making the newborns unable to attach and suckle effectively at the breast.¹⁴⁵ These conditions impair lactogenesis and may result in physiological delay in milk secretion and shortening the duration of breastfeeding.⁵² This, in turn, has direct impact on microbiota development because human milk oligosaccharides are thought to shape the diversification of the infant intestinal microbiota.⁵⁵

There are studies which have been conducted to investigate the relationship between caesarean section and breastfeeding outcomes. However, because of inconsistent findings, small sample sizes, use of effect estimates from unadjusted analysis and lack of clarity on how breastfeeding outcomes were measured (or not using standard definition of breastfeeding outcomes), the results of these previous studies are less clear. For instance, Prior *et al.*³³ in their 2012 systematic review and meta-analysis involving 53 studies from 33 low-, middle-, and high-income countries (with no data from African countries) found that early breastfeeding (i.e., any initiation or breastfeeding at hospital discharge) was lower among

infants delivered by caesarean section (OR, 0.57; 95% CI, 0.50 to 0.64) when compared to infants born vaginally. Nevertheless, according to Prior *et al.*, breastfeeding as an outcome of interest by mode of delivery was not reported as a primary outcome in 74% (39 of 53 studies) of the studies included in their systematic review and meta-analysis. Their meta-analysis also showed a high level of heterogeneity between studies, but they have attempted to investigate the sources of heterogeneity by doing meta-regression and subgroup analysis. Moreover, the majority of the studies included in their meta-analysis did not collect data on potential confounders and reported unadjusted estimates. Under these circumstances, the association between caesarean section and early initiation of breastfeeding is unclear.

Liston and colleagues,¹⁴⁶ in their a 15-year cohort study that examined the impact of caesarean section on neonatal outcomes in Nova Scotia, found that compared with spontaneous vaginal delivery, breastfeeding at hospital discharge was lower in both caesarean section without labour (aOR, 0.64; 95% CI, 0.60 to 0.67) and caesarean section in labour (aOR, 0.79; 95% CI, 0.77 to 0.83). This finding suggests that the negative association between caesarean section and breastfeeding at discharge appears to be stronger in pre-labour caesarean section than caesarean section in labour (emergency). As planned caesarean births (pre-labour caesarean) are usually performed prior to the onset of labour, there is a possibility that decreased early initiation of breastfeeding may be attributed to physiological reasons. In a study conducted in Canada, it was found, compared to women who gave birth vaginally, that women who gave birth following planned caesarean section have a 61% higher odds of early cessation of breastfeeding (i.e., having breastfed for 12 weeks or less) (OR, 1.61; 95 % CI, 1.14 to 2.26).¹⁴⁷ Moreover, several other studies that compared the effect of caesarean section (either emergency or planned) versus vaginal delivery have also found that caesarean section influences breastfeeding outcomes.¹⁴⁸⁻¹⁵¹ However, the results of some studies

comparing the effect of caesarean section on breastfeeding outcomes are inconsistent (i.e., some studies reported no association while the other reported negative associations between caesarean section and breastfeeding outcomes). The inconsistent findings from the previous studies are may be due to suboptimal control of confounders and other biases. Further research, particularly using rigorous study design and standard breastfeeding outcome definitions, is needed to understand the impact of caesarean section on breastfeeding outcomes.

2.2.6 Caesarean section and school outcomes

By linking the routine perinatal data with children's school achievement data, MacKay *et al.* conducted a study in 2010 aimed at examining the need for special education across the full range of gestation.¹¹¹ They noted that risk for special education need decreased as gestational age is increased. For instance, the odds ratio for special education need at 37-39 weeks was 1.16 (95% CI, 1.12 to 1.20) while the odds ratios at 33–36, 28–32 and 24–27 weeks were 1.53 (95% CI, 1.43 to 1.63), 2.66 (95% CI, 2.38 to 2.97), and 6.92 (95% CI, 5.58 to 8.58), respectively.¹¹¹ The need for special education is defined as a learning difficulty that necessitates educational support for children born, specifically, at 37-39 weeks of gestation. As planned caesarean section are usually performed from 37-39 weeks of gestation,¹⁵² it would be important to investigate the impact of caesarean section on children's educational outcomes. Moreover, as the intrauterine brain maturation continues until the end of gestation, elective caesarean section may interrupt this process and this may has adverse effects on neurodevelopment in children's later life.¹⁵²

It has also been supposed that gestational age contributes to the microbial composition of the infant's gut. For instance, by using 16S rRNA amplicon pyrosequencing technology, Barret

et al. (2013) ¹⁵³ found that the microbiota of preterm infants lack two of the main bacterial genera (i.e., *Bifidobacterium* and *Lactobacillus*) usually present in full-term infants. Thus, gestational age seems to be relevant because it may influence the neurodevelopment of infants through alteration in the composition of microbiota.

Curran *et al.* in their large 2017 study (n = 1,489,925) using data from Swedish medical birth register and national school register revealed that elective caesarean section (aOR, 1.06; 95% CI, 1.03 to 1.09) and emergency caesarean section (aOR, 1.12; 95% CI, 1.09 to 1.15) were associated with children's poor school achievement.¹⁵⁴ However, by considering the complex nature of the relationship between caesarean section and cognitive outcome, they stated that their findings should be interpreted carefully. Moreover, in a 2016 study involving 153,730 infants born at \geq 32 weeks of gestation in New South Wales, Australia, it was found that children born following caesarean section had a 14% higher risk of being classified as developmentally vulnerable at school entry when compared to children born by spontaneous vaginal delivery (aRR, 1.14; 95% CI, 1.06 to 1.22).¹⁵⁵

Using data from the Longitudinal Study of Australian children (n = 3,666), Polidano *et al.*¹⁵⁶ in their 2017 study examined the differences in child cognitive performance among caesarean versus vaginally born children. They found that children born following caesarean section performed lower by up to 1/10 of a standard deviation in numeracy test scores at age 8–9 compared with vaginally born children. However, another 2016 study (n= 3,609) using whole-of-population administrative data from South Australia by Smithers *et al.*¹⁵⁷ found no association between elective caesarean birth and children's poor school performance as measured by Australian National Assessment Program—Literacy and Numeracy (NAPLAN) tests at age eight.

In summary, the effect estimates of the studies that examined the association between caesarean section (compared with vaginal birth) and children's educational or cognitive performance range from no association to little association (i.e., a <15% increased risk of poor school performance). The different results reported by these studies may be due to suboptimal control of confounders and the little effects observed may be explained away by residual confounding.

2.3 Apgar scores

The Apgar score was devised and introduced in early 1950s by Dr Virginia Apgar as an overall measure of a baby's vitality at one minute after birth.¹⁵⁸ A five-minute Apgar score was introduced later as it had been shown to be a better predictor of neonatal survival rather than one-minute Apgar score.¹⁵⁹ For decades, the Apgar score has been routinely reported at one and/or five minutes for all newborns in almost every country in the world. The Apgar score is also reported every five minutes up to 20 minutes if the five-minute Apgar score is less than 7.¹⁶⁰ The Apgar score comprises five physiological signs, including heart rate, respiratory effort, reflex irritability, muscle tone and skin colour. A rating of 0, 1 or 2 is given for each sign, reflecting whether it was absent, present but not adequate, or normal.^{160,161} These result in a total score that ranges from 0 to 10.

Apgar scores can also be used as a physiological indicator of how well a newborn is making the transition from *in-utero* to the *ex-utero* environment, where the oxygen concentration is about 21%.¹⁶² For example, the skin color is evaluated to determine the perfusion of the skin with oxygenated blood; heart rate reflects the success in the delivery of oxygen to organs; reflex irritability can be used as a marker for nervous system integration; muscle tone (activity) represents the locomotor capacity of the newborn and respiratory rate represents pulmonary function, which facilitate the neuro-feedback-mediated effort to breathe.¹⁶² **Figure 2.3** shows where Apgar score falls in the course of birth-related events.





Source: Adapted and redrawn from Bovbjerg et al (2019).¹⁶³

2.3.1 Apgar scores and short-term perinatal outcomes

There are several previous studies that investigated the association between Apgar scores and short-term perinatal outcomes. For example, Razaz and colleagues¹⁶⁴ in their 2019 study included 1,551,436 singleton infants, with Apgar score of \geq 7 at 1, 5 and 10 minutes in Sweden to examine the association between Apgar scores and neonatal morbidity and mortality. They found that, compared with infants with Apgar score of 10, the risks of neonatal morbidity and mortality among infants with low Apgar score values within the normal range (7-10) is high, particularly at 5 and 10 minutes.

By linking routine discharge and mortality data for all births in Scotland, UK between 1992 and 2010, Iliodromiti *et al.*¹⁶⁵ analysed data on 1,029,207 live singleton births between 22 and 44 weeks of gestation to examine the strength of the association between five-minute Apgar score and the risk of neonatal and infant mortality. They found that low five-minute

Apgar score (0-3) was linked with increased risk of neonatal and infant death across all gestational age. For instance, compared with normal Apgar score (7-10), a low Apgar score (0–3) was associated with aRR of 359.4 (95% CI, 277.3 to 465.9) for early neonatal death whereas it was 30.5 (95% CI, 18.0 to 51.6) and 50.2 (42.8 to 59.0) for late neonatal death and infant death, respectively. For intermediate Apgar score (4-6), Iliodromiti *et al.* also noted similar associations, but with lower magnitude.¹⁶⁵

Furthermore, several other previous studies¹⁶⁶⁻¹⁶⁹ have shown that Apgar scores of <7 were related to morbidity, including neonatal respiratory distress, meconium aspiration, hypoxic-ischaemic encephalopathy and infant mortality. It has been suggested that the association between Apgar score of <7 and neonatal and infant mortality is mostly attributed to infections or anoxia.¹⁶⁵ It is clear from the existing literature that low Apgar score can predict neonatal morbidity and mortality. This may have important implications for obstetric practice and research.

2.3.2 Apgar scores and children's educational outcomes

Two large studies published in 2008 and 2011 involving 176,524 and 19,559 children in Sweden ^{170,171} have demonstrated that low Apgar score (i.e. <7) is associated with poorer cognition or school achievement. For example, compared with infants with Apgar scores \geq 7 at 0-5 minutes, infants with low Apgar score (<7) had 1.14 times higher odds of low intelligence quotient (IQ) (OR, 1.14; 95% CI, 1.03 to 1.27).¹⁷⁰

After a study of literature regarding the association between the value and duration of Apgar score and death or neurologic disability across 17 studies, Ehrenstein (2009)¹⁷² observed dose dependant patterns between the value and duration of low Apgar score and neurologic

disability and death among infants. Nonetheless, less than 5% risk for neurologic disability was observed.¹⁷² In another 2009 study in Sweden, compared with infants with Apgar scores of \geq 7 at five-minutes, infants with low Apgar score had a four-fold higher risk of neurologic disability (PR, 4.0; 95% CI, 2.2 to 7.2).¹⁷³

The majority of the available evidence regarding the association between Apgar scores and IQ comes from studies conducted by linking birth data with routine IQ test results at conscription around age 18.^{170,173,174} Moreover, almost all of the previous studies compared Apgar scores dichotomised at the familiar cut-points (i.e., <7 or <4). There seems no study that investigated whether Apgar score when left as a quasi-continuous measure (i.e., each of the Apgar score values (compared with 10)) is associated with children's educational outcomes. As each total value of the Apgar score may indicate different physiological conditions, it would be important to examine the associations between Apgar scores when left as quasi-continuous measure and children's later neurodevelopmental outcomes.

2.4 Research justification

Given the concern that the rates of obstetrical interventions during labour and birth such as induction of labour and caesarean section have been substantially increasing globally, there is growing interest in research examining the short-and longer-term effects of obstetrical interventions during labour and birth. This literature review has identified several studies regarding the effect of obstetrical interventions on short-and longer-term outcomes.

However, the studies reviewed in this chapter have highlighted a number of gaps. First, much of the available evidence regarding the impact of obstetrical interventions during labour and birth on short-and long-term outcomes were generated from high-income countries. Second, although the short-term effect of obstetrical interventions (more specifically induction of labour) are well described in the literature, studies investigating longer-term effects of elective labour induction at 39 weeks versus expectant management on children's educational outcomes have not been available.

Third, despite reporting on the associations between caesarean section and maternal and neonatal mortality, previous studies are limited in providing interpretations of what the findings mean using contextual models (i.e., by considering contextual factors such as unequal access, infrastructural, structural and health workforce constraints that could play a role in the association between caesarean section and neonatal mortality), particularly in LMICs, where there is less evidence. Finally, inferences from several previous studies appear to be limited methodologically. For instance, many of the previous studies are hampered by non-longitudinal study design, small sample sizes and inconsistent findings, which may be due to suboptimal control of confounders and other biases.

This thesis helps fill these research gaps and uses data from low-, middle- and high-income countries. The thesis reports the design and results of five studies using data from 33 sub-Saharan African countries (LMICs) and Australia (a high income country). The first two studies present results of the analyses concerning caesarean section in Ethiopia. The next study presents novel analyses examining caesarean section and breastfeeding in sub-Saharan Africa. The final two studies add to knowledge about Apgar scores and children's educational outcomes, and labour induction and children's education outcomes using high-quality whole-of-population linked administrative data from South Australia.

Specifically, the studies investigated:

- The prevalence and sociodemographic characteristics of caesarean section in Ethiopia;
- The temporal association between caesarean birth and neonatal death within the context of Ethiopia from 2000 to 2016 and the interpretation of the association using the 'Three Delays Model';
- The impact of caesarean section on breastfeeding indicators—early initiation of breastfeeding (within 1 hour), exclusive breastfeeding under 6 months and children ever breastfed (at least once)—in 33 sub-Saharan African countries;
- 4. The associations between Apgar scores and children's educational outcomes at eight years of age in South Australia
- The effect of elective labour induction at 39 weeks versus expectant management on children's educational outcomes at eight years of age.

Chapter 3: Methodology

The five research questions of this thesis, a review of the literature, and research justification were presented in Chapters 1 and 2. This chapter describes: (1) the two data sources used in this thesis, (2) the organisation of information on the exposure (measures of obstetrical interventions during labour and birth), the outcome and the confounders from the two data sources, and (3) the statistical analytic approaches used to answer the five research questions.

3.1 Data sources

Data for this thesis were drawn from the Demographic and Health Surveys (DHS) and the South Australian Early Childhood Data Project (SAECDP). The DHS are widely available high-quality nationally-representative household surveys in more than 90 low- and middleincome countries (LMICs). The SAECDP is an established project that encompasses highquality whole-of-population linked administrative data from state and federal sources in South Australia. In this thesis, the DHS data of 33 LMICs in sub-Saharan Africa and the SAECDP data from Australia (high-income country) were utilised to answer the five research questions related to how obstetrical interventions during labour and birth affect breastfeeding, neonatal health and children's educational outcomes. The two data sources encompass datasets that span low-, middle-, and high-income country contexts. **Figure 3.1** shows countries (highlighted in blue) where the data from the two data sources were drawn for this thesis.



Figure 3.1 Countries (highlighted in blue) where the data for studies included in the thesis were drawn.

Table 3.1 summarises the research questions addressed in the thesis and the data sources used to answer each research question. The use of two different data sources (i.e., DHS and SAECDP) allowed this thesis to capture the effects of obstetrical interventions during labour and birth on breastfeeding, neonatal mortality and children's educational outcomes across diverse health system resource settings. Both DHS and SAECDP contain rich information on a range of maternal, perinatal and sociodemographic characteristics as well as information on a range of maternal and child health outcomes. Moreover, the SAECDP contains rich information on children's longer-term neurodevelopmental outcomes (e.g., ~8-year-old children's school assessments). The detailed descriptions regarding the design, sampling and methodology of each data source are provided below.

Table 3.1 Data sources used to examine each research aim.

Research aim		Data source
1.	To estimate the prevalence and examine sociodemographic	DHS
	factors associated with caesarean section in Ethiopia.	
2.	To examine the changing temporal association between	DHS
	caesarean birth and neonatal death in Ethiopia from 2000 to	
	2016 as well as to provide an interpretation of the associations	
	using the 'Three Delays Model' in the context of Ethiopia.	
3.	To investigate the effect of caesarean section on breastfeeding	DHS
	indicators in each of the 33 countries in sub-Saharan Africa, as	
	well as to summarise the magnitude of these within-country	
	effects in an overall estimate using random-effects meta-	
	analyses.	
4.	To examine the effect of Apgar scores of 0-5, 6, 7, 8 and 9	SAECDP
	(compared with 10) on children's educational outcomes at	
	eight years of age.	
5.	To estimate the effect of elective induction of labour at 39	SAECDP
	weeks of gestation as compared with expectant management	
	on children's educational outcomes at eight years of age.	

3.1.1 The Demographic and Health Surveys (DHS)

The DHS are nationally-representative surveys that are conducted at household level to collect data on health topics, including maternal and child health, nutrition, environmental health, malaria, HIV/AIDS and reproductive health in over 90 LMICs. The DHS began in 1984 funded by USAID (U.S. Agency for International Development).¹⁷⁵ In order to allow comparison over time, the DHS surveys are usually conducted approximately every five years. The DHS are implemented by a single national implementing agency such as a National Statistical Office, a Ministry of Health or a university in LMICs. For instance, in Ethiopia, the DHS was implemented by the Central Statistical Agency of Ethiopia.

The DHS study design, sampling and methodology were standardised and replicable with quality assurance measures. For instance, each of the DHS surveys involved two-stage, stratified clustered probability sampling design.¹⁷⁵ The samples were drawn from an existing sampling frame (commonly the latest national census). The DHS uses stratification to minimise sampling errors and samples are stratified by geographic region and by urban/rural areas within each region. The samples are designed and selected independently within each stratum. In the first stage of selection, the primary sampling units (PSUs)—usually the national census enumeration areas—are selected with probability proportional to size within each stratum. The PSU makes up the survey cluster. In the second stage, a complete list of households was conducted for each cluster. Subsequently, a fixed number of households—about 25-30 households per cluster—were selected by equal probability systematic sampling in the selected cluster. The total number of households (sample size) included in each round of DHS in LMICs were generally between 5,000 and 30,000 households.

All women aged 15-49 years old and all men aged 15-54(59) years old who were either permanent residents of the selected households or visitors who stayed in the household the night before the survey were eligible for face-to-face interviews. The DHS use a 'standard model questionnaires' to collect data that are comparable across countries. Moreover, the DHS also have optional 'questionnaire modules' for special information on topics that are not contained in the model questionnaires. Each country adopts and applies the model questionnaires although they can also add some questions of their particular interest to model questionnaires. In many surveys, a household, women's, and men's questionnaires were completed. Data on children aged 0-59 months in each household were also collected. Some DHS standard surveys also collected additional data including biomarkers (e.g., anthropometry, anaemia testing, and HIV testing) and geographic information (e.g., GPS

data). Data were collected by local field teams, electronically entered, manually and automatically cleaned, and made publicly available by 'The DHS Program' (https://dhsprogram.com/Data/).

Each of the DHS surveys protocols were approved by Institutional Review Board (IRB) of each country as well as by the ICF IRB. The DHS Program required an application to obtain permission to use the DHS data for research purposes. Data approval was sought and granted for use of the data from sub-Saharan African countries by DHS administrators for this thesis. The detailed information regarding the methodology and standards for protecting the privacy of study participants in all DHS can be accessed from The DHS Program online (https://dhsprogram.com/What-We-Do/methodology.cfm).

3.1.2 The South Australian Early Childhood Data Project (SAECDP)

The SAECDP is an innovative population-based linked administrative research database in South Australia. The project encompasses more than 30 different government administrative datasets from state and federal sources, and include all children in South Australia who were born from 1999 to 2013.¹⁷⁶ The advantage of the SAECDP is that it has rich information on child health and development from the perinatal period into adolescence, which is important to investigate the short- and long-term outcomes in children. The datasets held by the SAECDP are diverse and includes¹⁷⁶ multiple datasets from the Births and Deaths Registry, Perinatal Statistics, the Birth Defects Registry, Public Hospital Inpatient and Emergency Presentation data, Congenital Abnormality data, Child Health Records, Family Home Visiting Program, the Edinburgh Postnatal Depression Scale, Postnatal Risk Questionnaire, Childhood Immunisations, Neonatal Hearing, Public School Enrolment Census, Reading achievement levels, English as an Additional Language/Dialect, the National Assessment Program—Literacy and Numeracy (NAPLAN), the Australian Early Development Census, Child Protection, Youth Justice and Public Housing.

De-identified linkage keys were generated by an independent data linkage agency (SA-NT DataLink) using basic personal identifiers such as name, address, date of birth and sex. All the SAECDP datasets are supplied in de-identified form. In this thesis, the Perinatal database and NAPLAN datasets were used and the de-identified linkage keys were used to merge these datasets.

Ethical approval for the SAECDP project was already obtained from South Australian Department for Health and Ageing Human Research Ethics Committee and the use of data from the SAECDP for this thesis was approved by the South Australian Department for Health and Ageing (HREC/13/SAH/106/AM08).

3.2 Exposures: - measures of obstetrical interventions during labour and birth

Consistent with epidemiologic literature, statistical analyses in this thesis were organised in three different sets of variables—the exposure (treatment), the outcome and the confounders. The exposures and outcomes were determined by the five research questions related to how obstetrical interventions during labour and birth affect breastfeeding, neonatal mortality and children's educational outcomes. The exposures examined in this thesis included—caesarean section (Studies 2 and 3), Apgar scores (Study 4) and elective induction of labour at 39 weeks of gestation (Study 5).

3.2.1 Caesarean section

The data on caesarean section was captured in DHS based on mother's self-report. For example, the self-reported data on caesarean section were collected by asking mothers a question that reads, 'Was (NAME) delivered by caesarean section, that is, did they cut your belly open to take the baby out?' in the 2016 Ethiopian DHS.¹¹⁶ A reliability study conducted by Stanton *et al.*,¹⁷⁷ demonstrated that the recall of caesarean section in DHS was good in developing countries. Singleton last born infants delivered by caesarean section versus infants born vaginally were used as exposure in research questions 2 and 3 (Chapters 5 and 6).

3.2.2 Apgar scores

In the perinatal database held by the SAECDP, both 1 and 5 minute total Apgar scores were captured. Scores on each of the five Apgar score components—heart rate, respiratory effort, reflex irritability, muscle tone and skin colour—were not collected. Infants with a range of both 1 and 5 minutes Apgar scores (i.e., 0-5, 6, 7, 8 and 9 (compared with 10)) were served as the exposure of interest in research question 4 (more details in Chapter 7). The grouping of Apgar scores 0-5 was done because the sample size for each Apgar score within this group (0-5) was very small.

3.2.3 Elective induction of labour at 39 weeks of gestation

In South Australia, the perinatal database included information about onset of labour, reason for induction of labour and gestational age at birth. Gestational age (at birth) dating was determined predominately based on early dating ultrasound. However, in the small number of women when dating ultrasound was not available, gestational age was determined from the first day of the last menstrual period and estimates at birth. Based on these three variables onset of labour, reason for induction of labour and gestational age—'elective induction of

labour at 39 weeks of gestation' variable was computed and used as the exposure of interest in research question 5 (Chapter 8). The comparison (unexposed) group for elective induction of labour at 39 weeks of gestation was 'expectant management', which was defined as deliveries following spontaneous labour and elective induction of labour or medicallyindicated induction or caesarean from 40-42 weeks of gestation.

3.3 Outcomes

The outcome variables examined in this thesis included caesarean section (Study 1), neonatal death (Study 2), breastfeeding indicators—early initiation of breastfeeding, exclusive breastfeeding under 6 months and children ever breastfed—(Study 3), and children's educational outcomes (Studies 4 and 5).

3.3.1 Neonatal death

Neonatal death was defined as infants who were born alive, regardless of gestational age at birth, but died within the first 28 days of life. The DHS collected information from mothers whether their all biological children were alive or dead at the time of interview. Mothers reported the age of all children alive and age at death for deceased children. Thus, neonatal death, as a binary outcome variable, was measured from two variables (whether the child is alive and age at death (in days)). Neonatal death was used as an outcome of interest in research question 2 (Chapter 5).

3.3.2 Breastfeeding indicators

In each DHS, respondents were asked a number of questions regarding breastfeeding practices, the length of breastfeeding, about children's consumption of liquids and solid food and micronutrient supplementation. Breastfeeding questions captured in DHS included whether the child was ever breastfed; how long after birth before the newborn was put to the breast; whether the baby was given anything to drink or eat other than breast milk; the type of drink given to the child and whether the child was still breastfeeding. Based on this information, a standard set of indicators for assessing breastfeeding practice, which are consistent with the WHO and partners definitions,¹⁷⁸ were defined. These included 'early initiation of breastfeeding', 'exclusive breastfeeding under 6 months' and 'children ever breastfed'.

According to the WHO and partners definitions of indicators for assessing infant and young child feeding practices,¹⁷⁸ 'early initiation of breastfeeding' is defined as being put to the breast within one hour of birth (i.e., the proportion of children born in the last 24 months who were put to the breast within an hour of birth). 'Exclusive breastfeeding under 6 months' was defined as the percentage of infants aged 0-5 months who were consumed only breast milk (no other liquids or foods). 'Children ever breastfeed' is the proportion of children born in the last 24 months were used as outcome of interest in research question 3 (Chapter 6).

3.3.3 Children's educational outcomes

In Australia, children's educational outcomes are assessed by the National Assessment Program—Literacy and Numeracy (NAPLAN) tests at year (grade) 3, 5, 7 and 9 annually. The NAPLAN gives an opportunity to assess how children are making progress against the national standards in literacy and numeracy in all year levels (i.e., from year 3 to year 9). NAPLAN has five test domains including reading, writing, spelling, grammar and numeracy. For each domain, NAPLAN results are reported as both direct scores (range from 0 to 1000) and grouped into proficiency bands (range from 1 to 10). For each year level, the reported

proficiency bands are different. For instance, for year 3, the reported proficiency bands range from 1–6 while for year 5, 7 and 9, the reported proficiency bands range from 3-8, 4-9 and 5-10, respectively. The proficiency bands are used to identify children's educational outcomes relative to the National Minimum Standard (NMS), which is also dissimilar for each year level. For year 3, band 2 is the NMS, whereas bands 4, 5 and 6 are the NMS for year 5, 7 and 9, respectively.

The NAPLAN test data at year 3 (~8 years of age), which is administered by South Australian Department for Education were used as children's educational outcomes measures in research questions 4 and 5 (more details in Chapters 8 and 9). The NAPLAN test scores at year 3 (~8 years of age) were used in this thesis to maximize the eligible study population in South Australia. Consistent with national reporting of NAPLAN scores, this thesis reports results where NAPLAN was categorised into children performing at/below the NMS (\leq band 2) and above NMS (bands 3–6). However, the SAECDP did not include NAPLAN scores from students attending private schools as these have not been made available for linkage. Private school attendance accounted for about 34% of the population of year 3 students in South Australia from 2008 to 2015.¹⁷⁹
3.4 Confounding

Confounder selection in this thesis was based on the modern epidemiological definition, which is a common causes of the exposure and the outcome. The confounders were identified based on *a priori* subject-matter and expert knowledge, as well as through the use of Directed Acyclic Graphs (DAGs). It has been emphasised by a number of authors that confounder identification should be based on *a priori* subject-matter or expert knowledge (i.e., based on the understanding of the causal network linking the variables under study).¹⁸⁰⁻¹⁸³

DAGs are causal diagrams that visually encode assumptions about causal links among exposure, outcome and confounders (covariates).¹⁸⁴ DAGs have arrows that function unidirectionally to infer causality. **Figure 3.2** shows an example of a DAG representing Study 3 in this thesis. The use of DAGs help identify variables that can be used to block all backdoor paths between exposure and outcome. The variables identified and adjusted for in statistical analyses in each study to eliminate confounding were described in detail in Chapters 5-8.



Figure 3.2 DAG for the association between caesarean section and breastfeeding indicators.

3.5 Statistical Analysis

Table 3.2 shows the different statistical analysis approaches used in each study in this thesis.All statistical analyses were conducted using STATA/SE V.15.1 (StataCorp, College Station,TX, USA).

Table 3.2 Studies in the thesis and their statistical analysis approaches.

Stu	ıdy	A	nalytic approaches
1.	Caesarean section in Ethiopia: prevalence and	•	Log-Poisson regression
	sociodemographic characteristics		
2.	The changing temporal association between	•	Log-Poisson regression
	caesarean birth and neonatal death in Ethiopia:	•	Linear regression
	secondary analysis of nationally representative	•	Application of the 'Three Delays Model'
	surveys		
3.	Impact of caesarean section on breastfeeding	•	Modified Poisson regression
	indicators: within-country and meta-analyses of	•	Random-effects meta-analysis
	nationally representative data from 33 countries in		
	sub-Saharan Africa		
4.	Associations between Apgar scores and children's	•	Augmented Inverse Probability Weighted
	educational outcomes at eight years of age		(AIPW) estimator
		•	Negative control outcome
5.	Elective labour induction versus expectant	•	AIPW estimator
	management in women and children's educational		
	outcomes at eight years of age		

3.5.1 Log-Poisson regression/modified Poisson regression

Log-Poisson regression (i.e., using a generalised linear model with a log link function and a Poisson distribution for response)¹⁸⁵ was used to calculate prevalence ratios (PR) in Studies 1 and 2. Log-Poisson regression is an alternative to logistic regression, which directly yields an estimated incidence rate ratio (IRR) rather than odds ratio (OR). However, as the

mathematical computations for IRR is identical to the PR or relative risk (RR), the effect estimate from log-Poisson regression (i.e., IRR) was interpreted in terms of PR. It has been widely discussed in literature and was concluded that the RR (PR) is a preferred effect estimate over the OR.¹⁸⁶⁻¹⁸⁸ This is because the OR was found to overestimate the strength of associations (i.e., odds ratios overestimate relative risks when the outcome is common).¹⁸⁹ The other reason is when estimating OR, changing the reference category of a binary variable will produce 'reciprocal' estimates while that is not the case when estimating PR.¹⁸⁹ Nevertheless, when the outcome is rare (i.e., <10%), OR can be used because it is reasonably a good estimate of RR.

Moreover, it was demonstrated by Zou¹⁹⁰ that the RR (PR) can be estimated directly using a modified Poisson regression (i.e., Poisson regression with a robust error variance). This approach, which allows the RR to be estimated directly, was termed as the modified Poisson Regression by Zou. In Study 3, the modified Poisson regression models was used to calculate unadjusted and adjusted PR and their 95% CIs for each breastfeeding indicator—early initiation of breastfeeding, exclusive breastfeeding under 6 months and children ever breastfed—associated with caesarean section versus vaginal birth in each 33 country in sub-Saharan Africa. The formula for the modified Poisson regression is given as follows: ¹⁹⁰

$$\log[\pi(x_i)] = \alpha + \beta x_i$$

$$l(\alpha,\beta) = C.\sum_{i=1}^{n} [y_i(\alpha + \beta x_i) - \exp(\alpha + \beta x_i)]$$

3.5.2 Random-effects meta-analysis

In Study 3, random-effects meta-analyses were also conducted to summarise the evidence (estimate) from each of the 33 countries in sub-Saharan Africa regarding the association between caesarean section and breastfeeding indicators. Random-effects meta-analysis using inverse variance weighting was selected to account for between-country variation in effect estimates. Statistical heterogeneity was measured using I² and the possible sources of heterogeneity were also explored using *a post hoc* subgroup analyses (more details in Study 3).

3.5.3 Augmented Inverse Probability Weighted (AIPW) estimator

In Studies 4 and 5, the average treatment effects (ATEs) or risk differences were calculated using AIPW estimator.¹⁹¹ The AIPW estimator is also called a doubly-robust estimator.^{192,193} The AIPW estimator was chosen over traditional regression because it delivers a marginal rather than a conditional estimate, and it fits regression models for both the exposure and outcome providing a doubly-robust property if either the regression model for the outcome or the exposure is correctly specified.¹⁹²

In this thesis, the AIPW estimator fit logistic regression models for the exposures conditional on each confounder listed in Studies 4 and 5 to estimate propensity scores for each individual. The propensity scores were used to weight the observed data. The AIPW also fit logistic regression models for the outcomes conditional on all confounders for each category of the exposures to obtain predicted outcomes for each individual in Studies 4 and 5. After estimating the propensity scores and predicted outcomes, the AIPW estimator combines the

57

estimated values along with observed values to calculate the doubly robust estimates of outcomes for each level of exposure for each individual.

Finally, the mean of the estimates of outcome under exposed (e.g., elective induction of labour (Study 5)) and unexposed (expectant management) was calculated. These estimated means were used to calculate ATE, which is a mean difference in outcome among exposed and unexposed groups.

3.5.4 Negative control outcome

Sensitivity analysis via 'negative control outcome'¹⁹⁴—a tool for detecting confounding and other bias—was used to examine the potential for residual confounding in Study 4. The negative control outcome is a means of ruling out the possible non-causal interpretations of results in epidemiologic studies. In the negative control outcome analysis, the estimate of the associations between the exposure (e.g., Apgar score, in Study 4) and the outcome (e.g., children's school performance), as well as the estimate of the associations between the exposure (e.g., family possession of school card, in Study 4) were compared. If the associations are of similar magnitude, this suggests that the association between the exposure and the outcome could be due to residual confounding.

3.5.5 Application of the 'Three Delays Model'

The 'Three Delays Model' is a conceptual framework developed by Thaddeus and Maine to examine factors contributing to maternal mortality with a specific focus on those that affect the 'interval between the onset of obstetric complication and its outcome'.¹⁹⁵ The 'Three Delays Model' summarises the various factors that affect this interval in three phases of

delay—delay in deciding to seek care (phase I delay); delay in identifying and reaching medical facility (phase II delay); and delay in receiving adequate and appropriate treatment (phase III delay). Some of the key factors that shape the model include the status of women; distance from health facility; availability and cost of transportation; condition of roads; distribution of health facilities; shortage of supplies, equipment and skilled birth attendants and adequacy of referral system.¹⁹⁵ **Figure 3.3** shows the pictorial presentation of the 'Three Delays Model'. Study 2, in this thesis, was guided by the 'Three Delays Model' to facilitate the interpretation of the association between caesarean section and neonatal mortality by considering the contextual factors in Ethiopia.



Figure 3.3 The 'Three Delays Model'.

Source: Adapted and redrawn from Soc Sci Med, 1994; 38(8): 1091-110.¹⁹⁵

Chapter 4: Caesarean section in Ethiopia: prevalence and sociodemographic characteristics

4.1 Preface

This chapter contains the first of the five studies contributing to this thesis. This study was published in *The Journal of Maternal-Fetal & Neonatal Medicine* in 2017 and examines the prevalence and sociodemographic characteristics of caesarean section in Ethiopia.

This chapter utilised high-quality nationally representative data collected for Ethiopian Demographic and Health Surveys in 2000, 2005, 2011 and 2016 to comprehensively assess the prevalence and sociodemographic characteristics associated with caesarean section in Ethiopia from 2000 to 2016. Although a review of the literature revealed that there have been studies that report on caesarean section in Ethiopia, all previous studies focused on specific geographic area within Ethiopia at a particular point in time. Thus, this study addresses an important research gap and may have important policy and practice implications.

4.2 Publication

Yisma E, Smithers LG, Lynch JW, Mol BW. Cesarean section in Ethiopia: prevalence and

sociodemographic characteristics. J Matern Fetal Neonatal Med 2019; 32(7): 1130-1135.

https://doi.org/10.1080/14767058.2017.1401606

4.2.1 Statement of Authorship

Title of Paper	Caesarean section in Ethiopia: p	prevalence and sociodemographic characteristics
Publication Status	Published	Accepted for Publication
	Submitted for Publication	Unpublished and Unsubmitted w ork w ritten in manuscript style
Publication Details	Yisma E, Smithers LG, Lynch J prevalence and sociodemograph 2019; 32 (7): 1130-5. <u>https://doi</u>	W, Mol BW. Cesarean section in Ethiopia: hic characteristics. <i>J Matern Fetal Neonatal Med</i> horg/10.1080/14767058.2017.1401606

Principal Author

Name of Principal Author (Candidate)	Engida Yisma
Contribution to the Paper	Conceived and designed the study, performed the statistical analysis, interpreted the data, and drafted the manuscript.
Overall percentage (%)	85%
Certification:	This paper reports on original research I conducted during the period of my Higher Degree by Research candidature and is not subject to any obligations or contractual agreements with a third party that would constrain its inclusion in this thesis. I am the primary author of this paper.
Signature	Date 5-Aug-2020

Co-Author Contributions

By signing the Statement of Authorship, each author certifies that:

- i. the candidate's stated contribution to the publication is accurate (as detailed above);
- ii. permission is granted for the candidate to include the publication in the thesis; and
- iii. the sum of all co-author contributions is equal to 100% less the candidate's stated contribution.

Name of Co-Author	Lisa G Smithers		
Contribution to the Paper	Contributed to the design of the study, in the manuscript.	terpretec	l the findings, and reviewed
Signature		Date	5-Aug-2020

Name of Co-Author	John W Lynch
Contribution to the Paper	Contributed to the design of the study, interpreted the findings, and reviewed the manuscript.
Signature	Date 6/8/20

Name of Co-Author	Ben W Mol		
Contribution to the Paper	Contributed to the design of the stud the manuscript.	ly, interpreted	I the findings, and reviewed
Signature		Date	04/08/2020

LIBRARY NOTE:

This article has been removed due to copyright.

It is also available online to authorised users at: https://doi.org/10.1080/14767058.2017.1401606

Chapter 5: The changing temporal association between caesarean birth and neonatal death in Ethiopia: secondary analysis of nationally representative surveys

5.1 Preface

This chapter contains the second of the five studies contributing to this thesis. The study was published in *BMJ Open* in 2019 and examines the association between caesarean birth and neonatal mortality in the context of Ethiopia over time.

In this chapter, both individual- and aggregate-level data analyses were first performed to examine the association between caesarean section and neonatal death using DHS data collected over a period in which the availability of caesareans increased in Ethiopia. The chapter then applied the 'Three Delays Model' to facilitate the interpretation of the association between caesarean section and neonatal mortality both empirically and theoretically. The different analyses conducted helped to explain pathways and differentials in caesarean section use and its association to neonatal mortality in Ethiopia by considering contextual factors such as unequal access, infrastructural, structural and health workforce constraints that could play a role in the association between caesarean section and neonatal mortality. This investigation is important to inform policy and practice because policy makers and program planners may consider context-specific health interventions to decrease maternal and neonatal mortality.

5.2 Publication

Yisma E, Mol BW, Lynch JW, Smithers LG. The changing temporal association between

caesarean birth and neonatal death in Ethiopia: secondary analysis of nationally

representative surveys. BMJ Open 2019; 9(10):e027235. http://dx.doi.org/10.1136/bmjopen-

<u>2018-027235</u>.

5.2.1 Statement of Authorship

Title of Paper	The changing temporal associati Ethiopia: secondary analysis of	ion between caesarean birth and neonatal death in nationally representative surveys
Publication Status	Published	Accepted for Publication
	Submitted for Publication	Unpublished and Unsubmitted w ork w ritten in manuscript style
Publication Details	Yisma E, Mol BW, Lynch JW, S between caesarean birth and neo nationally representative survey http://dx.doi.org/10.1136/bmjc	Smithers LG. The changing temporal association onatal death in Ethiopia: secondary analysis of s. <i>BMJ Open</i> 2019; 9(10):e027235. open-2018-027235.

Principal Author

Name of Principal Author (Candidate)	Engida Yisma
Contribution to the Paper	Conceived and designed the study, performed the statistical analysis, interpreted the data, and drafted the manuscript.
Overall percentage (%)	85%
Certification:	This paper reports on original research I conducted during the period of my Higher Degree by Research candidature and is not subject to any obligations or contractual agreements with a third party that would constrain its inclusion in this thesis. I am the primary author of this paper.
Signature	Date 5-Aug-2020

Co-Author Contributions

By signing the Statement of Authorship, each author certifies that:

- i. the candidate's stated contribution to the publication is accurate (as detailed above);
- ii. permission is granted for the candidate to include the publication in the thesis; and
- iii. the sum of all co-author contributions is equal to 100% less the candidate's stated contribution.

Name of Co-Author	Ben W Mol		
Contribution to the Paper	Contributed to the design of the manuscript.	the study, interpreted	d the findings, and reviewed
Signature		Date	04/08/2020

Name of Co-Author	John W Lynch
Contribution to the Paper	Contributed to the design of the study, interpreted the findings, and reviewed the manuscript.
Signature	Date 6/8/20

Name of Co-Author	Lisa G Smithers		
Contribution to the Paper	Contributed to the design of the study, in the manuscript.	nterpretec	d the findings, and reviewed
Signature		Date	5-Aug-2020

BMJ Open The changing temporal association between caesarean birth and neonatal death in Ethiopia: secondary analysis of nationally representative surveys

Engida Yisma (),^{1,2} Ben W Mol,^{3,4} John W Lynch,^{1,4,5} Lisa G Smithers^{1,4}

To cite: Yisma E, Mol BW, Lynch JW, *et al.* The changing temporal association between caesarean birth and neonatal death in Ethiopia: secondary analysis of nationally representative surveys. *BMJ Open* 2019;**9**:e027235. doi:10.1136/ bmjopen-2018-027235

Prepublication history and additional material for this paper are available online. To view these files, please visit the journal online (http://dx.doi. org/10.113/6bmjopen-2018-027235).

Received 12 October 2018 Revised 09 July 2019 Accepted 28 August 2019

Check for updates

© Author(s) (or their employer(s)) 2019. Re-use permitted under CC BY-NC. No commercial re-use. See rights and permissions. Published by BMJ.

¹School of Public Health, The University of Adelaide, Adelaide, South Australia, Australia ²School of Allied Health Sciences, College of Health Sciences, Addis Ababa University, Addis Ababa, Ethiopia ³Department of Obstetrics and Gynaecology, Monash University, Melbourne, Victoria, Australia ⁴Robinson Research Institute, School of Medicine. The University of Adelaide, Adelaide, South Australia, Australia ⁵Population Health Sciences, University of Bristol, Bristol, UK

Correspondence to

Engida Yisma; engida.derbie@adelaide.edu.au

ABSTRACT

Objective To examine the changing temporal association between caesarean birth and neonatal death within the context of Ethiopia from 2000 to 2016.

Design Secondary analysis of Ethiopian Demographic and Health Surveys.

Setting All administrative regions of Ethiopia with surveys conducted in 2000, 2005, 2011 and 2016.

Participants Women aged 15–49 years with a live birth during the 5 years preceding the survey.

Main outcome measures We analysed the association between caesarean birth and neonatal death using log-Poisson regression models for each survey adjusted for potential confounders. We then applied the 'Three Delays Model' to 2016 survey to provide an interpretation of the association between caesarean birth and neonatal death in Ethiopia.

Results The adjusted prevalence ratios (aPR) for neonatal death among neonates born via caesarean section versus vaginal birth increased over time, from 0.95 (95% Cl: 0.29 to 3.19) in 2000 to 2.81 (95% Cl: 1.11 to 7.13) in 2016. The association between caesarean birth and neonatal death was stronger among rural women (aPR (95% Cl) 3.43 (1.22 to 9.67)) and among women from the lowest quintile of household wealth (aPR (95% Cl) 7.01 (0.92 to 53.36)) in 2016. Aggregate-level analysis revealed that an increased caesarean section rates were correlated with a decreased proportion of neonatal deaths.

Conclusions A naïve interpretation of the changing temporal association between caesarean birth and neonatal death from 2000 to 2016 is that caesarean section is increasingly associated with neonatal death. However, the changing temporal association reflects improvements in health service coverage and secular shifts in the characteristics of Ethiopian women undergoing caesarean section after complicated labour or severe foetal compromise.

INTRODUCTION

Globally, 2.6 million neonatal deaths occurred within the first 28 days after birth, which accounted for 46% of all under-five deaths in 2016.¹ The majority of these deaths were from low-income and middle-income countries. According to the United Nations Interagency Group for Child Mortality Estimation,

Strengths and limitations of this study

- This was the first study to examine the temporal association between caesarean birth and neonatal death within the context of Ethiopia from 2000 to 2016.
- A number of analyses conducted after adjustment for potential confounders helped develop the possible scenarios to better understand the interpretation of the changing associations.
- We have used additional supporting evidence from the 2016 Ethiopian Demographic and Health Survey data which allowed us interpret the association between caesarean birth and neonatal death in view of contextual factors in Ethiopia using the 'Three Delays Model'.
- Given the very low base rates of caesarean delivery in Ethiopia, the interpretation of our findings may not reflect the context of other low-income and middle-income countries.

Southern Asia (39%) and sub-Saharan Africa (38%) comprised the top two regions with the highest proportion of newborn deaths, while five countries (India, Pakistan, Nigeria, the Democratic Republic of the Congo and Ethiopia) accounted for 50% of all newborn deaths.¹ Evidence shows that, compared with mortality among children aged 1–59 months, neonatal mortality is decreasing more slowly.¹ If the current trend continues, more low-income and middle-income countries will fail to achieve the Sustainable Development Goal target for neonatal mortality at least as low as 12 deaths per 1 000 live births.¹²

In contemporary obstetric practice, caesarean section remains an important intervention in preventing neonatal mortality and other adverse birth outcomes.³ However, caesarean section may be prone to misuse because of unequal access, social and cultural factors.^{4–6} In developing country settings, due to limited medical provisions and/or lack of skilled birth attendants, some women may

not benefit from caesarean birth though they are medically eligible, while ineligible women may sometimes have increased access. In the last decades, caesarean section rates have been increasing in low-income, middle-income and high-income countries.^{7–9} The World Health Organization (WHO) suggests that 'every effort should be made to provide caesarean sections to women in need, rather than striving to achieve a specific rate'.¹⁰

Previous studies conducted using aggregate-level and individual-level data have yielded inconsistent results about the association between caesarean birth and neonatal mortality. For instance, two ecological studies^{11 12} conducted using worldwide country-level data have found that caesarean birth was associated with lower neonatal mortality, while another two ecological studies¹³¹⁴ showed no association between caesarean birth and neonatal mortality, where caesarean section rates were higher than 10%. Inconsistent results for the association between caesarean birth and neonatal mortality were also reported by different studies based on individual-level data conducted in Africa, Latin America, Asia and USA.^{15–20} For example, a large study conducted by Villar et al in 410 health facilities in 24 areas in eight Latin American countries found that, with cephalic presentation, both intrapartum and elective caesarean were associated with 1.66 (95% CI: 1.26 to 2.20) and 1.99 (95% CI: 1.51 to 2.63) times higher odds of neonatal mortality up to hospital discharge, respectively.¹⁶ However, another study based on WHO global survey completed in nine countries in Asia found that both prelabour (adjusted odds ratio (aOR) 0.2 (95% CI: 0.1 to 0.3)) and intrapartum caesarean sections (aOR 0.3, 95% CI: 0.2 to 0.4) were associated with improved perinatal outcomes following breech presentation.¹⁵

On the other hand, using both country-level and individual-level data collected for nationally representative Demographic and Health Surveys (DHS), Kyu et al found an increased risk for neonatal death associated with caesarean versus vaginal births in countries with low (<5%) and medium (5%-15%) caesarean section rates.²¹ However, factors associated with caesarean section that increase risk for neonatal death in countries with low and moderate caesarean section rates remain ill-defined. Previous studies are limited by either inconsistent results or lack the interpretation of findings by considering the contextual factors.¹¹⁻²¹ In addition to the underlying indications for caesarean interventions like 'fetal distress', 'cord prolapse', 'prolonged and obstructed labour', 'fetal mal-presentation', 'major antepartum haemorrhage' and 'placenta praevia',^{22 23} several contextual factors such as unequal access, infrastructural and health workforce constraints could play a role in the association between caesarean section and neonatal death.

In low-income and middle-income countries, the DHS are the most representative and widely available highquality data sources for studies related to maternal and child health. We use Ethiopian DHS data from 2000, 2005, 2011 and 2016 to examine the changing temporal association between caesarean birth and neonatal death. We then apply the 'Three Delays Model' developed by Thaddeus and Maine²⁴ to facilitate the interpretation of the association between caesarean birth and neonatal death in Ethiopia using the 2016 data.

METHODS

Study design and data samples

We used data from the Ethiopian DHS completed in 2000, 2005, 2011 and 2016. The Ethiopian DHS are nationally representative cross-sectional surveys conducted in nine regional states (Tigray, Affar, Amhara, Oromia, Somali, Benishangul-Gumuz, SNNPR, Gambela and Harari) and two city administrations (Addis Ababa and Dire Dawa). Each of the surveys involved a two-stage, stratified, clustered sampling design. The survey datasets are deidentified and made freely available online. Permission to use these data was granted by the DHS Program. The details about the methodology and standards for protecting the privacy of study participants in all DHS can be accessed online (http://www.dhsprogram.com/What-We-Do/methodology.cfm).

Exposure

The DHS questionnaire asks women about pregnancy, antenatal and delivery care for live births they have reported in the past 5 years. The data on caesarean section and other variables in the DHS were collected based on mothers' self-report. For example, the self-reported data on caesarean section were collected by asking mothers a question that reads, 'Was (NAME) delivered by caesarean section, that is, did they cut your belly open to take the baby out?' in the 2016 survey. Stanton and colleagues²⁵ in their study demonstrated that the DHS caesarean section rates, compared with facility-based records of caesarean section rates, are reliable for national and global monitoring in developing countries. For this study, the exposure group were infants delivered by caesarean section and unexposed group comprised infants born vaginally.

Outcome

Neonatal death includes infants who were born alive in the 5 years before each survey, but died within the first 28 days of life. The outcome variable, neonatal death, was measured from two variables (whether the child is alive and age at death (in days)).

Confounding

The following potential confounders were identified based on a priori subject-matter and expert knowledge. They included place of delivery (public, private, non-governmental organisation and home), type of residence (urban/rural), sex of child (male/female), size of baby at birth (very large, larger than average, average, smaller than average, very small and do not know), mother's age at birth (in years), mother's education (no education, primary, secondary and higher), birth order (1, 2–3 and 4+) and household wealth quintile (poorest, poorer, middle, richer and richest). The size of baby at birth was assessed based on mother's perception (estimate) of baby size at birth. It has previously been shown that, in the absence of complete enumeration of birth weight, mother's perception of baby size at birth can be used as a proxy to birth weight in nationally representative surveys.²⁶ Mother's age at birth was calculated as a difference (in years) between infant's date of birth and mother's date of birth. The DHS computes the wealth index for each survey based on household assets using principal components analyses²⁷ and categorises households into wealth quintiles. These asset-based measures represent the wealth distribution relative to other households within the country. They are widely used and are consistent with comparisons to household expenditures and the measurement of inequalities in child mortality, education and healthcare use in low-income and middle-income countries.²⁸

Statistical analysis

Missing information is uncommon in DHS because the data are collected by a trained interviewers at a face-toface interview. All analyses (ie, Ethiopian DHS 2000, 2005, 2011 and 2016) were weighted to be nationally representative. As women may have had more than one births within the 5-year survey periods, we also accounted for both clustering of caesarean deliveries within women as well as the complex survey design during the data analyses using the unit of analysis (ie, children) study number and sample weights. We then conducted both individual-level and aggregate-level analyses. Our 2016 data analysis was also supplemented by an application of the 'Three Delays Model' to interpret the association between caesarean birth and neonatal death both empirically and theoretically. All analyses were conducted using STATA/SE V.15.1 (Stata Corporation).

Individual-level analysis

Associations between caesarean birth and neonatal death at individual-level were analysed using log-Poisson regression models using data from Ethiopian DHS conducted in 2000, 2005, 2011 and 2016. We calculated unadjusted and adjusted prevalence ratios (aPR) and their 95% CIs for each survey. We have then compared the strength of association between caesarean birth and neonatal death across all surveys analysed.

After noting the increasing association between caesarean birth and neonatal death over time, we conducted a series of analyses to explore what was during the change. We used the 2016 data because the association was more pronounced. We first restricted the analysis to participants living in regions with the highest caesarean section rates to examine whether the increased access to caesarean section affected the proportion of neonatal deaths. We then estimated the effect of caesarean birth on neonatal death in regions with low caesarean section rate (ranged: 0.4%–5.3%) or where access to caesarean

section is limited, by excluding births in relatively high caesarean section rate regions—Addis Ababa (21.4%) and Harari (9.0%).²⁹ Both low-level and high-level of caesarean use has risks exceeding the risks of spontaneous vaginal deliveries.^{15 30} It was demonstrated that low levels of caesarean are related to lack of access and can contribute to maternal and newborn deaths.^{21 31}

Given the very large rural-urban differences in caesarean section rates in Ethiopia,^{29 32} we also conducted similar analyses separately for rural women. In addition, we evaluated the association by restricting the analyses to births from the lowest quintile of household wealth, births from the highest quintile of household wealth, and births in public health facilities separately. These alternative analyses were exploratory in nature and helped us understand contextual factors leading to inequalities in caesarean use that may occur not only due to inadequate access among the poorest women, but also due to overuse among the richest population subgroups.^{33 34} The subgroup analyses allowed us to explain how contextual factors such as unequal access, infrastructural and workforce constraints could play role in the association between caesarean section and neonatal death because these factors will result in delay in accessing emergency caesarean section, which is usually accessible at specialised health facilities.

The 2016 DHS included an additional question regarding 'timing of decision to conduct caesarean section (ie, whether it was before or after the onset of labour pains)'. We used this variable as a proxy to the types of caesarean birth (indicative of intrapartum or prelabour caesarean section) and conducted analysis to examine the association between types of caesarean section and neonatal death. As this was confined only to 2016 data, we have provided the results in online supplementary table A1.

Aggregate-level analysis

Data on the caesarean section rates and proportion of neonatal deaths were disaggregated by urban-rural areas for each of the nine regional states and two city administrations in Ethiopia for each of the surveys completed in 2000, 2005, 2011 and 2016. However, the urban-rural stratification for Addis Ababa is only available for the 2005 survey. These results in a total of 85 data points (observations). In order to assess the correlation between caesarean section and neonatal death at the aggregate level, we conducted simple linear regression for overall surveys together and for individual surveys separately.

Application of the 'Three Delays Model'

The 'Three Delays Model' is a conceptual framework developed by Thaddeus and Maine to examine factors contributing to maternal mortality with specific focus on those that affect the 'interval between the onset of obstetric complication and its outcome'.²⁴ The 'Three Delays Model' summarises the various factors that affect *this interval* into three phases of delay—delay in

deciding to seek care (*phase I delay*); delay in identifying and reaching medical facility (*phase II delay*); and delay in receiving adequate and appropriate treatment (*phase III delay*). Some of the key factors that shape the model include status of women; distance from health facility; availability and cost of transportation; condition of roads; distribution of health facilities; shortage of supplies, equipment and skilled birth attendants and adequacy of referral system.²⁴ The pictorial presentation of the 'Three Delays Model' is provided in online supplementary figures A1–A4.

As maternal and neonatal mortality share many risk factors, we adopted the 'Three Delays Model' as a framework to help interpret the association between caesarean birth and neonatal mortality within the context of Ethiopia using the 2016 survey because factors contributing to the 'three delays' aggravate the underlying medical indications for caesarean intervention that make neonatal death difficult to prevent. The 2016 survey was selected for interpretation of the association between caesarean birth and neonatal death using the 'Three Delays Model' because the association was more pronounced in the 2016 data. Previous studies conducted in India,³⁵ Tanzania³⁶ and Uganda³⁷ have applied the 'Three Delays Model' to their analyses of perinatal deaths.

We have identified some contributing factors underlying the 'Three Delays Model' from the 2016 survey. For example, information regarding problems faced by women of reproductive age (15-49 years) in accessing healthcare to obtain medical advice or treatment for themselves when they are sick were gathered. It consisted of four questions: distance to health facility (big problem/not big problem); getting money for treatment (big problem/not big problem); getting permission to go for treatment (big problem/not big problem) and not wanting to go alone (big problem/not big problem). Furthermore, data on skilled assistance during delivery, and women's socioeconomic and demographic status are also available in the DHS. This information can particularly be important to understand and address the barriers that women face in seeking care during pregnancy and delivery.³² We have, therefore, analysed the 2016 data to describe these factors empirically in the context of Ethiopia.

Patient and public involvement

This research was done without patient involvement in setting the research question or the outcome measures, and in the design and implementation of the study. No patients were asked to advise on interpretation or writing up of results. There are no plans to disseminate the results of this research to study participants or the relevant patient community.

RESULTS

Table 1 shows the characteristics of mothers and children according to mode of delivery for each of the surveys

conducted in 2000, 2005, 2011 and 2016. Across the four DHS waves, women who underwent caesarean delivery were more likely to live in urban areas, had a higher level of education and were from the richest quintile of household wealth. They were also more likely to have male children. Caesarean deliveries were more frequent in women in the age category of 20-29 years, and among infants who had either very large or larger than average size of baby at birth. Figure 1 shows that the proportion of institutional deliveries increased from 5.0% in 2000 to 26.3% in 2016, whereas the national caesarean section rate increased from 0.7% in 2000 to 1.9% in 2016. However, the rate of caesarean delivery in Ethiopia varied widely across administrative regions (figure 2). For instance, Addis Ababa had the highest (21.4%) rate, while Somali had the lowest (0.4%) in 2016. The national proportion of neonatal deaths decreased from 4.8% in 2000 to 2.9% in 2016 (figure 1), but the proportion varies among administrative regions of Ethiopia (online supplementary table A2).

Table 2 shows that the aPR for neonatal death associated with caesarean versus vaginal births in 2000 survey was 0.95 (95% CI: 0.29 to 3.19) while in 2005, it was 1.53 (95% CI: 0.52 to 4.50). In 2011, the aPR for neonatal death associated with caesarean versus vaginal births was 1.15 (95% CI: 0.45 to 2.93), while it was 2.81-fold higher risk of neonatal death (aPR, 2.81; 95% CI: 1.11 to 7.13) in 2016.

Table 3 summarises the findings of the subgroup analyses based on the 2016 data. When women living in urban settings—Addis Ababa (caesarean section rate (21.4%)) and Harari (9.0%)—were excluded from the analyses, the corresponding aPR for neonatal death was increased to 3.55 (95% CI: 1.31 to 8.56). Similarly, when we restricted the analyses to include only rural women, the prevalence ratio for neonatal death associated with caesarean versus vaginal births was found to be 3.43 (95% CI: 1.22 to 9.67). The respective risk of neonatal death increased to 7.01 (95% CI: 0.92 to 53.36) when the analysis was limited to women from the lowest quintile of household wealth.

When we restricted the analyses to Addis Ababa, the capital of Ethiopia, the relative risk for neonatal death associated with caesarean versus vaginal births was 1.07 (95% CI: 0.20 to 5.73). Moreover, when the analysis was confined to women from the highest quintile of the household wealth, the risk of neonatal death was 2.72 (95% CI: 0.55 to 13.38).

Finally, figure 3 shows that an increase in caesarean section rate is weakly correlated with a decrease in the proportion of neonatal deaths (correlation coefficient (r)=-0.1839) when aggregate-level data for all surveys together were analysed. However, the relationship between caesarean birth and neonatal death is variable when the analysis is restricted to each survey year separately (figure 4).

Interpretation

The primary individual-level analyses showed that the aPR for neonatal death associated with caesarean versus

Table 1 Characteristics	tof the study participar Mode of delivery	nts according t	o the mode of delivery, E	Ethiopia DHS 20	000, 2005, 2011 and 201	Q			0
	DHS 2000		DHS 2005		DHS 2011		DHS 2016		
	Caesarean (n=86) %	Vaginal (n=12174) %	Caesarean (n=111) %	Vaginal (n=11 052) %	Caesarean (n=175) %	Vaginal (n=11697) %	Caesarean (n=213) %	Vaginal (n=10810) %	
Neonatal death									
Yes	4.5	4.8	6.7	3.9	5.5	3.7	8.3	2.7	
No	95.5	95.2	93.3	96.1	94.5	96.3	91.7	97.3	
Types of residence									
Urban	76.2	9.9	68.8	6.7	70.9	12.0	60.4	10.1	
Rural	23.8	90.1	31.2	93.3	29.1	88.0	39.6	89.9	
Region									
Tigray	4.0	6.4	5.4	6.3	12.5	6.2	6.8	6.5	
Affar	2.5	1.0	0.6	1.0	1.4	1.0	0.4	1.0	
Amhara	3.7	26.3	15.2	23.6	19.4	22.4	22.1	18.7	
Oromia	46.3	40.7	28.5	39.6	14.6	42.6	21.2	44.5	
Somali	2.3	1.2	4.2	4.3	1.4	3.1	1.0	4.7	
Benishangul-Gumuz	3.1	1.0	0.1	0.9	1.0	1.2	0.6	1.1	
SNNPR	18.5	21.2	21.6	22.4	17.4	21.1	21.0	20.8	
Gambela	0.6	0.2	0.5	0.3	2.0	0.3	0.2	0.2	
Harari	0.9	0.2	0.7	0.2	1.2	0.2	. .+	0.2	
Addis Ababa	16.9	1.4	22.1	1.2	27.6	1.5	24.5	1.8	
Dire Dawa	1.1	0.3	1.2	0.3	1.4	0.3	1.2	0.4	
Mother's age at birth									
<20	19.4	12.0	13.3	13.4	10.5	10.9	6.4	10.0	
20-29	70.8	51.3	59.8	51.2	62.8	55.7	58.4	54.6	
30–39	9.8	30.0	24.1	29.3	23.7	28.7	31.7	30.9	
40-49	0.0	6.7	2.9	6.2	3.3	4.7	3.4	4.5	
Mother's education									
No education	15.7	82.5	30.1	79.7	19.9	70.0	22.8	66.9	
Primary	21.8	13.0	14.5	16.6	44.7	26.8	35.1	26.6	(
Secondary	56.4	4.3	46.5	3.4	22.2	1.9	15.3	4.5	Ope
Higher	6.2	0.2	8.9	0.3	13.2	1.2	26.8	2.0	en a
Place of delivery*									aco
Public	96.3	4.0	90.2	3.9	83.5	7.6	84.0	23.6	ces
								Continued	S

Yisma E, et al. BMJ Open 2019;9:e027235. doi:10.1136/bmjopen-2018-027235

BMJ Open: first published as 10.1136/bmjopen-2018-027235 on 14 October 2019. Downloaded from http://bmjopen.bmj.com/ on October 14, 2019 at UNIVERSITY OF ADELAIDE LIBRARY. Protected by copyright.

5

Table 1 Continued								
	Mode of delivery							
	DHS 2000		DHS 2005		DHS 2011		DHS 2016	
	Caesarean (n=86) %	Vaginal (n=12174) %	Caesarean (n=111) %	Vaginal (n=11 052) %	Caesarean (n=175) %	Vaginal (n=11697) %	Caesarean (n=213) %	Vaginal (n=10810) %
Private	1.3	0.1	5.9	0.3	13.6	0.8	13.7	0.9
NGO	2.4	0.2	3.8	0.1	3.0	0.2	2.4	0.2
Home	0.0	95.6	0.0	95.7	0.0	91.4	0.0	75.2
Birth order								
+	75.5	18.6	53.2	17.0	53.5	18.5	41.4	18.2
2	17.5	16.5	26.0	15.5	17.0	17.0	25.6	16.0
e	2.2	13.7	5.7	14.5	10.3	14.2	18.2	14.2
4	1.0	11.3	1.5	12.9	6.0	12.6	3.3	12.5
S	0.6	10.4	2.2	10.8	8.4	10.5	5.3	11.4
6+	3.2	29.5	11.4	29.4	4.7	27.2	6.3	27.6
Sex of child								
Male	61.8	51.2	51.3	51.3	57.2	51.9	54.0	51.9
Female	38.2	48.8	48.7	48.7	42.8	48.1	46.0	48.1
Size of baby at birth†								
Very large	9.8	5.3	30.7	22.3	25.8	19.2	26.5	17.6
Larger than average	32.0	25.4	11.4	9.5	11.7	12.7	16.0	13.8
Average	34.6	35.7	37.4	40.0	42.3	38.2	36.1	41.7
Smaller than average	19.7	27.5	9.4	7.3	4.9	8.7	7.0	10.1
Very small	3.7	5.9	9.5	20.6	14.8	20.6	12.7	16.0
Do not know	0.2	0.2	1.1	0.4	0.6	0.4	1.7	0.8
Wealth quantile								
Poorest	1.9	21.1	0.6	22.1	2.0	23.1	7.2	24.2
Poorer	4.9	21.0	5.9	21.3	8.4	22.6	12.2	23.1
Middle	4.1	22.0	3.9	22.5	0.0	20.7	10.8	20.9
Richer	6.1	20.5	9.7	20.0	6.5	19.3	9.3	18.3
Richest	83.0	15.5	79.9	14.2	74.1	14.2	60.4	13.5
n=weighted. *Missing for 2000 (n=9). †Mother's estimate of baby! DHS. Demographic and Hee	s size at birth. ath Survev: NGO, Non-c	governmental O	rganisation.					

6



Figure 1 Trends in proportion of institutional deliveries, caesarean section and neonatal death in the 5 years before each of the surveys, Ethiopia DHS 2000, 2005, 2011 and 2016.

vaginal births increased from 0.95 (95% CI: 0.29 to 3.19) in 2000 to 2.81 (95% CI: 1.11 to 7.13) in 2016. These findings suggest that the circumstances for foetuses born in 2000 survey were different from the foetuses in 2016. Our subgroup analyses using 2016 data suggest that the association between caesarean birth and neonatal death was stronger among rural women (aPR (95% CI) 3.43 (1.22 to 9.67)) and among women from the lowest quintile of household wealth (aPR (95% CI) 7.01 (0.92 to 53.36)), but not for births in areas with wider availability of caesarean such as Addis Ababa (aPR (95% CI) 1.07 (0.20 to 5.73)). The changing association between

caesarean birth and neonatal death over time, and the stronger association observed among different subgroup analyses may be attributable to changes in the pattern of confounding by indication due to contextual factors such as unequal access, structural health-system deficiencies (insufficient equipment, supplies and drugs), infrastructural and health workforce constraints.

The national caesarean section and institutional delivery rates in Ethiopia are still low though increases in the past decade are notable. There is also substantial disparity in caesarean section rates, with very low rates in rural areas and among the poorest women,²⁹ suggesting unequal access which may be as a consequence of a range of geographic, social and economic barriers. The low caesarean rates may also be due to lack of skilled birth attendants, and poor health infrastructure (eg, shortage of medical care institutions, deficiencies in surgical facilities, surgical and anaesthesia personnel and equipment, and blood transfusion capacity).³⁸⁻⁴⁰ For instance, in Ethiopia, there are only 820 obstetricians, 10846 general practitioners, 996 emergency obstetric surgeons, 6 345 health officers, 41009 nurses, 8 635 midwives, 233 anaesthesiologists and 33320 health extension workers for the population of over 90 million in 2015.⁴¹ Similarly, there are only 3 547 functional health centres, 16 447 functional health posts and 189 functional hospitals in 2015.42

We know from previous research that inadequate access to timely caesarean section may result in perinatal asphyxia, uterine rupture, obstructed labour, and these can contribute to maternal and newborn deaths.⁴³ Conversely, it was demonstrated that maternal and neonatal mortality due to obstetric complications can be prevented with



Figure 2 Trends in caesarean section rates in the 5 years before each of the surveys according to the nine regional states and two city administrations, Ethiopia DHS 2000, 2005, 2011 and 2016.

Table 2Crude and multivariable-adjusted prevalenceratios for neonatal death associated with caesarean versusvaginal delivery, Ethiopia DHS 2000, 2005, 2011 and 2016

	Prevalence ratio (95% CI) for neonatal death
Ethiopia DHS 2000	
Vaginal delivery	1 (<i>Ref.</i>)
Caesarean delivery, crude (n=10873)	0.93 (0.38 to 2.30)
Caesarean delivery, model 1* (n=10853)	0.95 (0.29 to 3.19)
Ethiopia DHS 2005	
Vaginal delivery	1 (<i>Ref.</i>)
Caesarean delivery, crude (n=9 861)	1.74 (0.67 to 4.51)
Caesarean delivery, model 1* (n=9 861)	1.53 (0.52 to 4.50)
Ethiopia DHS 2011	
Vaginal delivery	1 (<i>Ref.</i>)
Caesarean delivery, crude (n=11654)	1.49 (0.62 to 3.61)
Caesarean delivery, model 1* (n=11654)	1.15 (0.45 to 2.93)
Ethiopia DHS 2016	
Vaginal delivery	1 (<i>Ref.</i>)
Caesarean delivery, crude (n=10641)	3.02 (1.37 to 6.66)
Caesarean delivery, model 1* (n=10641)	2.81 (1.11 to 7.13)

*Adjusted for place of delivery, type of residence (urban/rural), sex of child, size of baby at birth, mother's age at birth, mother's education, birth order and household wealth. DHS, Demographic and Health Surveys.





Table 3Crude and multivariable-adjusted prevalenceratios for neonatal death associated with caesarean versusvaginal delivery, Ethiopia DHS 2016

		Prevalence ratio (95% CI) for neonatal death
N	lain analysis	
	Vaginal delivery	1 (<i>Ref.</i>)
	Caesarean delivery, crude (n=10641)	3.02 (1.37 to 6.66)
	Caesarean delivery, model 1* (n=10641)	2.81 (1.11 to 7.13)
S	ubgroup analyses	
	Restricted to Addis Ababa† (n=461)	1.07 (0.20 to 5.73)
	Excluded Addis Ababa and Harari* (n=9 575)	3.35 (1.31 to 8.56)
	Restricted to births in public facility* (n=3 023)	2.78 (1.16 to 6.63)
	Restricted to rural mothers† (n=8 636)	3.43 (1.22 to 9.67)
	Restricted to women from lowest quintile of household wealth‡ (n=3 958)	7.01 (0.92 to 53.36)
	Restricted to women from highest quintile of household wealth‡ (n=2 092)	2.72 (0.55 to 13.38)
*/	Adjusted for place of delivery type of reside	nce (urban/rural)

*Adjusted for place of delivery, type of residence (urban/rural), sex of child, size of baby at birth, mother's age at birth, mother's education, birth order and household wealth.

†Adjusted for place of delivery, sex of child, size of baby at birth, mother's age at birth, mother's education, birth order and household wealth.

‡Adjusted for place of delivery, sex of child, size of baby at birth, mother's age at birth, mother's education and birth order.

timely access to caesarean section.⁴³ ⁴⁴ Delay, therefore, emerges as relevant factor in worsening the underlying obstetric indications for caesarean intervention thereby contributing to neonatal death. Context-specific factors that delay access to caesarean section may have the capacity to make women with labour problems undergo caesarean section after severe complication of labour or severe foetal compromise. Therefore, our interpretation is that caesarean section conducted after severe foetal compromise may not prevent neonatal deaths because they have already experienced such severity of complications that although live born, neonatal death is difficult to prevent.

There are two possible scenarios leading to caesarean section in Ethiopia. First, when women who have previously had a caesarean section, with breech presentation, or other risk factors such as eclampsia attend specialised health facilities, they are usually allowed to undergo caesarean section. Their caesarean section is commonly classified as 'elective or scheduled caesarean section'. Second, when caesarean section is performed for 'emergency reasons'. Full-term mothers with or without signs of labour will be admitted to health facilities where their progress is monitored and labour-augmenting or inducing medications may be administered. Decisions to perform caesarean section in these facilities or decision to refer

ി





Figure 4 The relationship between caesarean section rate and neonatal death by survey years.

the mother to nearby hospitals for caesarean delivery or other action depends on the condition of the mother and fetus during the progress of labour. In primary health facilities (ie, health posts and health centres), obstetric care providers usually use a 'Partograph',^{45–46} a routine labour monitoring instrument (chart) which helps the healthcare providers to identify slow progress in labour and take appropriate action. In hospitals, the decision to perform a caesarean section is reached when the labour is prolonged and/or the second stage of labour is complicated risking the life of mother and fetus.

Given these pathways to caesarean delivery in mind, our interpretation of the association between caesarean birth and neonatal death in Ethiopia using the 2016 survey may be shaped by examining factors contributing to delays in the 'Three Delays Model'. This is because delays to caesarean section aggravate the underlying medical indications for caesarean intervention. Table 4 shows factors affecting the length of delays in the 'Three Delays Model' according to sociodemographic characteristics in the 2016 survey.

Phase I delay: deciding to seek care

In Ethiopia, poorer and less-educated women are more likely to select a nearby health facility, especially in rural areas, where there is limited access to caesarean section and the possibility of benefiting from caesarean section is mainly through referral to higher levels of care. Women are more likely to undergo a caesarean section if they present to specialised health facilities. However, the outcome of delivery depends on how quick/competent the healthcare provider is in referring the mother or on intervening, and the severity of the underlying obstetric complications for caesarean intervention which may be affected by the delay in women's or family's decision to seek care. Poor health decision-making depends on numerous factors such as educational status, distance to health facility, economic status, sociocultural factors (eg, unsupportive spouse and lack of autonomy) and quality of care.^{24 47 48}

Table 4 shows that 'distance to health facility' was a big problem in accessing healthcare for about 60% of rural and 17% of urban women in Ethiopia. Similarly, 'getting money for treatment' is a big problem to access healthcare and was reported by 61% of rural and 35% of urban women in 2016. On the other hand, the status of women in a given society affects the decision to seek care. For instance, efforts to seek timely care are influenced by women's limited mobility because they need permission to travel from spouse and/or mother-in-law.²⁴ In Ethiopia, about 37% of rural and 15% of urban women reported 'getting permission to go for treatment' was a big problem to access healthcare.

Phase II delay: identifying and reaching a medical facility

Delay in reaching healthcare may occur when women who encounter obstetric complication live farther from health facilities, where the availability and cost of transportation is problematic. In one study conducted in rural India, Kumar *et al*⁴⁷ found that health facility births occur less likely among women living farther away from the health facilities, suggesting distance as an important barrier to in-facility births for rural women. In addition to the travel distance, the scarcity of transportation which may be accompanied by poor roads is also another obstacle for women with labour complications to timely reach even the closest health facility. As a result of this, women who arrive at the nearby facility

Table 4 Factors contri	buting to the 'Three	Delays Model	, according to so	ciodemographic c	characteristics, Ethiopia	a DHS 2016		
			Problems in ac	cessing healthca	Ire by women aged 15	⊢49 years†		
	Delivery by skilled provider*	Number of births	Distance to health facility	Getting money for treatment	Getting permission to go for treatment	Not wanting to go alone	At least one problem accessing healthcare	Number of women
Types of residence								
Urban	80.1	1 216	17.0	34.7	15.1	21.4	45.6	3 476
Rural	21.2	9 807	59.8	60.5	37.0	47.9	76.9	12207
Region								
Tigray	59.3	716	37.4	46.1	15.3	24.6	60.7	1 129
Affar	16.4	114	54.3	51.7	28.2	41.8	66.6	128
Amhara	27.7	2 072	33.7	35.3	15.4	34.6	55.7	3 714
Oromia	19.7	4 851	68.9	70.1	58.3	57.0	82.9	5 701
Somali	20.0	508	47.3	63.0	25.7	32.2	72.6	459
Benishangul-Gumuz	28.6	122	57.4	62.4	36.5	43.8	76.8	160
SNNPR	28.6	2 296	52.7	59.1	18.4	39.5	75.4	3 288
Gambela	46.9	27	41.0	44.3	24.3	33.7	61.2	44
Harari	51.2	26	18.1	28.2	16.3	13.8	30.8	38
Addis Ababa	96.8	244	10.8	29.2	8.7	14.5	40.0	930
Dire Dawa	56.7	47	57.4	64.5	58.7	55.2	71.4	06
Mother's education								
No education	17.2	7 284	59.2	62.9	37.6	47.1	78.0	7 498
Primary	38.6	2 951	50.3	55.7	31.9	43.2	71.1	5 490
Secondary	78.4	514	27.8	33.2	18.2	27.8	48.1	1 817
More than secondary	93.2	274	20.6	23.8	15.9	20.4	39.8	877
Wealth quintile								
Poorest	11.0	2 636	67.7	70.9	40.0	54.5	85.3	2 633
Poorer	20.8	2 520	66.8	67.0	42.1	52.7	82.9	2 809
Middle	24.2	2 280	59.4	61.0	35.2	47.6	77.3	2 978
Richer	28.5	1 999	49.8	50.2	33.8	41.2	68.2	3 100
Richest	70.3	1 588	22.1	35.2	17.0	23.4	47.7	4 163
Total	27.7	11 023	50.3	54.8	32.1	42.0	70.0	15683
*Percentage delivered by <i>ε</i> †Percentage of women age sociodemographic charact	t skilled provider (ie, do ad 15–49 who reporte eristics, Ethiopia DHS	octor, nurse, midv d that they have 2016.	wife, health officer serious problems ir	and health extensior מלחבים מימים מ	n worker). Ire for themselves when th	iey are sick, by t	ype of problem, according to	

following obstetric complications probably will travel further to specialised hospital due to emergency referral. It is clear that the obstetric complications encountered by mothers reaching nearby primary health facilities will be compounded by additional delays when they are referred for caesarean section. These scenarios highlight the likelihood of adverse delivery outcome followed by aggravated obstetric complications due to delays in reaching medical facility as high.

In Ethiopia, about 50% of women of reproductive age (15–49 years) reported 'distance to health facility' as a big problem to access healthcare (table 4). Moreover, access to caesarean situation in Ethiopia is worse than in most other settings.

Phase III delay: receiving adequate and appropriate treatment

Phase III delays occur within any health facilities and are indicators of inadequate care due to lack of facilities; inadequately trained obstetric care givers (skilled birth attendants) and deficiencies in surgical facilities, surgical and anaesthesia personnel and equipment, and blood transfusion as well as inadequate and inappropriate referral systems. These deficiencies will limit women's access to lifesaving procedures such as caesarean section. In Ethiopia, only 28.0% of all births were delivered by 'skilled providers' (ie, doctor, nurse, midwife, health officer and health extension worker) in the 2016 survey. Table 4 also shows that there are disparities in the proportion of births attended by skilled birth attendants by urban-rural place of residence, region, level of mother's education and household wealth. It is quite clear that insufficient number of skilled birth attendants at any health facility will lead to delay in receiving appropriate treatment among women with obstetric complications. Although health posts and health centres (primary healthcare unit) are the most accessible to the general population in Ethiopia, they are not fully equipped to deal with obstetric complications.^{41 42} As a result of this, women with obstetrics complications will have to travel on to better equipped institutions (secondary and tertiary levels of healthcare) with caesarean section capacity (eg, general hospitals and specialised hospitals) through referral. By the time women reach these well-equipped health facilities, the delays will have further aggravated the obstetric complications on the way. A schematic representation of the Ethiopian health system structure is provided in online supplementary figure A5.

On the other hand, delay in caesarean intervention may even happen if mothers with less severe obstetric complications were referred and presented to specialised health facilities in a timely manner. This is because a trial of labour is usually attempted before a decision to have caesarean section. For instance, some women who are referred from primary health facilities undergo induction and augmentation of labour because these interventions are only provided in health facilities with the capacity to provide caesarean section in Ethiopia. These practices, in turn, will result in delay in receiving caesarean section leading to worsening of the already existing obstetric complications. Thus, any delays to caesarean intervention have a higher chance of aggravating the already existing complications and increase the risk of neonatal death.

DISCUSSION

Our study examined the changing temporal association between caesarean birth and neonatal death within the context of Ethiopia from 2000 to 2016. The association between caesarean section and neonatal death increased over time and was variable among population subgroups. These changes over time, and variation across population subgroups may be attributable to changes in the pattern of confounding by indication due to contextual factors such as improvement in health service coverage, unequal access (eg, due to a range of geographic, social and economic barriers) and structural and health workforce constraints.

In Ethiopia, the proportion of women aged 15–49 years who received any antenatal care from a skilled provider has increased from 27% in 2000 to 62% in 2016.³² Health facility-based deliveries have increased from 5% in 2000 to 26% in 2016 (increased from 2% in 2000 to 20% in 2016 for rural women, and increased from 32% in 2000 to 79% in 2016 for urban women).³² The proportion of births in health facilities assisted by skilled birth attendants increased from 6% in 2000 to 28% in 2016.³² These figures reflect improvement in health service coverage in Ethiopia.

Moreover, since 2003, with the implementation of the Health Extension Program-a community-based primary healthcare program-the Ethiopian government has increased the number of health posts from 4 211 in 2005 to 16 447 in 2015.^{42 49} Likewise, the number of health centres was increased from 600 in 2005 to 3 586 in 2015.^{42 49} However, due to limitations in proper monitoring of labour for making timely decisions, especially on whether or not to initiate a referral from primary health facilities to higher level facilities, and due to poor transport and road networks which are still the common problems in low-income countries,⁵⁰ the underlying medical indications for caesarean intervention will be worsened by factors contributing to 'delays'. Delay in receiving adequate and appropriate care is still a common problem in low-income countries due to deficiencies in surgical facilities, surgical and anaesthesia personnel and equipment, blood transfusion capacity and shortage of skilled birth attendants.⁵¹⁻⁵³ There is also an inequitable distribution of the health workforce across urban and rural areas. For example, the majority of specialist doctors in Ethiopia serve in urban areas, where the total population distribution is only 19.4%.^{41 42} These situations often result in poor quality care to rural women, and the caesarean section conducted after a complicated labour may be associated with increased neonatal mortality due to confounding by indication.

Unlike previous studies, the present study takes into account the interpretation of the association between caesarean birth and neonatal death within the context of Ethiopia using DHS data. The change in the strength of effect estimates across DHS waves, and the different subgroup analyses suggest that neonatal mortality can be reduced by increasing timely access to caesarean section and timely decision for caesarean delivery via increasing health service coverage, improving infrastructure (eg, increasing number of health facilities), increasing the number of skilled birth attendants, improving quality of care and increasing awareness about antenatal care and health facility delivery among women. Moreover, provision of training to skilled birth attendants on close monitoring of labour and early detection of complications, equipping the primary health facilities (eg. health centres) to the level of caesarean capacity, and continuous financial investment in primary health facilities will be an important strategy to reduce neonatal mortality.

It appears that previous studies which used individual-level data are more likely to report an increased risk of neonatal death among infants born by caesarean section than the ecological studies. This may be due to the indications for the caesarean delivery (eg, the severity of the underlying causes) was involved in causing both caesarean delivery and neonatal death in studies which used individual-level data, suggesting the role of confounding by indication in the association between caesarean birth and neonatal death because an intended effect of caesarean birth is prevention of neonatal death. Therefore, the increased risk for neonatal death associated with caesarean birth, compared with vaginal birth, would appear to be intuitive given the fact that neonatal death rates after emergency caesarean section is strongly dependent on the underlying medical indication (eg, antenatally diagnosed foetal malformation or foetal growth restriction) for caesarean intervention.

In Ethiopia, the national rate of caesarean section increased from 0.7% in 2000 to 1.9% in 2016. On the other hand, neonatal mortality rate declined from 49 deaths per 1 000 live births in 2000 to 29 deaths per 1 000 births in 2016.³² Similarly, the pregnancy-related mortality ratio decreased from 871 pregnancy-related deaths per 100000 live births in 2000 to 412 pregnancy-related deaths per 100 000 live births in 2016.³² Our analyses based on aggregate-level data from Ethiopian DHS showed that an increase in caesarean section rate is correlated with a decrease in the proportion of neonatal deaths. Even though similar context-specific interpretation is applicable to ecological studies, additional explanation may also be necessary to interpret the association. For example, a change in neonatal mortality rate may be attributable to changes acting on the population as a whole-that is, changes in health coverage indicators, such as an increase in births attended by skilled birth attendants (increased from 6% in 2000 to 28% in 2016)³² and immunisation coverage (was 86.4%) in 2015).42

We acknowledge the following limitations of this study. First, as both the proportion of institutional deliveries and caesarean section rate is low in Ethiopia, especially in rural areas, the number of neonatal deaths following caesarean section may be low. However, since our analyses are weighted, we believe that the weight improves the representativeness of the data in terms of size, distribution and characteristics of the Ethiopian population. The weight may also ensure that our estimates are unbiased though the CI for some subgroup analyses are somewhat wide. Second, the interpretation of our study is specific to the context of Ethiopia and may not be generalisable to other developing countries in Africa or elsewhere. Another limitation is the mother's recall of the child's size at birth was used as a substitute for the child's birth weight in this study because the data for birth weight were not collected for more than 50% of the neonates in DHS.

CONCLUSIONS

A naïve interpretation of the changing temporal association between caesarean birth and neonatal death from 2000 to 2016 is that caesarean section is increasingly associated with neonatal death. However, the changing temporal association likely reflects improvements in health service coverage and secular shifts in the characteristics of Ethiopian women undergoing caesarean section after complicated labour or severe foetal compromise.

Twitter Engida Yisma @Engida_Yisma

Acknowledgements The authors are grateful to The DHS Program for providing the data sets used for this analysis. BWM is supported by an NHMRC Practitioner Fellowship (GNT1082548) and also reports consultancy for ObsEva, Merck Merck KGaA, and Guerbet. JWL is supported by an NHMRC Centre of Research Excellence (GNT1099422).

Contributors All authors (EY, BWM, JWL and LGS) contributed to the design of the study and the interpretation of data. EY performed the data analysis and drafted the manuscript. All other authors critically revised the draft manuscript. All authors read and approved the final manuscript. EY is the guarantor of the paper.

Funding The first author is fully supported by an Australian Government Research Training Programme Scholarship (RTPS).

Competing interests None declared.

Patient consent for publication Not required.

Provenance and peer review Not commissioned; externally peer reviewed.

Data availability statement Data are available in a public, open access repository.

Open access This is an open access article distributed in accordance with the Creative Commons Attribution Non Commercial (CC BY-NC 4.0) license, which permits others to distribute, remix, adapt, build upon this work non-commercially, and license their derivative works on different terms, provided the original work is properly cited, appropriate credit is given, any changes made indicated, and the use is non-commercial. See: http://creativecommons.org/licenses/by-nc/4.0/.

ORCID iD

Engida Yisma http://orcid.org/0000-0003-0703-1515

REFERENCES

1 United Nations Inter-agency Group for Child Mortality Estimation (UN IGME). Levels and trends in child mortality: report 2017, estimates developed by the UN Inter-agency Group for Child Mortality Estimation. New York: United Nations Children's Fund, 2017.

- 2 UN General Assembly. Transforming our world: the 2030 agenda for sustainable development, 21 October 2015, A/RES/70/1. Available: https://www.refworld.org/docid/57b6e3e44.html [Accessed 28 Jun 2019].
- 3 Costello A, Osrin D. Epidemiological transition, medicalisation of childbirth, and neonatal mortality: three Brazilian birth-cohorts. *Lancet* 2005;365:825–6.
- 4 Gibbons L, Belizan JM, Lauer JA, et al. Inequities in the use of cesarean section deliveries in the world. Am J Obstet Gynecol 2012;206:331.e1–19.
- 5 Leone T, Padmadas SS, Matthews Z. Community factors affecting rising caesarean section rates in developing countries: an analysis of six countries. Soc Sci Med 2008;67:1236–46.
- 6 de Mello e Souza C. C-sections as ideal births: the cultural constructions of beneficence and patients' rights in Brazil. *Camb Q Healthc Ethics* 1994;3:358–66.
- 7 Dumont A, de Bernis L, Bouvier-Colle MH, et al. Caesarean section rate for maternal indication in sub-Saharan Africa: a systematic review. Lancet 2001;358:1328–33.
- 8 Vogel JP, Betrán AP, Vindevoghel N, *et al.* Use of the Robson classification to assess caesarean section trends in 21 countries: a secondary analysis of two WHO multicountry surveys. *Lancet Glob Health* 2015;3:e260–70.
- 9 Ye J, Betrán AP, Guerrero Vela M, *et al*. Searching for the optimal rate of medically necessary cesarean delivery. *Birth* 2014;41:237–44.
- 10 World Health Organization. WHO statement on caesarean section rates. Geneva: World Health Organization, 2015.
- 11 Betrán AP, Merialdi M, Lauer JA, *et al.* Rates of caesarean section: analysis of global, regional and national estimates. *Paediatr Perinat Epidemiol* 2007;21:98–113.
- 12 Molina G, Weiser TG, Lipsitz SR, et al. Relationship between cesarean delivery rate and maternal and neonatal mortality. JAMA 2015;314:2263–70.
- 13 Ye J, Zhang J, Mikolajczyk R, et al. Association between rates of caesarean section and maternal and neonatal mortality in the 21st century: a worldwide population-based ecological study with longitudinal data. BJOG 2016;123:745–53.
- 14 Althabe F, Sosa C, Belizán JM, et al. Cesarean section rates and maternal and neonatal mortality in low-, medium-, and high-income countries: an ecological study. *Birth* 2006;33:270–7.
- 15 Lumbiganon P, Laopaiboon M, Gülmezoglu AM, et al. Method of delivery and pregnancy outcomes in Asia: the WHO global survey on maternal and perinatal health 2007-08. Lancet 2010;375:490–9.
- 16 Villar J, Carroli G, Zavaleta N, et al. Maternal and neonatal individual risks and benefits associated with caesarean delivery: multicentre prospective study. *BMJ* 2007;335.
- 17 MacDorman MF, Declercq E, Menacker F, et al. Infant and neonatal mortality for primary cesarean and vaginal births to women with 'no indicated risk,' United States, 1998-2001 birth cohorts. *Birth* 2006;33:175–82.
- 18 MacDorman MF, Declercq E, Menacker F, et al. Neonatal mortality for primary cesarean and vaginal births to low-risk women: application of an 'intention-to-treat' model. *Birth* 2008;35:3–8.
- 19 Källén K, Olausson PO. Neonatal mortality for low-risk women by method of delivery. *Birth* 2007;34:99–100. author reply 1-2.
- 20 Shah A, Fawole B, M'imunya JM, et al. Cesarean delivery outcomes from the WHO global survey on maternal and perinatal health in Africa. Int J Gynaecol Obstet 2009;107:191–7.
- 21 Kyu HH, Shannon HS, Georgiades K, *et al.* Caesarean delivery and neonatal mortality rates in 46 low- and middle-income countries: a propensity-score matching and meta-analysis of Demographic and Health Survey data. *Int J Epidemiol* 2013;42:781–91.
- 22 Begum T, Rahman A, Nababan H, et al. Indications and determinants of caesarean section delivery: evidence from a population-based study in Matlab, Bangladesh. PLoS One 2017;12:e0188074.
- 23 Belizán JM, Minckas N, McClure EM, et al. An approach to identify a minimum and rational proportion of caesarean sections in resource-poor settings: a global network study. *Lancet Glob Health* 2018;6:e894–901.
- 24 Thaddeus S, Maine D. Too far to walk: maternal mortality in context. Soc Sci Med 1994;38:1091–110.
- 25 Stanton CK, Dubourg D, De Brouwere V, et al. Reliability of data on caesarean sections in developing countries. Bull World Health Organ 2005;83:449–55.
- 26 Channon AAR. Can mothers judge the size of their newborn? assessing the determinants of a mother's perception of a baby's size at birth. *J Biosoc Sci* 2011;43:555–73.
- 27 Filmer D, Pritchett LH. Estimating wealth effects without expenditure data–or tears: an application to educational enrollments in states of India. *Demography* 2001;38:115–32.

- 28 Filmer D, Scott K. Assessing asset indices. *Demography* 2012;49:359–92.
- 29 Yisma E, Smithers LG, Lynch JW, *et al*. Cesarean section in Ethiopia: prevalence and sociodemographic characteristics. *J Matern Fetal Neonatal Med* 2019;32:1130–5.
- 30 Souza JP, Gülmezoglu A, Lumbiganon P, et al. Caesarean section without medical indications is associated with an increased risk of adverse short-term maternal outcomes: the 2004-2008 WHO Global Survey on Maternal and Perinatal Health. BMC Med 2010;8:71.
- 31 Ronsmans C, Etard JF, Walraven G, et al. Maternal mortality and access to obstetric services in West Africa. Trop Med Int Health 2003;8:940–8.
- 32 Central Statistical Agency CSA/Ethiopia, ICF. Ethiopia Demographic and Health Sruvey 2016. Addis Ababa, Ethiopia: CSA and ICF, 2017.
- 33 Boerma T, Ronsmans C, Melesse DY, et al. Global epidemiology of use of and disparities in caesarean sections. Lancet 2018;392:1341–8.
- 34 Boatin AA, Schlotheuber A, Betran AP, et al. Within country inequalities in caesarean section rates: observational study of 72 low and middle income countries. *BMJ* 2018;360.
- 35 Upadhyay RP, Rai SK, Krishnan A. Using three delays model to understand the social factors responsible for neonatal deaths in rural Haryana, India. *J Trop Pediatr* 2013;59:100–5.
- 36 Mbaruku G, van Roosmalen J, Kimondo I, et al. Perinatal audit using the 3-delays model in Western Tanzania. Int J Gynaecol Obstet 2009;106:85–8.
- 37 Waiswa P, Kallander K, Peterson S, et al. Using the three delays model to understand why newborn babies die in eastern Uganda. Trop Med Int Health 2010;15:964–72.
- 38 Holmer H, Lantz A, Kunjumen T, et al. Global distribution of surgeons, anaesthesiologists, and obstetricians. Lancet Glob Health 2015;3(Suppl 2):S9–11.
- 39 Ologunde R, Vogel JP, Cherian MN, et al. Assessment of cesarean delivery availability in 26 low- and middle-income countries: a crosssectional study. Am J Obstet Gynecol 2014;211:504.e1–12.
- 40 Say L, Raine R. A systematic review of inequalities in the use of maternal health care in developing countries: examining the scale of the problem and the importance of context. *Bull World Health Organ* 2007;85:812–9.
- 41 WHO. Primary health care systems (PRIMASYS): case study from Ethiopia, abridged version. Geneva: World Health Organization, 2017Licence: CC BY-NC-SA 3.0 IGO.
- 42 Federal Ministry of Health. Health and health related indicators 2007 E.C. (2014/15 G.C). Addis Ababa, Ethiopia: Federal Ministry of Health, 2015.
- 43 Betran AP, Torloni MR, Zhang JJ, *et al*. WHO statement on caesarean section rates. *BJOG* 2016;123:667–70.
- 44 Thomas S, Meadows J, McQueen KAK. Access to cesarean section will reduce maternal mortality in low-income countries: a mathematical model. *World J Surg* 2016;40:1537–41.
- 45 Yisma E, Dessalegn B, Astatkie A, et al. Knowledge and utilization of partograph among obstetric care givers in public health institutions of Addis Ababa, Ethiopia. BMC Pregnancy Childbirth 2013;13:17.
- 46 Yisma E, Dessalegn B, Astatkie A, *et al.* Completion of the modified World Health Organization (WHO) partograph during labour in public health institutions of Addis Ababa, Ethiopia. *Reprod Health* 2013;10:23.
- 47 Kumar S, Dansereau EA, Murray CJL. Does distance matter for institutional delivery in rural India? *Appl Econ* 2014;46:4091–103.
- 48 Anselmi L, Lagarde M, Hanson K. Health service availability and health seeking behaviour in resource poor settings: evidence from Mozambique. *Health Econ Rev* 2015;5:62.
- 49 Federal Ministry of Health. Health and health related indicators 1997 E.C. (2004/05 G.C). Addis Ababa, Ethiopia: Federal Ministry of Health, 2005.
- 50 Atuoye KN, Dixon J, Rishworth A, et al. Can she make it? transportation barriers to accessing maternal and child health care services in rural Ghana. BMC Health Serv Res 2015;15:333.
- 51 Bergström S. Training non-physician mid-level providers of care (associate clinicians) to perform caesarean sections in low-income countries. Best Pract Res Clin Obstet Gynaecol 2015;29:1092–101.
- 52 Orji EO, Ojofeitimi EO, Esimai AO, et al. Assessment of delays in receiving delivery care at a tertiary healthcare delivery centre in Nigeria. J Obstet Gynaecol 2006;26:643–4.
- 53 Miller S, Abalos E, Chamillard M, et al. Beyond too little, too late and too much, too soon: a pathway towards evidence-based, respectful maternity care worldwide. Lancet 2016;388:2176–92.

Chapter 6: Impact of caesarean section on breastfeeding indicators: within-country and meta-analyses of nationally representative data from 33 countries in sub-Saharan Africa

6.1 Preface

This chapter contains the third of the five studies contributing to this thesis. The study was published in *BMJ Open* in 2019.

The previous chapter, Chapters 5, of this thesis examined the association between caesarean birth and neonatal mortality from the perspective of one country's context and suggests that timely access and timely caesarean intervention is important to reduce neonatal mortality in Ethiopia. Chapter 6 expands on caesarean section, as one of the most common obstetrical interventions during childbirth, and examined the consequences of caesarean section on three different breastfeeding outcomes—early initiation of breastfeeding within one hour, exclusive breastfeeding under 6 months and children ever breastfed (at least once)—in 33 countries in sub-Saharan Africa. This chapter presents the results of novel analysis on the effect caesarean section on breastfeeding outcomes (indicators), which were analysed in two phases. In the first phase, to understand how the estimates of caesarean section on breastfeeding differ across countries, analyses were conducted for each of the 33 countries in sub-Saharan Africa. As there were three breastfeeding outcomes, three different outcome regression analysis were performed for each country. In the second phase, the evidence (estimates) from each of the 33 countries were summarized in an overall estimate using random-effects meta-analysis for each breastfeeding outcome. As there is no previous studies that have comprehensively assessed the impact of caesarean section on breastfeeding indicators in sub-Saharan Africa covering a large number of countries, this article (Study 3) addresses an important research gap. Study 3 also provides high-quality public health evidence regarding the link between caesarean section and breastfeeding from the perspective of low-and middle-income countries in sub-Saharan Africa.

6.2 Publication

Yisma E, Mol BW, Lynch JW, Smithers LG. Impact of caesarean section on breastfeeding

indicators: within-country and meta-analyses of nationally representative data from 33

countries in sub-Saharan Africa. BMJ Open 2019; 9(9): e027497.

http://dx.doi.org/10.1136/bmjopen-2018-027497.

6.2.1 Statement of Authorship

Title of Paper	Impact of caesarean section on b analyses of nationally representa	preastfeeding indicators: within-country and meta- ative data from 33 countries in sub-Saharan Africa
Publication Status	Published	Accepted for Publication
	Submitted for Publication	Unpublished and Unsubmitted w ork w ritten in manuscript style
Publication Details	Yisma E, Mol BW, Lynch JW, S breastfeeding indicators: within- representative data from 33 cour 9 (9): e027497. <u>http://dx.doi.org/</u>	Smithers LG. Impact of caesarean section on -country and meta-analyses of nationally ntries in sub-Saharan Africa. <i>BMJ Open</i> 2019; /10.1136/bmjopen-2018-027497.

Principal Author

Name of Principal Author (Candidate)	Engida Yisma
Contribution to the Paper	Conceived and designed the study, performed the statistical analysis, interpreted the data, and drafted the manuscript.
Overall percentage (%)	85%
Certification:	This paper reports on original research I conducted during the period of my Higher Degree by Research candidature and is not subject to any obligations or contractual agreements with a third party that would constrain its inclusion in this thesis. I am the primary author of this paper.
Signature	Date 5-Aug-2020

Co-Author Contributions

By signing the Statement of Authorship, each author certifies that:

- i. the candidate's stated contribution to the publication is accurate (as detailed above);
- ii. permission is granted for the candidate to include the publication in the thesis; and
- iii. the sum of all co-author contributions is equal to 100% less the candidate's stated contribution.

Name of Co-Author	Ben W Mol		
Contribution to the Paper	Contributed to the design of the the manuscript.	e study, interpreted	l the findings, and reviewed
Signature		Date	04/08/2020

Name of Co-Author	John W Lynch
Contribution to the Paper	Contributed to the design of the study, interpreted the findings, and reviewed the manuscript.
Signature	Date 6/8/20

Name of Co-Author	Lisa G Smithers		
Contribution to the Paper	Contributed to the design of the study, in the manuscript.	nterpretec	d the findings, and reviewed
Signature		Date	5-Aug-2020

BMJ Open Impact of caesarean section on breastfeeding indicators: within-country and meta-analyses of nationally representative data from 33 countries in sub-Saharan Africa

Engida Yisma,⁹^{1,2} Ben W Mol,^{3,4} John W Lynch,^{1,3,5} Lisa G Smithers^{1,3}

To cite: Yisma E, Mol BW, Lynch JW, *et al.* Impact of caesarean section on breastfeeding indicators: withincountry and meta-analyses of nationally representative data from 33 countries in sub-Saharan Africa. *BMJ Open* 2019;**9**:e027497. doi:10.1136/ bmjopen-2018-027497

► Prepublication history and additional material for this paper are available online. To view these files, please visit the journal online (http://dx.doi. org/10.1136/bmjopen-2018-027497).

Received 25 October 2018 Revised 15 July 2019 Accepted 17 July 2019



© Author(s) (or their employer(s)) 2019. Re-use permitted under CC BY-NC. No commercial re-use. See rights and permissions. Published by BMJ.

For numbered affiliations see end of article.

Correspondence to

Engida Yisma; engida.derbie@adelaide.edu.au

ABSTRACT

Objective To examine the impact of caesarean section on breastfeeding indicators—early initiation of breastfeeding, exclusive breastfeeding under 6 months and children ever breastfed (at least once)—in sub-Saharan Africa. **Design** Secondary analysis of Demographic and Health Surveys (DHS).

Setting Thirty-three low-income and middle-income countries with a survey conducted between 2010 and 2017/2018.

Participants Women aged 15–49 years with a singleton live last birth during the 2 years preceding the survey. **Main outcome measures** We analysed the DHS data to examine the impact of caesarean section on breastfeeding indicators using the modified Poisson regression models for each country adjusted for potential confounders. For each breastfeeding indicator, the within-country adjusted prevalence ratios (aPR) were pooled in random-effects meta-analysis.

Results The within-country analyses showed, compared with vaginal birth, caesarean section was associated with aPR for early initiation of breastfeeding that ranged from 0.24 (95% CI 0.17 to 0.33) in Tanzania to 0.89 (95% CI 0.78 to 1.00) in South Africa. The aPR for exclusive breastfeeding under 6 months ranged from 0.58 (95% CI 0.34 to 0.98) in Angola to 1.93 (95% CI 0.46 to 8.10) in Cote d'Ivoire, while the aPR for children ever breastfed ranged from 0.91 (95% CI 0.82 to 1.02) in Gabon to 1.02 (95% CI 0.99 to 1.04) in Gambia. The meta-analysis showed caesarean section was associated with a 46% lower prevalence of early initiation of breastfeeding (pooled aPR, 0.54 (95% CI 0.48 to 0.60)). However, meta-analysis indicated little association with exclusive breastfeeding under 6 months (pooled aPR, 0.94 (95% CI 0.88 to 1.01)) and children ever breastfed (pooled aPR, 0.98 (95% CI 0.98 to 0.99)) among caesarean versus vaginally born children.

Conclusions Caesarean section had a negative influence on early initiation of breastfeeding but showed little difference in exclusive breastfeeding under 6 months and children ever breastfed in sub-Saharan Africa.

INTRODUCTION

Breastfeeding provides protection against infection, prevents neonatal death and

Strengths and limitations of this study

- This study used nationally representative Demographic and Health Surveys (DHS) data on a large number of countries (n=33 countries) over a long time period (2010–2017/2018).
- We performed analyses adjusted for potential confounders for each of the 33 countries in sub-Saharan Africa to examine the impact of caesarean section on breastfeeding indicators, and this helped us understand how the estimates (adjusted prevalence ratios) differ across countries.
- In addition to the within-country adjusted analyses, we conducted random-effects meta-analysis to summarise the available evidence regarding the estimate of the impact of caesarean section on breastfeeding indicators in sub-Saharan Africa.
- DHS data do not distinguish whether the caesarean section was medically indicated or not.

improves childhood nutritional status.1 2 Moreover, breastfeeding has other numerous benefits for newborns, including reduction in risk of diarrhoea and respiratory tract infections, otitis media, asthma and allergies.^{1 3–7} Breastfeeding has also been shown to reduce risks of type 2 diabetes,^{1 8} breast and ovarian cancer for mothers.¹⁹ The World Health Organization (WHO) recommends mothers initiate breastfeeding within 1 hour of birth and exclusive breastfeeding under 6 months of life, with continued breastfeeding together with complementary feeding for 2 years to enhance growth, development and health of the child. $^{10\,11}$ However, only 37% of infants younger than 6 months are exclusively breastfed in low-income and middle-income countries,¹ suggesting the need for interventions to increase uptake and duration of breastfeeding.

Obstetrical interventions during labour and childbirth, such as caesarean section,

influence women's breastfeeding practices and are among the causes for concern, given the rising caesarean section rates worldwide. The breastfeeding experiences of women who underwent caesarean section can be influenced by several factors, including mothers' health and emotional reactions to the surgery as well as infant health and behaviour. For instance, women's restricted mobility in the early days after caesarean birth may hinder efforts to provide basic infant care, including breastfeeding.¹² As the postoperative pain may be severe, particularly in the first 24 hours, it affects the breastfeeding experience of women.¹³¹⁴ Likewise, due to the potential for physical separation of the mother and infant, given the higher risk of infant admission to neonatal intensive care unit as a consequence of respiratory disorders, there might be a lesser likelihood of timely initiation of breastfeeding.¹⁸

On the other hand, it is hypothesised that the hormonal pathway that stimulates 'lactogenesis' may be disturbed by caesarean section due to maternal stress or decreased oxytocin secretion, and this, in turn, may delay milk production.^{16–18} As planned caesarean births are usually performed prior to the onset of labour, there is a possibility that decreased breastfeeding initiation may be attributed to these physiological reasons.

Previous evidence suggested that caesarean birth was associated with lower prevalence of early initiation of breastfeeding among mothers. For instance, in a study conducted in Canada, it was demonstrated that fewer women who had planned caesarean birth reported the practice of early initiation of breastfeeding when compared with women who did not have planned caesarean birth or had vaginal birth.¹⁹ Similarly, in a systematic review and meta-analysis involving 53 studies from 33 low-income, middle-income and high-income countries with no country from Africa, it was found that early initiation of breastfeeding was lower among infants delivered by caesarean section (pooled OR, 0.57; 95% CI 0.50 to 0.64) compared with infants born vaginally.²⁰

Currently, much of the available evidence regarding the impact of caesarean section on breastfeeding is generated from high-income countries. In low-income and middle-income countries, even though the caesarean section rate is increasing, there is substantial inequalities which may suggest inadequate access among the poorest women and overuse of caesarean section for non-medical indications among the richest population subgroups.²¹ Moreover, unequal access to caesarean section due to deficiencies in transport, surgical facilities and shortage of skilled birth attendants will result in delay in accessing emergency caesarean section, which is usually accessible at specialised health facilities, and this may lead to negative outcomes.^{22–24} For example, breastfeeding after emergency caesarean section performed due to complicated labour may be more stressful for both mother and baby. However, there is limited insight regarding the impact of caesarean section on breastfeeding indicators-early initiation of breastfeeding (within the first hour), exclusive breastfeeding under 6 months and children ever

breastfed (at least once)—in low-income and middle-income countries in sub-Saharan Africa. The purpose of this study was to examine the impact of caesarean section on breastfeeding indicators in 33 countries in sub-Saharan Africa and to summarise the findings in a random-effects meta-analysis by combining the effect estimates analysed individually to provide a consolidated effect estimate of the impact of caesarean section on breastfeeding indicators in sub-Saharan Africa.

METHODS

Data sources

We obtained data from most recent Demographic and Health Surveys (DHS) conducted in 33 sub-Saharan African countries from 2010 to 2017/2018. These countries were selected based on the availability of the most recent standard DHS data in each nation. We restricted our analysis to sub-Saharan Africa to avoid incomparability issues and the time period (2010-2017/2018) was selected to insure enough coverage of a full range of contemporaneous events in DHS within each country (eg, increase in caesarean section rates and changes in breastfeeding practices).

The DHS are widely available high-quality data source in low-income and middle-income countries and are comparable for studies across nations. DHS uses a standardised methodology and identical core questionnaire to collect nationally representative information about sociodemographic characteristics and health indicators such as maternal and child health, nutrition, HIV/ AIDS, malaria and family planning. The details about the methodology and standards for protecting the privacy of study participants in all DHS can be accessed from the DHS Program (https://dhsprogram.com/What-We-Do/ methodology.cfm). The DHS Program produces the final edited data files and make freely available to data users worldwide. The DHS Country Report, which includes the comprehensive survey results and country survey specificities (ie, sample design, non-response rate, estimates of sampling errors, data quality tables) for each country can also be accessed online (https://dhsprogram.com/publications/publications-by-type.cfm).

Exposure

The DHS questionnaire asks women about pregnancy, antenatal care and methods of delivery, including caesarean section for the most recent birth in the 2 years before the survey. The self-reported data on caesarean section rates collected for DHS, compared with facility-based records of caesarean sections, are found to be reliable in developing countries.²⁵ In the current study, the exposure group comprised singleton last born children delivered by caesarean section while the unexposed group comprised children born vaginally in the 2 years before the survey.

Outcome

Respondents were asked a number of questions regarding breastfeeding practices, the length of breastfeeding,

about children's consumption of liquids and solid food and micronutrient supplementation. Breastfeeding questions included whether the child was ever breastfed; how long after birth before the newborn was put to the breast; whether the baby was given anything to drink or eat other than breast milk; the type of drink given to the child and whether the child was still breastfeeding. Based on this information, three breastfeeding indicator variablesearly initiation of breastfeeding, exclusive breastfeeding under 6 months and children ever breastfed-were computed and used as an outcome of interest for this study. According to the WHO definitions,¹¹ 'early initiation of breastfeeding' is defined as being put to the breast within an hour of birth. 'Exclusive breastfeeding under 6 months' is consuming only breastmilk (no other fluids or foods) from 0 to 5 months. 'Children ever breastfed' is the proportion of children born in the last 24 months who were ever breastfed (at least once).

Confounding

The following potential confounders were identified based on a priori subject matter and expert knowledge. They included pregnancy planning, birth weight, region of residence, sex of child, mother's age at birth (in years), mother's education, birth order, number of antenatal visits, maternal tobacco use, place of delivery, household wealth quintile, mother's occupation, distance to health facility and urban/rural residence. Mother's age at birth was calculated as a difference (in years) between infant's date of birth and mother's date of birth. DHS computes the wealth index for each survey based on household assets using principal components analyses²⁶ and categorises households into wealth quintiles. These asset-based measures represent the wealth distribution relative to other households within each country. They are widely used and are consistent with comparisons to household expenditures and the measurement of inequalities in child mortality, education and healthcare use in low-income and middle-income countries.²⁷ The detail description of the variables included in this study along with Directed Acyclic Graph (DAG) are presented in online web appendices 1-5.

Statistical analysis

All analyses were weighted to be nationally representative and the analyses involved two phases. First, the modified Poisson regression models (using generalised linear model with a log link and a Poisson distribution in STATA²⁸) was used for each country to calculate unadjusted and adjusted prevalence ratios (aPR) and their 95% CIs for each breastfeeding indicator (early initiation of breastfeeding, exclusive breastfeeding under 6 months and children ever breastfed) associated with individual-level caesarean delivery versus vaginal birth. The formula for the modified Poisson regression is given by Zou²⁹:

$$\log\left[\pi\left(x_{i}\right)\right] = \alpha + \beta x$$

Second, meta-analyses were done to obtain pooled aPR of these associations for each breastfeeding indicator. To account for between-country variation in effect estimates, we conducted random-effects meta-analysis using inverse variance weighting using 'metan' command in STATA. All the 33 countries data analysed individually were included in the meta-analyses of the effect of caesarean section on early initiation of breastfeeding and children ever breastfed. However, for exclusive breastfeeding under 6 months, data from Chad were unable to be included because of small sample with almost no exclusive breastfeeding under 6 months and the model could not coverage. We used the I² to measure statistical heterogeneity and the possible sources of heterogeneity were explored using a post hoc subgroup analyses based on the following subgroups: (1) categorising the countries by region-East Africa, Southern Africa, West Africa and Central Africa; (2) by rate of caesarean section categories—low (<5%), medium (5%–15%) and high (>15%) and (3) by prevalence of early initiation of breastfeeding categories—≤50% and>50%. The subcategories for geographic regions are according to United Nations geoscheme while the subcategories for caesarean section rates and prevalence of early initiation of breastfeeding were decided based on a previous study³⁰ and expert opinion, respectively. All analyses in this study were conducted using STATA/SE V.15.1 (Stata Corporation).



Figure 1 Flow scheme for country selection based on inclusion and exclusion criteria. DHS, Demographic and Health Surveys.

Table 1Among last-born children who were born in the last 2 years before the survey, percentage of live singleton births
delivered by caesarean section, prevalence of early initiation of breastfeeding, prevalence of exclusive breastfeeding under 6
months and prevalence of children ever breastfed, 33 sub-Saharan African countries (2010–2017/2018)

Country	Year of DHS	Caesarean section rate (%)	Prevalence of early initiation of BF (%)	Prevalence of exclusive BF under 6 months (%)	Prevalence of children ever breastfed (%)
Ethiopia	2016	2.6	73.5	57.8	96.9
Namibia	2013	15.6	71.5	48.7	95.8
Burkina Faso	2010	2.0	42.3	24.7	99.2
Burundi	2016–2017	5.3	85.4	83.6	99.0
Тодо	2013–2014	7.1	60.8	57.1	98.1
Benin	2017–2018	4.9	54.3	41.5	96.7
Cameroon	2011	4.7	39.8	20.2	97.4
Comoros	2012	11.4	33.8	12.5	93.1
Congo Bra	2011–2012	6.6	23.5	20.6	94.8
Congo DR	2013–2014	5.1	52.2	47.7	98.2
Cote d'Ivoire	2011–2012	2.9	31.0	12.4	96.8
Gabon	2012	10.1	32.6	6.1	89.9
Senegal	2010–2011	4.8	48.3	39.2	97.8
Gambia	2013	1.9	51.7	46.8	98.8
Ghana	2014	12.0	55.6	52.7	98.4
Guinea	2012	3.0	16.4	20.7	98.1
Liberia	2013	4.1	61.6	55.8	98.4
Mali	2012–2013	2.9	57.9	33.8	97.3
Mozambique	2011	3.9	76.8	41.3	97.4
Niger	2012	1.2	53.2	23.3	98.9
Nigeria	2013	2.1	33.3	17.5	97.9
Rwanda	2014–2015	13.0	80.8	88.0	98.8
Sierra Leone	2013	3.8	54.4	32.5	97.2
Zambia	2014–2015	4.5	65.9	73.0	97.8
Chad	2014–2015	1.5	23.0	0.3	98.1
Angola	2015–2016	3.9	48.4	37.5	95.0
Tanzania	2015–2016	6.3	51.5	59.2	98.4
Zimbabwe	2015	5.7	58.0	47.3	98.4
Malawi	2015–2016	6.4	76.7	61.2	97.9
Lesotho	2014	9.9	65.3	66.9	95.4
Kenya	2014	8.0	62.3	61.9	98.8
Uganda	2016	7.2	66.2	65.6	97.6
South Africa	2016	24.3	67.6	32.2	84.4

BF, breastfeeding; Congo Bra, Congo-Brazzaville; Congo DR, Democratic Republic of the Congo; DHS, Demographic and Health Surveys.

Patient and public involvement

No patients were involved in setting the research question or the outcome measures nor were they involved in the design and implementation of the study. No patients were asked to advise on interpretation or writing up of results. There are no plans to disseminate the results of the research to study participants or the relevant patient community.

RESULTS

The preparation of a DAG a priori helped identify the minimum set of variables needed to reduce confounding in the association between the exposure (caesarean section) and the outcomes (breastfeeding indicators). The DAG is shown in online supplementary appendix 1.

Data were eligible for inclusion in this study if they were from a sub-Saharan Africa and involved standardised DHS

6

and all search the second as the first

a all shake at the

Country	Year of DHS	Sample size	Unadjusted PR (95% CI)	Adjusted PR* (95% CI)
Ethiopia	2016	4021	0.50 (0.35 to 0.71)	0.46 (0.32 to 0.66)
Namibia	2013	2021	0.71 (0.62 to 0.81)	0.71 (0.62 to 0.80)
Burkina Faso	2010	5745	0.90 (0.68 to 1.18)	0.77 (0.60 to 1.00)
Burundi	2016–2017	5182	0.40 (0.34 to 0.48)	0.40 (0.33 to 0.47
Тодо	2013-2014	2707	0.60 (0.48 to 0.75)	0.55 (0.45 to 0.68
Benin	2017–2018	5337	0.46 (0.37 to 0.58)	0.47 (0.38 to 0.59
Cameroon	2011	4604	0.86 (0.68 to 1.09)	0.82 (0.65 to 1.04
Comoros	2012	1228	0.50 (0.31 to 0.81)	0.52 (0.33 to 0.83
Congo Bra	2011-2012	3754	0.42 (0.22 to 0.82)	0.44 (0.23 to 0.85
Congo DR	2013–2014	7189	0.47 (0.36 to 0.60)	0.44 (0.34 to 0.57
Cote d'Ivoire	2011-2012	3037	0.44 (0.24 to 0.81)	0.48 (0.26 to 0.90
Gabon	2012	2452	0.27 (0.14 to 0.50)	0.29 (0.15 to 0.56
Senegal	2010-2011	4809	0.59 (0.43 to 0.81)	0.57 (0.42 to 0.78
Gambia	2013	3429	0.69 (0.43 to 1.11)	0.77 (0.47 to 1.25
Ghana	2014	2281	0.47 (0.36 to 0.60)	0.46 (0.35 to 0.59
Guinea	2012	2748	0.49 (0.16 to 1.50)	0.47 (0.15 to 1.46
Liberia	2013	3001	0.43 (0.28 to 0.65)	0.45 (0.30 to 0.69
Mali	2012–2013	3884	0.64 (0.50 to 0.82)	0.60 (0.47 to 0.77
Mozambique	2011	4519	0.80 (0.70 to 0.91)	0.79 (0.70 to 0.90
Niger	2012	4668	0.75 (0.54 to 1.03)	0.58 (0.42 to 0.81
Nigeria	2013	12175	0.62 (0.47 to 0.81)	0.49 (0.38 to 0.64
Rwanda	2014–2015	3127	0.50 (0.45 to 0.57)	0.50 (0.45 to 0.57
Sierra Leone	2013	4569	0.70 (0.56 to 0.87)	0.68 (0.56 to 0.84
Zambia	2014–2015	5013	0.58 (0.48 to 0.71)	0.56 (0.47 to 0.68
Chad	2014–2015	6493	0.32 (0.13 to 0.75)	0.35 (0.15 to 0.82
Angola	2015–2016	5738	0.52 (0.36 to 0.74)	0.47 (0.34 to 0.67
Tanzania	2015–2016	4153	0.29 (0.21 to 0.41)	0.24 (0.17 to 0.33
Zimbabwe	2015	2330	0.36 (0.26 to 0.51)	0.35 (0.25 to 0.49
Malawi	2015–2016	6561	0.66 (0.58 to 0.74)	0.67 (0.60 to 0.75
Lesotho	2014	1368	0.58 (0.45 to 0.75)	0.59 (0.46 to 0.76
Kenya	2014	3762	0.49 (0.38 to 0.61)	0.49 (0.39 to 0.61
Uganda	2016	5892	0.54 (0.46 to 0.63)	0.49 (0.42 to 0.57
South Africa	2016	1358	0.85 (0.75 to 0.96)	0.89 (0.78 to 1.00

*Adjusted for pregnancy planning, birth weight, region of residence, sex of child, mother's age at birth, mother's education, number of antenatal visits, birth order, maternal tobacco use, place of delivery, types of residence (rural/urban), distance to health facility, mother's occupation and household wealth.

Congo Bra, Congo-Brazzaville; Congo DR, Democratic Republic of the Congo; DHS, Demographic and Health Surveys; PR, prevalence ratio.

survey collection processes that enabled calculation of the breastfeeding indicators. Furthermore, to ensure contemporaneous analyses, only surveys with data collected from 2010 or later were included. The flow scheme for country selection is displayed in figure 1.

A descriptive overview of the data is presented in table 1. The last three columns provide the prevalence of breastfeeding indicators while the third column provides

the caesarean section rates among singleton last-born children who were born in the past 2 years before the survey. The proportion of live singleton births delivered by caesarean section ranged from 1.2% in Niger to 24.3% in South Africa. The prevalence of early initiation of breastfeeding was highest in Burundi (85.4%) and lowest in Guinea at 16.4%. The prevalence of exclusive breastfeeding under 6 months varied from 0.3% in Chad


Figure 2 Association between caesarean birth and early initiation of breastfeeding in sub-Saharan Africa. aPR, adjusted prevalence ratio; Congo Bra, Congo-Brazzaville; Congo DR, Democratic Republic of the Congo.

to 88.0% in Rwanda. The prevalence of children ever breastfed is high across sub-Saharan African countries, with 97% of these countries have a prevalence of more than 90%.

Early initiation of breastfeeding

At the individual country level, the fully adjusted analyses in table 2 show that, compared with vaginal birth, caesarean section was associated with aPR for early initiation of breastfeeding that ranged from 0.24 (95% CI 0.17 to 0.33) in Tanzania to 0.89 (95% CI 0.78 to 1.00) in South Africa. Figure 2 shows the random-effects meta-analysis where caesarean birth was associated with a 46.0% lower prevalence of early initiation of breastfeeding (pooled aPR, 0.54 (95% CI 0.48 to 0.60)). However, the heterogeneity associated with pooled estimates for early initiation of breastfeeding was high (χ^2 p value=0.000, I²=84.5%).

Subgroup analyses

Subgroup analyses were conducted to understand the possible source of heterogeneity associated with pooled estimate of the association between caesarean section and early initiation of breastfeeding based on geographic regions, rate of caesarean section and prevalence of early initiation of breastfeeding. These investigations did not indicate that any of these variables are important sources of heterogeneity (ie, the heterogeneity remains high). The forest plots from these subgroups investigation are presented in online web appendix 6, figures A-C.

Exclusive breastfeeding under 6 months

At the individual country level, the fully adjusted analyses in table 3 show that, compared with vaginal birth, caesarean section was associated with aPR for exclusive breastfeeding under 6 months that ranged from 0.58 (95% CI 0.34 to 0.98) in Angola to a high aPR of 1.93 (95% CI 0.46 to 8.10) in Cote d'Ivoire. Figure 3 shows the random-effects meta-analysis where caesarean birth was associated with a 6.0% lower prevalence of exclusive breastfeeding under 6 months (pooled aPR, 0.94 (95% CI 0.88 to 1.01)).

Children ever breastfed

At the individual country level, the fully adjusted analyses in table 4 show that, compared with vaginal birth, caesarean section was associated with aPR for children ever breastfed that ranged from 0.91 (95% CI 0.82 to 1.02) in Gabon to 1.02 (95% CI 0.98 to 1.06) in Guinea and 1.02 (95% CI 0.99 to 1.04) in Gambia. Figure 4 shows the random-effects meta-analysis which indicated a 2% lower prevalence of children ever breastfed among caesarean versus vaginally born children (pooled aPR, 0.98 (95% CI 0.98 to 0.99)).

DISCUSSION Main finding

Main findings

Our study examined the impact of caesarean section on breastfeeding indicators-early initiation of breastfeeding, exclusive breastfeeding under 6 months and children ever breastfed-in sub-Saharan Africa. The within-country aPR for early initiation of breastfeeding and exclusive breastfeeding under 6 months varies widely across countries but not for children ever breastfed. On the other hand, our finding in meta-analysis, combining the 33 countries data from sub-Saharan Africa shows that caesarean section had a negative influence on early initiation of breastfeeding (a 46% reduction in the prevalence of early initiation of breastfeeding). Our other meta-analyses showed little difference in exclusive breastfeeding under 6 months and children ever breastfed between infants born by caesarean section versus vaginal birth in sub-Saharan Africa.

Interventions such as physical and psychological support for women to initiate and establish successful breastfeeding after caesarean birth may be essential because early initiation of breastfeeding is linked to a greater success in establishing breastfeeding, helps the uterus to return to its normal size quickly, guarantees that the newborn receive 'colostrum' (the first breastmilk), avoids baby needing artificial feeds and reduces neonatal mortality.^{31–33}

In this study, the rate of caesarean section in the past 2years before the surveys ranged from 1.2% in Niger to 24.3% in South Africa. This figure may reveal that caesarean section rate is low among some countries in sub-Saharan Africa which may be due to inadequate access, especially among the poorest women.³⁴ The prevalence

Caesarean versus vaginai birtins					
Country	Year of DHS	Sample size	Unadjusted PR (95% CI)	(95% CI)	
Ethiopia	2016	1081	0.87 (0.55 to 1.37)	0.80 (0.50 to 1.26)	
Namibia	2013	517	0.88 (0.66 to 1.19)	0.90 (0.66 to 1.22)	
Burkina Faso	2010	1433	1.01 (0.48 to 2.14)	0.95 (0.46 to 1.95)	
Burundi	2016–2017	1213	0.95 (0.84 to 1.08)	0.97 (0.86 to 1.09)	
Тодо	2013–2014	589	1.24 (0.99 to 1.55)	1.10 (0.88 to 1.38)	
Benin	2017–2018	1339	0.99 (0.69 to 1.42)	1.01 (0.73 to 1.39)	
Cameroon	2011	1094	2.18 (1.37 to 3.47)	1.64 (1.06 to 2.52)	
Comoros	2012	318	1.13 (0.40 to 3.19)	1.29 (0.49 to 3.35)	
Congo Bra	2011-2012	934	0.78 (0.34 to 1.80)	1.02 (0.38 to 2.78)	
Congo DR	2013–2014	1895	0.88 (0.60 to 1.29)	0.75 (0.51 to 1.11)	
Cote d'Ivoire	2011-2012	760	2.51 (1.03 to 6.09)	1.93 (0.46 to 8.10)	
Gabon	2012	621	1.05 (0.20 to 5.55)	1.10 (0.15 to 8.05)	
Senegal	2010-2011	1308	1.18 (0.79 to 1.78)	1.14 (0.75 to 1.71)	
Gambia	2013	939	0.80 (0.37 to 1.72)	0.84 (0.39 to 1.79)	
Ghana	2014	598	0.98 (0.72 to 1.32)	1.01 (0.74 to 1.37)	
Guinea	2012	701	0.98 (0.23 to 4.13)	0.67 (0.19 to 2.45)	
Liberia	2013	708	1.36 (1.01 to 1.84)	1.46 (1.09 to 1.97)	
Mali	2012-2013	984	1.51 (1.01 to 2.25)	1.53 (1.02 to 2.31)	
Mozambique	2011	1025	0.91 (0.57 to 1.45)	0.84 (0.54 to 1.30)	
Niger	2012	1281	0.83 (0.30 to 2.24)	0.69 (0.24 to 1.92)	
Nigeria	2013	2881	1.69 (1.02 to 2.82)	0.77 (0.48 to 1.26)	
Rwanda	2014–2015	690	0.91 (0.81 to 1.01)	0.92 (0.82 to 1.02)	
Sierra Leone	2013	1095	1.11 (0.68 to 1.82)	1.30 (0.85 to 1.99)	
Zambia	2014–2015	1170	0.84 (0.65 to 1.09)	0.89 (0.68 to 1.17)	
Angola	2015–2016	1588	0.65 (0.36 to 1.18)	0.58 (0.34 to 0.98)	
Tanzania	2015–2016	997	0.80 (0.58 to 1.09)	0.76 (0.55 to 1.07)	
Zimbabwe	2015	590	0.95 (0.61 to 1.47)	1.09 (0.73 to 1.64)	
Malawi	2015-2016	1605	1.20 (1.03 to 1.39)	1.16 (0.99 to 1.37)	
Lesotho	2014	322	0.96 (0.68 to 1.34)	1.10 (0.85 to 1.44)	
Kenya	2014	845	0.79 (0.55 to 1.13)	0.73 (0.53 to 1.00)	
Uganda	2016	1459	0.91 (0.75 to 1.10)	0.90 (0.75 to 1.09)	
South Africa	2016	339	0.69 (0.42 to 1.13)	0.76 (0.49 to 1.19)	

and a local to a large a state of all as

*Adjusted for pregnancy planning, birth weight, region of residence, sex of child, mother's age at birth, mother's education, number of antenatal visits, birth order, maternal tobacco use, place of delivery, types of residence (rural/urban), distance to health facility, mother's occupation and household wealth.

Congo Bra, Congo-Brazzaville; Congo DR, Democratic republic of the Congo; DHS, Demographic and Health Surveys; PR, prevalence ratio.

of early initiation of breastfeeding ranged from 16.4% to 85.4% across the 33 sub-Saharan African countries with one-third of these countries have a prevalence of less than 50%, suggesting a considerable proportion of newborns are not breastfed within 1 hour postbirth. Similarly, there were a substantial disparities in the prevalence of exclusive breastfeeding under 6 months in sub-Saharan Africa (ranged from 0.3% to 88.0%), with a very low prevalence in Chad (0.3%) and Gabon (6.1%). The very low

prevalence of exclusive breastfeeding under 6 months in Chad and Gabon may warrant the need for urgent interventions such as improving counselling skills of health workers on exclusive breastfeeding and increasing the family and/or community support for breastfeeding³⁵ in these countries. However, the prevalence of children ever breastfed is high (90% plus) among the majorities of countries (ie, 32 countries) and ranged from 84.4% in South Africa to 99.2% in Burkina Faso.



Figure 3 Association between caesarean birth and exclusive breastfeeding under 6 months in sub-Saharan Africa. aPR, adjusted prevalence ratio; Congo Bra, Congo-Brazzaville; Congo DR, Democratic Republic of the Congo.

Early initiation of breastfeeding

Our within-country adjusted analyses and random-effects meta-analysis of 33 countries revealed that the prevalence of early initiation of breastfeeding was lower among infants delivered by caesarean section compared with infants born vaginally. This finding is consistent with a previous study conducted in Canada that reported more women who underwent planned caesarean birth did not initiate breastfeeding.¹⁹ Moreover, Takahashi et al⁸⁶ in their study conducted based on WHO global surveys completed in 24 countries in Africa, Latin America and Asia found that caesarean delivery was negatively associated with the rate of early initiation of breastfeeding. It has been suggested that early initiation of breastfeeding has numerous benefits for newborns. For example, early initiation of breastfeeding, which can be supported and facilitated by skin-to-skin contact between mother and infant, may promote exclusive breastfeeding under 6 months of life. Findings from our meta-analysis which showed a 46% reduction in the practice of early initiation of breastfeeding among infants born by caesarean versus vaginal births is of public health importance, and health programmes and healthcare providers in low-income and middle-income countries should consider interventions to promote and support early initiation of breastfeeding.

Support to improve early initiation of breastfeeding should be a priority to improve neonatal survival and enhance long-term health of infants. Not initiating breastfeeding within an hour is associated with an increased risks of neonatal mortality and reduced opportunity of benefiting from the immune properties that the 'colostrum' provides to the newborns. $^{32\,33}$

An earlier meta-analysis that included studies conducted outside of Africa²⁰ reported a similar result to ours: the negative effect of caesarean section on early initiation of breastfeeding is more pronounced among children born after caesarean section than after vaginal birth. Interventions such as immediate or early skin-to-skin contact, parent education and use of hand expressed breastmilk (to establish and maintain an adequate milk supply) may help mothers to practice early initiation of breastfeeding following caesarean birth in sub-Saharan Africa.

Exclusive breastfeeding under 6 months and children ever breastfed

The present study showed that the within-country aPR for exclusive breastfeeding under 6 months associated with caesarean versus vaginally born infants varied from a low aPR of 0.58 (95% CI 0.34 to 0.98) in Angola to a high aPR of 1.93 (95% CI 0.46 to 8.10) in Cote d'Ivoire. These findings suggest that caesarean section favours the practice of exclusive breastfeeding under 6 months for some countries, while it has adverse effects for others. These associations would appear to be counter-intuitive, given that caesarean section may be a barrier for exclusive breastfeeding under 6 months, yet given that institutional delivery, where the procedure is performed, caesarean section is among the facilitators of exclusive breastfeeding.³⁷ In developing countries, most poor women give birth at home assisted by traditional birth attendants. Cultural beliefs and traditional feeding practices may prevent women from initiating breastfeeding immediately postbirth, which, in turn, affects the practice of exclusive breastfeeding under 6 months.^{2 38} Therefore, when poor and/or uneducated women undergo caesarean section in developing countries, there is a possibility that women fed their newborn exclusively breast milk because health workers provide immediate breastfeeding support at delivery in health facilities. This was confirmed by a previous study which reported interventions such as individual immediate breastfeeding support at delivery, counselling or group education and lactation management increase exclusive breastfeeding by 49% (95% CI 33 to 68) and any breastfeeding by 66%(95% CI 34 to 107).³⁹

On the other hand, the findings of our meta-analysis suggested little influence in exclusive breastfeeding under 6 months (pooled aPR, 0.94 (95% CI 0.88 to 1.01) and children ever breastfed (pooled aPR, 0.98 (95% CI 0.98 to 0.99) among caesarean versus vaginally born infants in sub-Saharan Africa. The current findings are comparable with two previous studies which found that caesarean section has little influence on exclusive breastfeeding (risk ratio, 1.08; 95% CI 0.82 to 1.41)³⁹ and any breastfeeding practice at 6 months among mothers who initiated breastfeeding (pooled OR, 0.95; 95% CI 0.89 to 1.01).²⁰

Country	Year of DHS	Sample size	Unadjusted PR (95% CI)	Adjusted PR* (95% CI)
Ethiopia	2016	4021	0.93 (0.84 to 1.04)	0.93 (0.84 to 1.04
Namibia	2013	2021	0.99 (0.96 to 1.03)	1.00 (0.96 to 1.03
Burkina Faso	2010	5745	0.98 (0.94 to 1.01)	0.98 (0.94 to 1.02
Burundi	2016–2017	5182	0.98 (0.96 to 1.01)	0.98 (0.96 to 1.01
Тодо	2013-2014	2707	0.97 (0.94 to 1.00)	0.97 (0.94 to 1.0 ⁻
Benin	2017–2018	5337	0.99 (0.96 to 1.02)	0.99 (0.97 to 1.02
Cameroon	2011	4604	0.98 (0.95 to 1.01)	1.00 (0.96 to 1.03
Comoros	2012	1228	0.97 (0.91 to 1.04)	0.98 (0.92 to 1.04
Congo Bra	2011-2012	3754	0.93 (0.86 to 1.01)	0.94 (0.87 to 1.02
Congo DR	2013-2014	7189	0.98 (0.96 to 1.00)	0.98 (0.96 to 1.0 ⁻
Cote d'Ivoire	2011-2012	3037	0.97 (0.91 to 1.04)	0.99 (0.93 to 1.00
Gabon	2012	2452	0.90 (0.80 to 1.02)	0.91 (0.82 to 1.02
Senegal	2010-2011	4809	0.95 (0.91 to 1.00)	0.95 (0.91 to 1.0
Gambia	2013	3429	1.00 (0.98 to 1.02)	1.02 (0.99 to 1.04
Ghana	2014	2281	0.97 (0.94 to 1.00)	0.97 (0.94 to 1.0
Guinea	2012	2748	1.00 (0.97 to 1.03)	1.02 (0.98 to 1.0
Liberia	2013	3001	0.95 (0.88 to 1.02)	0.94 (0.88 to 1.0 ⁻
Mali	2012-2013	3884	1.00 (0.98 to 1.03)	1.01 (0.98 to 1.0
Mozambique	2011	4519	1.00 (0.98 to 1.02)	1.01 (0.99 to 1.04
Niger	2012	4668	0.96 (0.89 to 1.04)	0.97 (0.89 to 1.0
Nigeria	2013	12175	0.93 (0.89 to 0.97)	0.93 (0.89 to 0.98
Rwanda	2014–2015	3127	0.98 (0.97 to 1.00)	0.99 (0.97 to 1.00
Sierra Leone	2013	4569	0.94 (0.88 to 0.99)	0.93 (0.89 to 0.99
Zambia	2014–2015	5013	0.98 (0.95 to 1.01)	1.00 (0.96 to 1.03
Chad	2014–2015	6493	0.95 (0.89 to 1.01)	0.96 (0.90 to 1.03
Angola	2015–2016	5738	0.99 (0.94 to 1.05)	0.97 (0.92 to 1.0
Tanzania	2015-2016	4153	0.97 (0.94 to 1.00)	0.96 (0.93 to 1.0
Zimbabwe	2015	2330	0.98 (0.95 to 1.01)	0.98 (0.95 to 1.0
Malawi	2015-2016	6561	0.99 (0.96 to 1.01)	0.98 (0.96 to 1.0
Lesotho	2014	1368	0.96 (0.88 to 1.04)	0.98 (0.92 to 1.04
Kenya	2014	3762	1.01 (1.00 to 1.02)	1.01 (1.00 to 1.0
Uganda	2016	5892	0.98 (0.96 to 1.00)	0.98 (0.96 to 1.0
South Africa	2016	1358	0.93 (0.86 to 1.00)	0.96 (0.89 to 1.0

Table 4 Crude and multivariable-adjusted prevalence ratios for children ever breastfed, associated with caesarean versus

*Adjusted for pregnancy planning, birth weight, region of residence, sex of child, mother's age at birth, mother's education, number of antenatal visits, birth order, maternal tobacco use, place of delivery, types of residence (rural/urban), distance to health facility, mother's occupation and household wealth.

Congo Bra, Congo-Brazzaville; Congo DR, Democratic Republic of the Congo; DHS, Demographic and Health Surveys; PR, prevalence ratio.

Strengths and limitations of this study

The strength of the present study include the use of nationally representative data on a large number of countries (n=33 countries) over a long time period (2010-2017/2018). As the small number of previous studies on caesarean section and breastfeeding are from countries outside the African continent, the present study provides insight into the link between caesarean section and breastfeeding in low-income and middle-income countries in

sub-Saharan Africa. This is important because caesarean section rate is increasing in low-income and middle-income countries, and early initiation of breastfeeding may be compromised when mothers undergo caesarean section due to several factors such as mothers' emotional health and restricted mobility in the early days after the surgery. Moreover, in addition to the within-country adjusted analyses of nationally representative surveys data, we have also summarised our findings in a meta-analysis,

Country	Year	aPR (95% CI)	Weight (%)
Ethiopia	2016	• 0.93 (0.84 to 1.04)	0.63
Namibia	2013	1.00 (0.96 to 1.03)	3.27
Burkina Faso	2010		2.77
Burundi	2016-17		4.57
Togo	2013-14	0.97 (0.94 to 1.01)	3.27
Benin	2017-18	- 0.99 (0.97 to 1.02)	4.57
Cameroon	2011	1.00 (0.96 to 1.03)	3.27
Comoros	2012	0.98 (0.92 to 1.04)	1.53
Congo Bra	2011-12	0.94 (0.87 to 1.02)	1.06
Congo DR	2013-14	- 0.98 (0.96 to 1.01)	4.57
Cote d'Ivoire	2011-12	0.99 (0.93 to 1.06)	1.35
Gabon	2012	• 0.91 (0.82 to 1.02)	0.63
Senegal	2010-11	0.95 (0.91 to 1.00)	2.37
Gambia	2013	1.02 (0.99 to 1.04)	4.57
Ghana	2014	0.97 (0.94 to 1.01)	3.27
Guinea	2012	1.02 (0.98 to 1.06)	2.77
Liberia	2013 -	0.94 (0.88 to 1.01)	1.35
Mali	2012-13	1.01 (0.98 to 1.03)	4.57
Mozambique	2011	1.01 (0.99 to 1.04)	4.57
Niger	2012 -	0.97 (0.89 to 1.05)	0.94
Nigeria	2013 -	0.93 (0.89 to 0.98)	2.37
Rwanda	2014-15	 0.99 (0.97 to 1.00) 	6.23
Sierra Leone	2013 -	0.93 (0.89 to 0.99)	2.03
Zambia	2014-15	1.00 (0.96 to 1.03)	3.27
Chad	2014-15	0.96 (0.90 to 1.03)	1.35
Angola	2015-16	0.97 (0.92 to 1.02)	2.03
Tanzania	2015-16	0.96 (0.93 to 1.00)	3.27
Zimbabwe	2015	0.98 (0.95 to 1.01)	3.86
Malawi	2015-16		4.57
Lesotho	2014	0.98 (0.92 to 1.04)	1.53
Kenya	2014	 1.01 (1.00 to 1.02) 	7.02
Uganda	2016		5.38
South Africa	2016 -	0.96 (0.89 to 1.03)	1.19
Overall (I-square	d = 49.2%, p = 0.001)	0.98 (0.98 to 0.99)	100.00
NOTE: Weights a	re from random effects analysis		
	.8	1 1.1	
	Favours caesarean bir	th Favours vaginal birth	

Figure 4 Association between caesarean birth and children ever breastfed in sub-Saharan Africa. aPR, adjusted prevalence ratio; Congo Bra, Congo-Brazzaville; Congo DR, Democratic Republic of the Congo.

combining data analysed individually to provide a consolidated effect estimate of the impact of caesarean section on breastfeeding indicators in sub-Saharan Africa.

We acknowledge the following limitations of this study. First, despite the high-quality of DHS data in low-income and middle-income countries, the ascertainment of breastfeeding practice depends on retrospective maternal reports, which is at risk of misreporting. Nevertheless, a study in Mexico⁴⁰ showed that the reliability of retrospective maternal reports of any breastfeeding practice was high (intraclass correlation coefficient=0.94). Furthermore, a review of 10 studies from developed and developing countries suggested a less than 3-year maternal recall period of any breastfeeding and breastfeeding duration is valid and reliable but did not include any study specifically assessing early initiation of breastfeeding.⁴¹ On the other hand, the proportion of children exclusively breastfed may be overestimated because exclusive breastfeeding under 6 months is ascertained based on 'previous day recall period', and some children who are given other liquids irregularly may not have received other liquids or solid foods before the survey day.¹¹ A study from South Africa also revealed that maternal recall of exclusive breastfeeding is limited in accuracy when compared with the 'objective stable isotope method' for assessing exclusive breastfeeding.⁴² Furthermore, DHS in developing countries are prone to incomplete and inconsistent reporting. Nonetheless, to overcome this problem, the DHS Program performs extensive data editing, including imputation of incomplete dates⁴³ and make the final edited data files accessible to data users worldwide.

Second, DHS data do not distinguish whether the caesarean section was medically indicated or not. This would be useful to study in the future as it would be important to know whether delayed initiation of breastfeeding occurs in non-medically indicated caesarean birth, as this could be addressed in institutional settings. It may be more challenging to address delayed initiation of breastfeeding among medically indicated emergency caesarean section if mothers or neonates are too sick to commence breastfeeding. However, mothers should be supported to start early expression of breast milk to establish and maintain an adequate milk supply when the newborn is too ill to breastfeed, or if the obstetric complication prevents mother from breastfeeding directly. Third, as DHS lack data on income and expenditures to measure socioeconomic position (SEP) of the household, we have used wealth index as measure of relative SEP of the household. While previous studies suggests that asset-based index is resistant to most economic shocks and is less variable in response to income and expenditure fluctuations, the wealth index can be considered to be a more stable measure of SEP than consumption expenditure in low-income and middle-income countries.44 45

Finally, heterogeneity associated with pooled estimates for 'early initiation of breastfeeding' among caesarean versus vaginally born infants is high and the source/s of heterogeneity remain unclear after subgroups analyses performed to explore why the effect estimates differ. Although we cannot be sure about the source of heterogeneity, the clinical presentation of mothers and newborn during the perinatal period may be considered as one possible source. For example, for women who delivered by emergency caesarean section after severe obstetric complication, initiation of breastfeeding with an hour may be very difficult or impossible, while for women with less labour complication delivered by emergency caesarean section or planned caesarean section, initiation of breastfeeding within 1 hour after birth may be a possibility. These scenarios may suggest clinical heterogeneity between the analyses performed in each country because of the clinical difference (eg, due to confounding by indication) between countries data analysed. It has previously been suggested that clinical heterogeneity (ie, clinical deference between studies or trials) should be explored.⁴⁶

CONCLUSIONS

Caesarean section had a negative influence on early initiation of breastfeeding but showed little association with exclusive breastfeeding under 6 months and children ever breastfed among infants born after caesarean section versus vaginal birth in sub-Saharan Africa. Health interventions to promote and support women to initiate and maintain breastfeeding after caesarean birth should be considered.

6

Open access

Author affiliations

¹School of Public Health, The University of Adelaide, Adelaide, South Australia, Australia

²School of Allied Health Sciences, College of Health Sciences, Addis Ababa University, Addis Ababa, Ethiopia

³Robinson Research Institute, School of Medicine, The University of Adelaide, Adelaide, South Australia, Australia

⁴Department of Obstetrics and Gynaecology, Monash University, Melbourne, Victoria, Australia

⁵Population Health Sciences, University of Bristol, England, UK

Acknowledgements The authors are grateful to The DHS Program for providing the data sets used for this analysis. BWM is supported by an NHMRC Practitioner Fellowship (GNT1082548) and also reports consultancy for ObsEva, Merck Merck KGaA, and Guerbet. JWL is supported by an NHMRC Centre of Research Excellence (GNT1099422).

Contributors All authors contributed to the design of the study and the interpretation of data. EY performed the data analysis and drafted the manuscript. All other authors critically revised the draft manuscript. All authors read and approved the final manuscript. EY is the guarantor of the paper.

Funding The first author is fully supported by an Australian Government Research Training Programme (RTP) Scholarship.

Competing interests None declared.

Patient consent for publication Not required.

Provenance and peer review Not commissioned; externally peer reviewed.

Data availability statement Data are available in a public, open access repository.

Open access This is an open access article distributed in accordance with the Creative Commons Attribution Non Commercial (CC BY-NC 4.0) license, which permits others to distribute, remix, adapt, build upon this work non-commercially, and license their derivative works on different terms, provided the original work is properly cited, appropriate credit is given, any changes made indicated, and the use is non-commercial. See: http://creativecommons.org/licenses/by-nc/4.0/.

REFERENCES

- Victora CG, Bahl R, Barros AJD, et al. Breastfeeding in the 21st century: epidemiology, mechanisms, and lifelong effect. *The Lancet* 2016;387:475–90.
- 2. Sharma IK, Byrne A. Early initiation of breastfeeding: a systematic literature review of factors and barriers in South Asia. *Int Breastfeed J* 2016;11:17.
- 3. Bowatte G, Tham R, Allen KJ, *et al*. Breastfeeding and childhood acute otitis media: a systematic review and meta-analysis. *Acta Paediatr* 2015;104:85–95.
- Giugliani ERJ, Horta BL, Loret de Mola C, et al. Effect of breastfeeding promotion interventions on child growth: a systematic review and meta-analysis. Acta Paediatr 2015;104:20–9.
- Horta BL, Victora CG, World Health Organization. Short-Term effects of breastfeeding: a systematic review on the benefits of breastfeeding on diarrhoea and pneumonia mortality, 2013. Available: http://www.who.int/iris/handle/10665/95585
- Lodge CJ, Tan DJ, Lau MXZ, et al. Breastfeeding and asthma and allergies: a systematic review and meta-analysis. Acta Paediatr 2015;104:38–53.
- Ip S, Chung M, Raman G, et al. A summary of the agency for healthcare research and quality's evidence report on breastfeeding in developed countries. *Breastfeed Med* 2009;4(Suppl 1):S17–30.
- Aune D, Norat T, Romundstad P, et al. Breastfeeding and the maternal risk of type 2 diabetes: a systematic review and dose– response meta-analysis of cohort studies. Nutr Metab Cardiovasc Dis 2014;24:107–15.
- Chowdhury R, Sinha B, Sankar MJ, et al. Breastfeeding and maternal health outcomes: a systematic review and meta-analysis. Acta Paediatr 2015;104:96–113.
- 10. World Health Organization. *Guideline: protecting, promoting and supporting breastfeeding in facilities providing maternity and newborn services*. Geneva: World Health Organization, 2017.
- WHO, UNICEF, USAID, AED, UCDAVIS, IFPRI. Indicators for assessing infant and young child feeding practices, Part I: Definitions. Conclusions of a consensus meeting held 6–8 November 2007. Washington D.C, USA. Geneva: World Health Organization,

2008. https://www.who.int/nutrition/publications/infantfeeding/ 9789241596664/en/

- Tully KP, Ball HL. Postnatal unit bassinet types when rooming-in after cesarean birth: implications for breastfeeding and infant safety. J Hum Lact 2012;28:495–505.
- Tully KP, Ball HL. Maternal accounts of their breast-feeding intent and early challenges after caesarean childbirth. *Midwifery* 2014;30:712–9.
- 14. Karlström A, Engström-Olofsson R, Norbergh K-G, *et al.* Postoperative pain after cesarean birth affects breastfeeding and infant care. *J Obstet Gynecol Neonatal Nurs* 2007;36:430–40.
- Kolås T, Saugstad OD, Daltveit AK, et al. Planned cesarean versus planned vaginal delivery at term: Comparison of newborn infant outcomes. *Am J Obstet Gynecol* 2006;195:1538–43.
- Zanardo V, Svegliado G, Cavallin F, et al. Elective cesarean delivery: does it have a negative effect on breastfeeding? Birth 2010;37:275–9.
- Evans KC, Evans RG, Royal R, et al. Effect of caesarean section on breast milk transfer to the normal term newborn over the first week of life. Arch Dis Child Fetal Neonatal Ed 2003;88:380F–2.
- Hyde MJ, Mostyn A, Modi N, et al. The health implications of birth by caesarean section. *Biological Reviews* 2012;87:229–43.
- Hobbs AJ, Mannion CA, McDonald SW, et al. The impact of caesarean section on breastfeeding initiation, duration and difficulties in the first four months postpartum. *BMC Pregnancy Childbirth* 2016;16:90.
- Prior E, Santhakumaran S, Gale C, et al. Breastfeeding after cesarean delivery: a systematic review and meta-analysis of world literature. Am J Clin Nutr 2012;95:1113–35.
- Boatin AA, Schlotheuber A, Betran AP, et al. Within country inequalities in caesarean section rates: observational study of 72 low and middle income countries. BMJ 2018;360.
- Ologunde R, Vogel JP, Cherian MN, et al. Assessment of cesarean delivery availability in 26 low- and middle-income countries: a crosssectional study. Am J Obstet Gynecol 2014;211:504.e1–12.
- Say L, Raine R. A systematic review of inequalities in the use of maternal health care in developing countries: examining the scale of the problem and the importance of context. *Bull World Health Organ* 2007;85:812–9.
- Holmer H, Lantz A, Kunjumen T, *et al.* Global distribution of surgeons, anaesthesiologists, and obstetricians. *Lancet Glob Health* 2015;3(Suppl 2):S9–11.
- Stanton CK, Dubourg D, De Brouwere V, et al. Reliability of data on caesarean sections in developing countries. Bull World Health Organ 2005;83:449–55.
- Filmer D, Pritchett LH. Estimating wealth effects without expenditure data--or tears: an application to educational enrollments in states of India. *Demography* 2001;38:115–32.
- 27. Filmer D, Scott K. Assessing asset indices. *Demography* 2012;49:359–92.
- Cummings P. Methods for estimating adjusted risk ratios. Stata J 2009;9:175–96.
- 29. Zou G. A modified poisson regression approach to prospective studies with binary data. *Am J Epidemiol* 2004;159:702–6.
- Kyu HH, Shannon HS, Georgiades K, et al. Caesarean delivery and neonatal mortality rates in 46 low- and middle-income countries: a propensity-score matching and meta-analysis of demographic and health survey data. Int J Epidemiol 2013;42:781–91.
- 31. Ballard O, Morrow AL. Human milk composition: nutrients and bioactive factors. *Pediatr Clin North Am* 2013;60:49–74.
- Smith ER, Hurt L, Chowdhury R, et al. Delayed breastfeeding initiation and infant survival: a systematic review and meta-analysis. *PLoS One* 2017;12:e0180722.
- Edmond KM, Zandoh C, Quigley MA, et al. Delayed breastfeeding initiation increases risk of neonatal mortality. *Pediatrics* 2006;117:e380–6.
- Boerma T, Ronsmans C, Melesse DY, *et al.* Global epidemiology of use of and disparities in caesarean sections. *The Lancet* 2018;392:1341–8.
- Kavle JA, LaCroix E, Dau H, et al. Addressing barriers to exclusive breast-feeding in low- and middle-income countries: a systematic review and programmatic implications. *Public Health Nutr* 2017;20:3120–34.
- Takahashi K, Ganchimeg T, Ota E, et al. Prevalence of early initiation of breastfeeding and determinants of delayed initiation of breastfeeding: secondary analysis of the WHO Global Survey. Sci Rep 2017;7:44868.
- Balogun OO, Dagvadorj A, Anigo KM, et al. Factors influencing breastfeeding exclusivity during the first 6 months of life in developing countries: a quantitative and qualitative systematic review. *Matern Child Nutr* 2015;11:433–51.

Open access

- Montagu D, Yamey G, Visconti A, et al. Where do poor women in developing countries give birth? A Multi-Country analysis of demographic and health survey data. PLoS One 2011;6:e17155.
- Rollins NC, Bhandari N, Hajeebhoy N, *et al.* Why invest, and what it will take to improve breastfeeding practices? *The Lancet* 2016;387:491–504.
- Cupul-Uicab LA, Gladen BC, Hernández-Ávila M, et al. Reliability of reported breastfeeding duration among reproductive-aged women from Mexico. *Matern Child Nutr* 2009;5:125–37.
- Li R, Scanlon KS, Serdula MK. The validity and reliability of maternal recall of breastfeeding practice. *Nutr Rev* 2005;63:103–10.
- Mulol H, Coutsoudis A. Limitations of maternal recall for measuring exclusive breastfeeding rates in South African mothers. *Int Breastfeed J* 2018;13:19.
- Croft T. "Date Editing and Imputation". In Demographic and Health Surveys World Conference Proceedings, II: 1337-1356, Columbia, Maryland: IRD/ORC Macro., 1991. Available: https://dhsprogram. com/publications/publication-DHSG3-DHS-Questionnaires-and-Manuals.cfm
- Howe LD, Galobardes B, Matijasevich A, et al. Measuring socioeconomic position for epidemiological studies in low- and middleincome countries: a methods of measurement in epidemiology paper. Int J Epidemiol 2012;41:871–86.
- Liverpool-Tasie LSO, Winter-Nelson A. Asset versus consumption poverty and poverty dynamics in rural Ethiopia. *Agricultural Economics* 2011;42:221–33.
- Thompson SG. Systematic review: why sources of heterogeneity in meta-analysis should be investigated. *BMJ* 1994;309:1351–5.

6

Chapter 7: Associations between Apgar scores and children's educational outcomes at eight years of age

7.1 Preface

This chapter contains the fourth of the five studies contributing to this thesis. This study was published in *The Australian and New Zealand Journal of Obstetrics and Gynaecology* in 2020.

In Chapter 5 and 6, the influence of obstetrical interventions during childbirth (more specifically caesarean section) on neonatal health and breastfeeding were identified. Others have shown that pre-labour caesarean section was found to reduce the postnatal rise in cortisol, a hormone that prepare the foetus for birth and supports the multi-organ extra-uterine transition.⁶³ Caesarean section may, therefore, influence neonatal survival during the transition from foetal to neonatal life.¹⁹⁶ The relative success of a newborn making the transition at birth could immediately be gauged by Apgar score, which is an overall measure of a baby's vitality at one and five minutes after birth.^{158,162} As bad experiences *in utero*, and poor conditions existing during childbirth may influence the normal transition, it is also thought that these poor conditions may predispose to poor cognitive performance at later-life.

In Chapter 7, the link between the transition at birth as measured by one-and five-minute Apgar scores and children's educational outcomes at eight years of age was examined using high-income country datasets. This is because there are no readily-available datasets with Apgar scores from births in most LMICs, including Ethiopia nor are there data linkage platforms to link perinatal data with school outcomes in LMICs. Using high-quality wholeof-population linked administrative data from South Australia, this chapter (Study 4) presents the results of novel analysis regarding the associations between Apgar scores of 0-5, 6, 7, 8 and 9 (compared with 10) and children's educational outcomes as measured by the National Assessment Program—Literacy and Numeracy (NAPLAN) tests at eight years of age.

7.2 Publication

Yisma E, Mol BW, Lynch JW, Murthy NM, Smithers LG. Associations between Apgar

scores and children's educational outcomes at eight years of age. Aust N Z J Obstet Gynaecol

2020; 1–7. https://doi.org/10.1111/ajo.13220

7.2.1 Statement of authorship

Title of Paper	Associations between Apgar scores and children's educational outcomes at eight years of age		
Publication Status	Published Accepted for Publication		
	Submitted for Publication Unpublished and Unsubmitted work written manuscript style		
Publication Details	Yisma E, Mol BW, Lynch JW, Murthy NM, Smithers LG. Associations between Apgar scores and children's educational outcomes at eight years of age. <i>Aust N Z J Obstet Gynaecol</i> 2020; 1–7. <u>https://doi.org/10.1111/ajo.13220</u>		

Principal Author

Name of Principal Author (Candidate)	Engida Yisma				
Contribution to the Paper	Contributed to the design of the study, performed statistical analysis, interpreted the data, and drafted the manuscript.				
Overall percentage (%)	80%				
Certification:	This paper reports on original research I conducted during the period of my Higher Degree by Research candidature and is not subject to any obligations or contractual agreements with a third party that would constrain its inclusion in this thesis. I am the primary author of this paper.				
Signature	Date 4-Aug-2020				

Co-Author Contributions

By signing the Statement of Authorship, each author certifies that:

- i. the candidate's stated contribution to the publication is accurate (as detailed above);
- ii. permission is granted for the candidate to include the publication in the thesis; and
- iii. the sum of all co-author contributions is equal to 100% less the candidate's stated contribution.

Name of Co-Author	Ben W Mol			
Contribution to the Paper	Contributed to the design of the study, interpreted the findings, and reviewed the manuscript.			
Signature	Date 04	4/08/2020		

Name of Co-Author	John W Lynch
Contribution to the Paper	Contributed to the design of the study, interpreted the findings, and reviewed the manuscript.
Signature	Date 6/8/20

Name of Co-Author	Murthy N Mittinty			
Contribution to the Paper	Contributed to the data analysis, interpretation and revision of the manuscript.			
Signature		Date	06/08/2020	

Name of Co-Author	Lisa G Smithers			
Contribution to the Paper	Contributed to the design of the study, interpreted the findings, and reviewed the manuscript.			
Signature		Date	5-Aug-2020	

DOI: 10.1111/ajo.13220

ORIGINAL ARTICLE

Associations between Apgar scores and children's educational outcomes at eight years of age

Engida Yisma^{1,2,3}, Ben W. Mol^{3,4}, John W. Lynch^{1,3,5}, Murthy N. Mittinty^{1,3} and Lisa G. Smithers^{1,3}

¹School of Public Health, The University of Adelaide, Adelaide, South Australia, Australia

²School of Allied Health Sciences, College of Health Sciences, Addis Ababa University, Addis Ababa, Ethiopia

³Robinson Research Institute, School of Medicine, The University of Adelaide, Adelaide, South Australia, Australia

⁴Department of Obstetrics and Gynaecology, Monash University, Melbourne, Victoria, Australia

⁵Population Health Sciences, Bristol Medical School, University of Bristol, Bristol, UK

Correspondence: Mr Engida Yisma, School of Public Health, The University of Adelaide, Mail Drop DX 650 550, Adelaide, SA 5005, Australia. Email: engida.derbie@adelaide. edu.au

Conflicts of Interest: The authors report no conflicts of interest.

Received: 11 February 2020; Accepted: 25 June 2020 **Background:** Low Apgar scores are associated with neonatal morbidity and mortality, but effects of Apgar scores of 0–5, 6, 7, 8 and 9 (compared with 10) on longerterm neurodevelopmental outcomes are less clear.

Aim: To examine the associations between Apgar scores of 0–5, 6, 7, 8 and 9 (compared with 10) and children's educational outcomes as measured by the Australian National Assessment Program—Literacy and Numeracy (NAPLAN) tests at age eight. **Materials and Methods:** We merged perinatal data including all children born in South Australia from 1999 to 2008 with school assessment data (NAPLAN). School assessments included five learning areas (domains)—reading, writing, spelling, grammar and numeracy. Each domain was categorised according to performing at or below National Minimum Standards (≤NMS). Effects were estimated using Augmented Inverse Probability Weighting (AIPW) accounting for a range of maternal, perinatal and sociodemographic characteristics.

Results: Risk differences comparing five-minute Apgar scores of 0–5 with Apgar scores of 10 for children performing \leq NMS for each domain were: reading (0.07 (95% CI –0.16 to 0.29)), writing (0.27 (95% CI –0.14 to 0.68)), spelling (0.15 (95% CI –0.10 to 0.40)), grammar (0.04 (95% CI –0.21 to 0.29)) and numeracy (0.21 (95% CI –0.04 to 0.45)). Risk differences for children performing \leq NMS were also evident when Apgar score of 6 was compared with Apgar score of 10.

Conclusions: Children with five-minute Apgar scores of 0–5 and 6, compared with Apgar score of 10, are at higher risk of scoring at/below the NMS on the NAPLAN assessments at eight years.

KEYWORDS

Apgar score, cognition, epidemiology, school assessment

INTRODUCTION

The Apgar score was introduced in the early 1950s by Dr Virginia Apgar as an overall measure of a baby's vitality at one minute after birth.¹ For decades, the Apgar score has been routinely reported at one and five minutes after birth. The Apgar score comprises five physiological signs—heart rate, respiratory effort, reflex irritability, muscle tone and skin colour. A rating of 0, 1 or 2 is given for each sign, reflecting whether it was absent, present but not adequate, or normal.^{2,3} These result in a total score that ranges from 0 to 10.

Apgar scores can also be used as a physiological indicator of how well a newborn is making the transition from *in utero* to the *ex utero* environment.⁴ For example, the skin colour is evaluated to determine the perfusion of the skin with oxygenated blood; heart rate reflects the success in the delivery of oxygen to organs; reflex irritability can be used as a marker for nervous system integration; muscle tone represents the locomotor capacity of the newborn; and respiratory rate represents pulmonary function, which facilitates the neuro-feedback-mediated effort to breathe.⁴

Although Apgar scores are useful to predict neonatal survival,³ as originally intended, there are several factors that could influence Apgar scores. For example, factors such as maternal sedation or anaesthesia, congenital malformations, trauma, gestational age, infections and low birth weight often affect Apgar scores.^{2,5} The health status of newborns may be misclassified because some components of the Apgar score such as skin colour, muscle tone and reflex irritability are partially subjective by nature—that is, the scores are affected by newborn assessment skill (inter-observer variability) and the context of the delivery.^{2,4}

There is controversy regarding the use of the Apgar score alone as evidence about the diagnostic information and/or as a determining factor to initiate resuscitation.² For instance, according to the American Academy of Pediatrics and the American College of Obstetricians and Gynecologists, 'Apgar score alone cannot be considered to be evidence of or a consequence of asphyxia'.² As neonatal resuscitation should be initiated before one minute of birth, the Apgar score is no longer used as a valuable determinant to initiate resuscitation in contemporary clinical practice, but it can be used to evaluate the effectiveness of resuscitation.^{2,4} These limitations may make the interpretation of 'what the Apgar score is measuring' or even 'what low Apgar score is a risk for' unclear.⁵ However, since there is no other better marker that provides a useful clinical shorthand or a mechanism to record fetal-to-neonatal transition, the Apgar score continues to be used in routine clinical practice. Few studies have attempted to link the transition at birth as measured by the Apgar scores and cognitive outcomes in later life.⁶⁻⁹

We undertook this study to examine whether five-minute Apgar scores of 0–5, 6, 7, 8 and 9 (compared with 10) among singleton infants without malformations who were delivered at 37 weeks of gestation or later with vertex presentation was associated with children's educational outcomes at eight years of age.

MATERIALS AND METHODS

Data source

This study utilised data from the South Australian Early Childhood Data Project (SAECDP). The SAECDP is an established project that involves linking government administrative data from state and federal sources. We have merged the perinatal and the National Assessment Program—Literacy and Numeracy (NAPLAN) datasets using de-identified linkage keys generated by an independent data linkage agency (SA-NT DataLink). Ethics approval was provided by the South Australian Department for Health and Ageing (HREC/13/SAH/106/AM08).

Inclusion and exclusion criteria

We included live-born singleton infants without malformations who were delivered at 37 weeks of gestation or later with vertex presentation in South Australia from 1999 to 2008 and have undertaken year 3 NAPLAN. The study excluded infants with birth weight <2500 g or those with no information on five-minute Apgar score. Infants born before 37 weeks' completed gestation or those with birth weight <2500 g were excluded from analysis because these are known conditions that affect both Apgar scores and children's educational outcomes.

Exposure

The South Australian Supplementary Birth Record captures the total Apgar scores measured at one and five minutes after birth. Scores for individual Apgar components were not available. The exposed group comprised infants with a range of five-minute Apgar scores (i.e., 0–5, 6, 7, 8 and 9) while the unexposed group comprised infants with Apgar score of 10.

Outcome

In Australia, children's educational outcomes are assessed by standardised NAPLAN tests at years 3, 5, 7 and 9. The NAPLAN test scores at year 3 (~eight years of age) were used in this study to maximise the eligible study population in South Australia.

NAPLAN has five test domains including reading, writing, spelling, grammar and numeracy. For each domain, NAPLAN results are reported as both direct scores (range from zero to 1000) and grouped into proficiency bands (range from one to six, for year 3). The proficiency bands are used to identify children's educational outcomes relative to the National Minimum Standard (NMS), which is band 2 for year 3. Consistent with national reporting of NAPLAN scores, we report results where NAPLAN was categorised into children performing \leq NMS (\leq band 2) and >NMS (bands 3–6, for year 3). However, our data did not include NAPLAN scores from private schools which accounted for about 34% of the population of year 3 students in South Australia from 2008 to 2015.¹⁰

Confounding

We identified potential confounding based on *a priori* subject matter and expert knowledge, and through the use of Directed Acyclic Graphs (Fig. S1). The potential confounders included method of delivery (normal spontaneous, forceps, lower segment caesarean section (elective), lower segment caesarean section (emergency)), maternal smoking during second half of pregnancy (yes/no), gestational diabetes (yes/no), pregnancy hypertension (yes/no), intrauterine growth restriction (yes/no), gestational age (in weeks), birth weight for gestational age z-score, mother's age at birth (in years), index of relative socioeconomic disadvantage, types of antenatal care (hospital-based, private obstetrical, general practitioner, other), number of antenatal visits (\leq 7, 8–12, \geq 13), maternal occupation, maternal ethnicity (Caucasian, Aboriginal or Torres Strait Islander, others), Australian Remoteness Index for Areas (city, inner regional, outer regional, remote and very remote), mother had a partner (yes/no), maternal analgesia use for labour (yes/no), maternal anaesthesia use for delivery (yes/no), and infant gender (male/female). Birthweight for gestational age *z*-scores were calculated based on Australian national birthweight percentiles by gender and gestational age.¹¹

Statistical analysis

In order to examine the association between five-minute Apgar scores and children's educational outcomes, we calculated multivalued¹² average treatment effects or risk differences of Apgar scores of 0–5, 6, 7, 8 and 9 (compared with 10) on whether a child met the NMS on the NAPLAN tests using augmented inverse probability weighted (AIPW) estimator.¹³ The AIPW estimator is also called double-robust estimator.^{14,15} The AIPW estimator was chosen over traditional regression because it delivers a marginal rather than a conditional estimate, and it fits regression models for both the exposure (Apgar) and outcome (NAPLAN) providing a doubly robust property if either the regression model for the outcome or the exposure is correctly specified.¹⁴

As part of our *a priori* statistical analysis plan, we undertook a negative control outcome¹⁶ analysis to examine the potential for unmeasured confounding. We used 'family possession of school card' as a negative control outcome—where no association or an association of much smaller magnitude would be expected for the association between five-minute Apgar score and the negative control outcome. The estimate of the associations between Apgar scores and children's educational outcomes, and Apgar scores and possession of school card were compared. If the associations between Apgar scores and children's educational outcomes were more likely to be due to residual confounding.

Furthermore, as Apgar score is also captured at one minute in the perinatal database, the above analyses were repeated to estimate the association between one-minute Apgar scores and children's educational outcomes at eight years of age for the purpose of completeness. It has been suggested that the five-minute Apgar is a better predictor of neonatal survival than the one-minute Apgar score.¹⁷

As missing information is minimal (<3%) for variables used in our analyses, we performed complete case analyses. All analyses were conducted using STATA/SE version 15.1 (StataCorp, College Station, TX, USA).

RESULTS

Figure 1 shows the flow of the participants within this study, starting at singleton live births born with vertex presentations in

South Australia from 1999 to 2008, and the proportion successfully linked to school assessments. The numbers differ by each NAPLAN domain because some children did not sit all tests (eg, if they were absent on a particular day due to illness). The timing of the assessments also affects the linkage to perinatal data. For instance, only children in the perinatal database who attended grade 3 during the period from 2008 to 2015 (when NAPLANs were collected) were able to be linked.

Perinatal and sociodemographic characteristics of the mothers and infants according to five-minute Apgar scores are shown in Table 1. Apgar scores <7 at five minutes were more common in infants of mothers who delivered by emergency caesarean section or ventouse, delivered at 37 weeks and ≥41 weeks gestation, smoked during pregnancy, who are Aboriginal or Torres Strait Islander or living in more disadvantaged areas, and who were given analgesia or anaesthesia during labour and delivery.

Table 2 shows the comparison of the proportion of children's educational outcomes at/below the NMS according to a range of five-minute Apgar scores. For example, for Apgar scores of \leq 5 vs 10, the proportion of children performing at/below the NMS for each NAPLAN domain were compared as follows: reading (23.62%) and (20.01%), writing (13.91%) and (11.95%), spelling (24.81%) and (20.37%), grammar (21.85%) and (19.30%), and numeracy (27.24%) and (22.29%). We have also reported the results based on Apgar scores of <7 vs Apgar scores of 7–10 for children performing \leq NMS for each NAPLAN domain in Table S1.

Table 3 shows the association across a range of five-minute Apgar score values and children's educational outcomes at eight years of age. For example, infants with Apgar scores of \leq 5 had at/below NMS reading (risk difference (RD) = 0.07 (-0.16 to 0.29)), writing (RD = 0.27 (-0.14 to 0.68)), spelling (RD = 0.15 (-0.10 to 0.40)), grammar (RD = 0.04 (-0.21 to 0.29)) and numeracy (RD = 0.21 (-0.04 to 0.45)) scores, compared with Apgar score of 10. The negative control outcome analysis results of the association between a range of five-minute Apgar score values and children's possession of school card can be accessed in Table S2.

The pattern of results is also the same when all analyses were repeated using one-minute Apgar score (Tables S3–S5).

DISCUSSION

Main findings

Our study examined the association between Apgar scores and children's educational outcomes at around age eight years. The primary adjusted analysis indicated that Apgar scores of \leq 5 and Apgar score of 6, compared with Apgar score of 10, were more likely to be associated with scoring \leq NMS on each domain of NAPLAN assessments. The sensitivity analysis performed using the negative control outcome to examine the extent to which the association between a range of Apgar score values and children's



FIGURE 1 The flowchart of the study population. NAPLAN, National Assessment Program—Literacy and Numeracy.

educational outcomes is likely generated by confounding helped in the interpretation of our findings. For example, the point estimate for Apgar scores of \leq 5 vs 10 and the negative control outcome (family possession of school card) was much smaller in magnitude than the estimate for the five-minute Apgar scores of \leq 5 vs 10 and children's educational outcomes. Thus, a possible interpretation of our results is that residual confounding does not explain away the association between Apgar scores of \leq 5 vs 10 and children's educational outcomes.

Comparison with previous findings

The majority of the available evidence regarding the association between Apgar scores and IQ comes from studies conducted by linking birth data with routine IQ test results at conscription around age 18.^{68,9} One of these studies conducted by Seidman *et al.*,⁶ showed no association between low Apgar scores and IQ at 18 years while another study by Ehrenstein *et al.*⁸ showed modest association between five-minute Apgar scores <7 and low cognitive function (prevalence ratio, 1.33; 95% CI 0.94–1.88). Consistent with our study, all of the previous studies involved linking the perinatal data with school results or IQ tests in later life. Furthermore, almost all previous studies examined the

association between five-minute Apgar scores dichotomised at <7 and cognition or school outcomes later in life.^{6,8,9,18} However, in our study, we examined the associations between the five-minute Apgar scores of \leq 5, 6, 7, 8 and 9 (compared with 10) and children's educational outcomes.

In the present study, both the crude comparison of the percentage of children performing at/below the NMS and the AIPW analysis together with the negative control outcome analysis suggest that, compared with Apgar score of 10, infants with Apgar scores of \leq 5 and Apgar score of 6 were associated with scoring at or below the NMS on each NAPLAN assessment. The associations were more consistently evident for writing and numeracy domains. These findings may warrant the need for more immediate intensive follow-up of infants born with Apgar scores below 6, and the need for educational support at school age.

Strengths and limitations of this study

The strength of the present study included the use of whole-ofpopulation perinatal data and a standardised NAPLAN test data. Moreover, we have improved the scientific rigour of this study by using AIPW estimator and negative control outcome analyses. We have also examined the associations between one-minute **TABLE 1**Maternal, sociodemographic, and birth characteristicsaccording to five-minute Apgar scores: term singleton live birthsin South Australia, 1999–2008 (N = 65 202†)

	Apgar scores at five minutes			
Characteristics	<7	7–10		
Method of delivery				
Normal spontaneous	348 (53.5%)	41 243 (63.9%)		
Forceps	40 (6.2%)	3192 (4.9%)		
LSCS (elective)	33 (5.1%)	7018 (10.9%)		
LSCS (emergency)	127 (19.5%)	8628 (13.4%)		
Ventouse	102 (15.7%)	4471 (6.9%)		
Smoked during second half of preg	gnancy (<i>n</i> = 63 54	41)		
No	479 (76.2%)	48 605 (77.3%)		
Yes	150 (23.8%)	14 307 (22.7%)		
Maternal asthma				
No	616 (94.8%)	60 075 (93.1%)		
Yes	34 (5.2%)	4477 (6.9%)		
Gestational diabetes				
No	621 (95.5%)	62 219 (96.4%)		
Yes	29 (4.5%)	2333 (3.6%)		
Pregnancy hypertension (all types)				
No	581 (89.4%)	59 845 (92.7%)		
Yes	69 (10.6%)	4707 (7.3%)		
Suspected intrauterine growth rest	triction			
No	642 (98.8%)	63 327 (98.1%)		
Yes	8 (1.2%)	1225 (1.9%)		
BWGA z-score, mean (SD)	0.0 (1.1)	0.0 (1.0)		
Gestational age (in weeks)				
37	40 (6.2%)	3410 (5.3%)		
38	99 (15.2%)	11 746 (18.2%)		
39	115 (17.7%)	13 912 (21.6%)		
40	259 (39.8%)	25 726 (39.9%)		
41+	137 (21.1%)	9758 (15.1%)		
Types of antenatal care (<i>n</i> = 65 059	9)			
Hospital-based	311 (48.0%)	28 101 (43.6%)		
Private obstetrician	107 (16.5%)	16 367 (25.4%)		
General practitioner	164 (25.3%)	14 388 (22.3%)		
Other (eg, none, home birth, midwife)	66 (10.2%)	5555 (8.6%)		
Number of antenatal visits				
≤7	64 (9.8%)	6222 (9.6%)		
8–12	392 (60.3%)	40 572 (62.9%)		
≥13	194 (29.8%)	17 758 (27.5%)		
Maternal ethnicity (<i>n</i> = 65 201)				
Caucasian	598 (92.0%)	59 626 (92.4%)		
Aboriginal or Torres Strait Islander	22 (3.4%)	1736 (2.7%)		
Others	30 (4.6%)	3189 (4.9%)		

TABLE 1 (Continued)

	Apgar scores at five minutes		
Characteristics	<7	7–10	
Maternal occupation (63 296)			
Managers	43 (6.8%)	4225 (6.7%)	
Professionals	51 (8.1%)	5726 (9.1%)	
Para-professionals	29 (4.6%)	3013 (4.8%)	
Tradespersons	21 (3.3%)	2243 (3.6%)	
Clerks	80 (12.7%)	8154 (13.0%)	
Salespersons	121 (19.2%)	10 135 (16.2%)	
Machine operators and labourers	52 (8.3%)	3287 (5.2%)	
Students	24 (3.8%)	2048 (3.3%)	
Home duties	159 (25.2%)	19 404 (31.0%)	
Unemployed, pensioners, other	50 (7.9%)	4431 (7.1%)	
Australian Remoteness Index for A	reas (<i>n</i> = 64 194)	1	
City	430 (66.2%)	44 083 (68.3%)	
Inner regional	59 (9.1%)	6319 (9.8%)	
Outer regional	122 (18.8%)	10 899 (16.9%)	
Remote and very remote	39 (6.0%)	3243 (5.0%)	
Mother had partner (<i>n</i> = 65 186)			
No	119 (18.3%)	10 000 (15.5%)	
Yes	531 (81.7%)	54 536 (84.5%)	
Maternal age at birth (years), mean (SD)	27.9 (5.9)	28.6 (5.7)	
IRSAD (<i>n</i> = 65 120)			
Most disadvantaged	125 (19.3%)	11 537 (17.9%)	
2	78 (12.0%)	6067 (9.4%)	
3	71 (10.9%)	6483 (10.1%)	
4	100 (15.4%)	8158 (12.7%)	
5	60 (9.2%)	6440 (10.0%)	
6	39 (6.0%)	4617 (7.2%)	
7	48 (7.4%)	5695 (8.8%)	
8	55 (8.5%)	6329 (9.8%)	
9	47 (7.2%)	5551 (8.6%)	
Most advantaged	26 (4.0%)	3594 (5.6%)	
Maternal analgesia use for labour			
No	130 (20.0%)	18 992 (29.4%)	
Yes	520 (80.0%)	45 560 (70.6%)	
Maternal anaesthesia use for delive	ery		
No	275 (42.3%)	29 260 (45.3%)	
Yes	375 (57.7%)	35 292 (54.7%)	
Infant gender ($n = 65\ 201$)			
Male	383 (58.9%)	33 015 (51.1%)	
Female	267 (41.1%)	31 536 (48.9%)	

BWGA, birthweight for gestational age; IRSD, Index of Relative Socioeconomic Disadvantage; LSCS, lower segment caesarean section. †The total size of population included in this table is based on at least one National Assessment Program—Literacy and Numeracy domain recorded in the database.

5

(Continues)

TABLE 2 Proportion of children performing at or below the NMS according to a range of five-minute Apgar score values (i.e., 0–5, 6, 7, 8, 9 and 10)

Domain	N	Apgar ≤ 5 ≤NMS/total (%)	Apgar 6 ≤ NMS/ total (%)	Apgar 7 ≤ NMS/ total (%)	Apgar 8 ≤ NMS/ total (%)	Apgar 9 ≤ NMS/ total (%)	Apgar 10 ≤ NMS/ total (%)
Reading	60 722	64/271 (23.62%)	74/333 (22.22%)	179/952 (18.80%)	721/3365 (21.43%)	7823/39 780 (19.67%)	3206/16 021 (20.01%)
Writing	60 514	37/266 (13.91%)	47/327 (14.37%)	121/945 (12.80%)	460/3350 (13.73%)	4880/39 678 (12.30%)	1905/15 948 (11.95%)
Spelling	60 830	67/270 (24.81%)	82/331 (24.77%)	206/952 (21.64%)	762/3375 (22.58%)	8199/39 880 (20.56%)	3264/16 022 (20.37%)
Grammar	60 830	59/270 (21.85%)	76/331 (22.96%)	184/952 (19.33%)	702/3375 (20.80%)	7673/39 880 (19.24%)	3093/16 022 (19.30%)
Numeracy	60 581	73/268 (27.24%)	86/333 (25.83%)	205/941 (21.79%)	771/3358 (22.96%)	8811/39 686 (22.20%)	3566/15 995 (22.29%)

NMS, National Minimum Standard.

TABLE 3 Risk differences comparing 5-minute Apgar scores of 0–5, 6, 7, 8 and 9 (compared with 10) for children performing at or below the NMS for each National Assessment Program—Literacy and Numeracy domain

Five-minute Apgar scores	Reading n = 60 722 RD [†] (95% CI)	Writing n = 60 514 RD (95% CI)	Spelling n = 60 830 RD (95% CI)	Grammar <i>n</i> = 60 830 RD (95% Cl)	Numeracy <i>n</i> = 60 581 RD (95% Cl)
0–5 vs 10	0.07 (-0.16, 0.29)	0.27 (-0.14, 0.68)	0.15 (-0.10, 0.40)	0.04 (-0.21, 0.29)	0.21 (-0.04, 0.45)
6 vs 10	0.16 (-0.07, 0.39)	0.20 (-0.13, 0.52)	0.18 (-0.05, 0.40)	0.14 (-0.08, 0.35)	0.12 (-0.10, 0.34)
7 vs 10	-0.05 (-0.19, 0.08)	0.03 (-0.15, 0.22)	0.02 (-0.11, 0.16)	0.01 (-0.13, 0.15)	0.00 (-0.13, 0.13)
8 vs 10	0.04 (-0.03, 0.12)	0.10 (0.00, 0.21)	0.09 (0.02, 0.17)	0.05 (-0.03, 0.13)	0.03 (-0.04, 0.10)
9 vs 10	0.00 (-0.04, 0.03)	0.03 (-0.02, 0.08)	0.02 (-0.02, 0.05)	0.01 (-0.03, 0.04)	0.01 (-0.02, 0.05)
10	(ref.)	(ref.)	(ref.)	(ref.)	(ref.)

CI, confidence interval; NMS, National Minimum Standard; RD, risk difference.

[†]RD calculated from augmented inverse probability weighted analysis (adjusted for a number of confounders listed in Methods section) showing the proportion of children performing < NMS on each assessment for children with five-minute Apgar scores.

Apgar scores and children's school outcome by conducting similar analyses.

The study has some potential limitations. First, misclassification of the condition of newborns as measured by Apgar scores could result from the partially subjective nature of Apgar scores—that is, due to inter-observer variability¹⁹ and the context of the delivery.⁵ However, the misclassification of the Apgar scores would occur independent of the child's NAPLAN outcome. Second, although Apgar scores have been used worldwide to describe the status of newborns after birth, both obstetrics and paediatrics professional associations² emphasised the limitations of use of Apgar scores to predict long-term outcomes, including neurologic outcome. Nonetheless, there is no better summary measure of fetal or neonatal condition, which accurately predicts long-term outcomes that has been developed and widely implemented.^{5,20,21} Third, scores for individual Apgar components were not available.

Finally, our data only included NAPLAN scores from public schools. However, the majority (more than two-thirds) of year 3 students were enrolled in public schools in South Australia from 2008 to 2015.¹⁰ So in this case, our estimate would be biased if the associations between Apgar score and NAPLAN were vastly different among private school attenders and led to a different estimate if the total target population had been available. However, we cannot see any reason for this to be the case.

In conclusion, we found that, compared with Apgar score of 10, the five-minute Apgar scores of \leq 5 and 6 have negative associations with scoring \leq NMS across all NAPLAN domains.

ACKNOWLEDGEMENTS

We thank the data custodians, the Australian Government Department of Education and Training, the South Australian Government Department for Health and Ageing and the Department for Education and Child Development for providing the de-identified datasets for analysis. We thank SA-NT DataLink for the data linkage. The views expressed in this paper do not necessarily reflect the views or policies of the data custodians. The first author (EY) is fully supported by an Australian Government Research Training Program Scholarship (RTPS). BWM is supported by a National Health and Medical Research Council (NHMRC) Investigator Grant (GNT1176437) and also reports consultancy for ObsEva, Merck KGaA, iGenomix and Guerbet. JWL is supported by an NHMRC Centre of Research Excellence (GNT1099422).

REFERENCES

 Apgar V. A proposal for a new method of evaluation of the newborn infant. *Curr Res Anesth Analg* 1953; **32**(4): 260–267.

- American Academy of Paediatrics Committee on Fetus and Newborn; American College of Obstetricians and Gynecologists Committee on Obstetric Practice. The Apgar score. *Pediatrics* 2015; **136**: 819–822.
- 3. Apgar V, Holaday DA, James LS *et al*. Evaluation of the newborn infant; second report. *J Am Med Assoc* 1958; **168**: 1985–1988.
- 4. Torday JS, Nielsen HC. The molecular Apgar score: a key to unlocking evolutionary principles. *Front Pediatr* 2017; **5**: 45.
- Tiemeier H, McCormick MC. The Apgar paradox. Eur J Epidemiol 2019; 34(2): 103–104.
- Seidman DS, Paz I, Laor A *et al*. Apgar scores and cognitive performance at 17 years of age. *Obstet Gynecol* 1991; **77**: 875–878.
- Stuart A, Otterblad Olausson P, Kallen K. Apgar scores at 5 minutes after birth in relation to school performance at 16 years of age. *Obstet Gynecol* 2011; **118**(2 Pt 1): 201–208.
- Ehrenstein V, Pedersen L, Grijota M *et al.* Association of Apgar score at five minutes with long-term neurologic disability and cognitive function in a prevalence study of Danish conscripts. *BMC Pregnancy Childbirth* 2009; **9**: 14.
- Odd DE, Rasmussen F, Gunnell D *et al*. A cohort study of low Apgar scores and cognitive outcomes. *Arch Dis Child Fetal Neonatal Ed* 2008; **93**(2): F115–F120.
- Australian Bureau of Statistics. Schools, Australia, 2018, Table 42b Number of Full-time and Part-time Students, 2006–2018', data cube: Excel spreadsheet, cat. no. 4221.0; 2018.
- Dobbins TA, Sullivan EA, Roberts CL, Simpson JM. Australian national birthweight percentiles by sex and gestational age, 1998– 2007. *Med J Aust* 2012; **197**(5): 291–294.
- 12. StataCorp. *Stata: Release 15. Statistical Software*. College Station, TX: StataCorp LLC; 2017.
- Glynn AN, Quinn KM. An introduction to the augmented inverse propensity weighted estimator. *Political Analysis* 2017; 18(1): 36–56.
- 14. Funk MJ, Westreich D, Wiesen C *et al*. Doubly robust estimation of causal effects. *Am J Epidemiol* 2011; **173**: 761–767.
- 15. Emsley R, Lunt M, Pickles A, Dunn G. Implementing double-robust estimators of causal effects. *Stata J* 2018; **8**: 334–353.
- Lipsitch M, Tchetgen Tchetgen E, Cohen T. Negative controls: a tool for detecting confounding and bias in observational studies. *Epidemiology* 2010; 21: 383–388.
- Schmidt B, Kirpalani H, Rosenbaum P, Cadman D. Strengths and limitations of the Apgar score: a critical appraisal. *J Clin Epidemiol* 1988; **41**: 843–850.
- Tweed EJ, Mackay DF, Nelson SM et al. Five-minute Apgar score and educational outcomes: retrospective cohort study

of 751,369 children. Arch Dis Child Fetal Neonatal Ed 2016; **101**(2): F121-F126.

- 19. O'Donnell CP, Kamlin CO, Davis PG *et al*. Interobserver variability of the 5-minute Apgar score. *J Pediatr* 2006; **149**: 486–489.
- Ward PM. Validation for Dr Apgar's score. Arch Dis Child 2019; 104(1): 1–2.
- 21. Papile LA. The Apgar score in the 21st century. *N Engl J Med* 2001; **344**: 519–520.

SUPPORTING INFORMATION

Additional supporting information may be found online in the Supporting Information section at the end of the article.

Figure S1. Proposed Directed Acyclic Graphs (DAGs) describing the associations between Apgar scores and children's educational outcomes.

Table S1. Proportion of children performing at or below the National Minimum Standards (NMS) according to five-minute Apgar scores and risk differences comparing Apgar scores of <7 with scores of 7–10 for children performing at or below the NMS for each National Assessment Program—Literacy and Numeracy (NAPLAN) domain.

Table S2. Risk differences comparing a range of five-minute Apgar scores (ie, 0–5, 6, 7, 8 and 9) (compared with 10) for children having school card according to National Assessment Program—Literacy and Numeracy (NAPLAN) domains.

Table S3. Proportion of children performing at or below the National Minimum Standards (NMS) according to one-minute Apgar scores and risk differences comparing Apgar scores of <7 with scores of 7–10 for children performing at or below NMS for each domain.

Table S4. Proportion of children having school card according to one-minute Apgar score and risk differences comparing Apgar scores of <7 with scores of 7–10 for children having school card.

Table S5. Risk differences comparing a range of one-minute Apgar score values (ie 0–5, 6, 7, 8 and 9) (compared with 10) for children performing at or below the National Minimum Standards (NMS) for each domain.

Chapter 8: Elective labour induction versus expectant management in women and children's educational outcomes at eight years of age

8.1 Preface

This chapter contains the final of a series of five studies contributing to this thesis. This study was published in *Ultrasound in Obstetrics and Gynecology* in 2020.

In Chapters 5, 6 and 7, the association between obstetrical interventions during childbirth (caesarean section and Apgar score) and short-term and long-term outcomes, including neonatal mortality, breastfeeding and children's educational outcomes were examined. These studies revealed that obstetrical interventions during childbirth have an influence on neonatal mortality, breastfeeding and children's educational outcomes at later age. In this chapter, the focus is to estimate whether elective induction of labour at 39 weeks of gestation (the most common intervention during labour) as compared with expectant management affects children's educational outcomes at eight years of age.

As no previous study investigated the longer-term neurodevelopmental effect of elective induction of labour at 39 weeks, this chapter (Study 5) provides the first contribution to the ongoing scientific discourse on the effect of elective induction of labour at 39 weeks and longer-term neurodevelopmental outcomes in children.

8.2 Publication

Yisma E, Mol BW, Lynch JW, Murthy NM, Smithers LG. Labour induction versus expectant management in women and children's educational outcomes at eight years of age. *Ultrasound Obstet Gynecol* 2020. <u>https://doi.org/10.1002/uog.23141</u>

8.2.1 Statement of authorship

Title of Paper	Elective labour induction versus expectant management in women and children's educational outcomes at eight years of age		
Publication Status	Published Ccepted for Publication		
	Submitted for Publication	Unpublished and Unsubmitted w ork w ritten in manuscript style	
Publication Details	Yisma E, Mol BW, Lynch JW, Murthy NM, Smithers LG. Elective labour induction vs expectant management in women and children's educational outcomes at 8 years of age. <i>Ultrasound Obstet Gynecol</i> 2020. https://doi.org/10.1002/uog.23141		

Principal Author

Name of Principal Author (Candidate)	Engida Yisma		
Contribution to the Paper	Contributed to the design of the study, performed statistical analysis, interpreted the data, and drafted the manuscript.		
Overall percentage (%)	80%		
Certification:	This paper reports on original research I conducted during the period of my Higher Degree by Research candidature and is not subject to any obligations or contractual agreements with a third party that would constrain its inclusion in this thesis. I am the primary author of this paper.		
Signature	Date 4-Aug-2020		

Co-Author Contributions

By signing the Statement of Authorship, each author certifies that:

- i. the candidate's stated contribution to the publication is accurate (as detailed above);
- ii. permission is granted for the candidate to include the publication in the thesis; and
- iii. the sum of all co-author contributions is equal to 100% less the candidate's stated contribution.

Name of Co-Author	Ben W Mol	
Contribution to the Paper	Contributed to the design of the study, interpreted the findings, and reviewe the manuscript.	d
Signature	Date 04/08/2020	

Name of Co-Author	John W Lynch
Contribution to the Paper	Contributed to the design of the study, interpreted the findings, and reviewed the manuscript.
Signature	Date 6/8/20

Name of Co-Author	Murthy N Mittinty		
Contribution to the Paper	Contributed to the data analysis, interpretation	ion and	l revision of the manuscript.
Signature	Da	Date	06/08/2020

Name of Co-Author	Lisa G Smithers		
Contribution to the Paper	Contributed to the design of the study, interpreted the findings, and reviewed the manuscript.		
Signature		Date	5-Aug-2020

Elective labor induction *vs* expectant management in women and children's educational outcomes at 8 years of age

E. YISMA^{1, 2, 3}, B. W. MOL^{3, 4}, J. W. LYNCH^{1, 3, 5}, M. N. MITTINTY ^{1, 3}, L. G. SMITHERS^{1, 3}

¹School of Public Health, The University of Adelaide, Adelaide, Australia

²School of Allied Health Sciences, College of Health Sciences, Addis Ababa University, Addis Ababa, Ethiopia

³Robinson Research Institute, School of Medicine, The University of Adelaide, Adelaide, Australia

⁴Department of Obstetrics and Gynaecology, Monash University, Melbourne, Australia

⁵Population Health Sciences, Bristol Medical School, University of Bristol, UK

Correspondence to: Engida YISMA

School of Public Health, The University of Adelaide, Adelaide, Australia e-mail: <u>engida.derbie@adelaide.edu.au</u> Mail Drop DX 650 550 SA 5005 Australia

Short title: Elective labour induction and children's educational outcomes

This article has been accepted for publication and undergone full peer review but has not been through the copyediting, typesetting, pagination and proofreading process which may lead to differences between this version and the Version of Record. Please cite this article as doi: 10.1002/uog.23141

This article is protected by copyright. All rights reserved.

Keywords: Elective induction of labour, expectant management, school assessment, cognition, epidemiology

Contribution

What are the novel findings of this work?

Using innovative data linkage of the whole-of-population administrative perinatal data and the National Assessment Program—Literacy and Numeracy data from South Australia, we found that elective induction of labour at 39 weeks of gestation, compared with expectant management, did not affect children's educational outcomes at eight years of age.

What are the clinical implications of this work?

The findings of this study provide the first presentation of the effect of elective induction of labour at 39 weeks of gestation as compared with expectant management on children's educational outcomes.

Objective

To estimate the effect of elective induction of labour at 39 weeks of gestation on children's educational outcomes as measured by the Australian National Assessment Program— Literacy and Numeracy (NAPLAN) tests at year 3 (~8 years of age), compared with expectant management.

Methods

We merged perinatal data, including information regarding all infants in South Australia from 1999 to 2008, with children's school assessment data (i.e., NAPLAN data). The study population included all singleton births born without malformations at 39-42 weeks of gestation in vertex presentation. Children had to have undertaken year-3 NAPLAN (~8 years of age). We excluded births from women who had a contraindication to vaginal delivery and with conditions possibly justifying elective delivery before 39 weeks of gestation. Our outcome of interest was children's educational outcomes as measured by NAPLAN. The NAPLAN included five learning domains (reading, writing, spelling, grammar and numeracy). Each domain was categorised according to performing at or below versus above the National Minimum Standards (NMS). Average Treatment Effects (ATEs) of elective induction of labour at 39 weeks for children performing at/below the NMS for each domain were estimated using Augmented Inverse Probability Weighted (AIPW) estimator accounting for potential confounders.

Results

Among 53,843 children born at 39-42 weeks with vertex presentation from 1999-2008 and those who were expected to participate in year 3 NAPLAN from 2008-2015, a total of 31,120 children had at least one year 3 NAPLAN domain. Of these (i.e., 31,120 children), 1,353 children were delivered after elective induction of labour at 39 weeks while 29,767 children were born following expectant management. The ATEs (mean differences) comparing

elective induction of labour at 39 weeks with expectant management for children scoring at/below the NMS on each domain were: reading (0.01 (95% CI -0.02 to 0.03)), writing (0.02 (95% CI 0.00 to 0.04)), spelling (0.01 (95% CI -0.01 to 0.04)), grammar (0.02 (95% CI -0.01 to 0.04)) and numeracy (0.03 (95% CI -0.00 to 0.05)).

Conclusion

Elective induction of labour at 39 weeks of gestation did not affect children's standardised literacy and numeracy testing outcomes at eight years of age when compared with expectant management.

Introduction

It has been recommended that non-medically indicated deliveries such as deliveries following elective induction of labour should be avoided for women who are below 39 weeks of gestation because they are associated with adverse short- and long-term outcomes.¹ Conversely, due to the increased risk of perinatal morbidity and mortality associated with late-term and post-term pregnancies, it has been suggested that deliveries following non-medically indicated induction of labour should be considered from 41 weeks of gestation.^{2,3}

However, elective induction of labour has been considered as unsafe when the gestational age is between 39 and 40 weeks because of the belief that it is associated with increased risk of caesarean delivery and adverse perinatal outcomes than spontaneous labour. This assumption was made based on observational studies⁴⁻⁹ in which women who were induced were compared with women in spontaneous labour at the same gestational age. It is known that the comparison of induction of labour with spontaneous labour at the same gestational age is inappropriate because spontaneous labour is not a definite alternative to induction of labour.¹⁰ In clinical practice, women and obstetric care providers are required to choose either induction of labour or expectant management, which comprises the onset of spontaneous labour, medically-indicated induction or caesarean at a later date.

Several studies¹¹⁻¹⁷ have found that elective induction of labour at 39 weeks reduces adverse perinatal outcomes, including caesarean delivery. However, there has been limited research reported on the link between labour induction at 39 weeks and children's educational outcomes.¹⁸ We speculate on two potential mechanisms linking elective labour induction and children's educational outcomes. First, the risk of special educational need, compared to children born at 40 weeks, increased among children born from 37–39 weeks gestation.¹⁹ Second, it is thought that labour induction increases stress and the presence of adverse prenatal and perinatal stress may result in epigenetic modification of gene expression, which may permanently change neuroendocrine and behavioural responses in children's later life.²⁰

We undertook this study to examine whether elective induction of labour at 39 weeks, compared with expectant management, had an effect on children's educational outcomes at eight years.

This article is protected by copyright. All rights reserved.

Materials and Methods

Data source

This study utilised data from the South Australian Early Childhood Data Project (SAECDP). The SAECDP is an established project that involves linking government administrative data from state and federal sources. We have merged the perinatal data and the National Assessment Program—Literacy and Numeracy (NAPLAN) datasets using de-identified linkage keys generated by an independent data linkage agency (SA-NT DataLink). The ethics approval was provided by the South Australian Department for Health and Ageing (HREC/13/SAH/106/AM08).

Eligibility criteria

Of the total 285,063 children in South Australia who were born from 1999 to 2013, we included all births at 39-42 weeks of gestation in vertex presentation, which were live singletons without malformations from 1999-2008 and children who were expected to participate in year 3 NAPLAN from 2008-2015 (N= 53,843). The study excluded births from women who had contraindication to vaginal delivery (previous caesarean births), who had elective caesarean delivery, and women who were with conditions possibly justifying elective delivery prior to 39 weeks of gestation²¹—these include pregnancy hypertension, gestational diabetes, suspected foetal-growth restriction, or disorders associated with antepartum haemorrhage such as placenta previa. As the aim of our study was to compare elective induction of labour at 39 weeks with expectant management, births following spontaneous or medically-indicated deliveries at 39 weeks were excluded.

All students in year 3 were expected to participate in NAPLAN testing. However, the SAECDP did not include NAPLAN scores from students attending private schools as these have not been made available for linkage. Private school attendance accounted for about 34% of the population of year 3 students in South Australia from 2008 to 2015.²² Moreover, some students from public school did not take NAPLAN due to exempt, absent and withdrawal. We excluded children with no NAPLAN score due to exempt, absent and withdrawal from

2008-2015. Consequently, a total of 31,120 children who had at least one NAPLAN domain met the inclusion criteria.

Exposure

In South Australia, as the perinatal database included information about onset of labour, reason for induction of labour, and gestational age at birth, the exposure variable was computed from these three variables. The exposed group comprised births following elective induction of labour at 39 weeks of gestation. The control (unexposed) group comprised births following expectant management—that is, deliveries following spontaneous labour and elective induction of labour or medically-indicated induction or caesarean from 40-42 weeks of gestation. Gestational age (at birth) dating was determined predominately based on early dating ultrasound. However, in the small number of women when dating ultrasound was not available, gestational age was determined from the first day of the last menstrual period and estimates at birth.

Outcome

In South Australia and other jurisdictions, children's educational outcomes are assessed by standardized NAPLAN tests at years 3, 5, 7 and 9. The NAPLAN test scores at year 3 (~8 years of age) were used in this study to maximize the eligible study population in South Australia. The NAPLAN has five test domains including reading, writing, spelling, grammar and numeracy. For each domain, NAPLAN results are reported as both direct scores (range from 0 to 1000) and grouped into proficiency bands (range from 1 to 6, for year 3). The proficiency bands are used to identify children's educational outcomes relative to the National Minimum Standard (NMS), which is band 2 for year 3. Consistent with the national reporting of NAPLAN scores, we report results where NAPLAN was categorised into children performing at or below the NMS (\leq band 2) versus above the NMS (bands 3–6). Moreover, we also categorised the NAPLAN scores into children scoring band 1 versus bands 2–6 for the purpose of comparison.

Confounding

We identified the potential confounding based on *a priori* subject matter and expert knowledge. The potential confounders in this study included maternal smoking during the second half of pregnancy (yes/no), birth weight for gestational age z-score, mother's age at birth (in years), index of relative socioeconomic disadvantage, types of antenatal care (hospital-based, private obstetrical, general practitioner, other), number of antenatal visits (\leq 7, 8-12, \geq 13), maternal occupation, maternal ethnicity (Caucasian, Aboriginal or Torres Strait Islander, others), Australian remoteness index for areas (city, inner regional, outer regional, remote and very remote), mother had a partner (yes/no), and infant sex (male/female). Birthweight for gestational age z-scores were calculated based on the Australian national birthweight percentiles by sex and gestational age.²³

Statistical analysis

In order to estimate the effect of induction of labour at 39 weeks and children's educational outcomes, we calculated the Average Treatment Effects (ATEs) of elective induction of labour at 39 weeks of gestation compared with expectant management on whether a child met the NMS on each NAPLAN domain using Augmented Inverse Probability Weighted (AIPW) estimator.²⁴ The AIPW estimator is also called a doubly-robust estimator.^{25,26} The AIPW estimator was chosen over traditional regression because it delivers a marginal rather than a conditional estimate, and it fits regression models for both the exposure (elective induction of labour) and the outcome (NAPLAN) providing a doubly-robust property if either the regression model for the outcome or the exposure is correctly specified.²⁵

In this study, the AIPW estimator fits a logistic regression model for the exposure conditional on each confounder listed above to estimate propensity scores for each individual. The propensity scores were used to weight the observed data. The AIPW also fits a logistic regression model for the outcome conditional on all confounders for each category of the exposure to obtain predicted outcomes for each individual. After estimating the propensity scores and predicted outcomes, the AIPW estimator combines the estimated values along with observed values to calculate the doubly robust estimates of outcome for each level of exposure for each individual.²⁵ Finally, the mean of the estimates of outcome under exposed (elective induction of labour) and control (expectant management) was calculated. These estimated means were used to calculate ATE, which is a mean difference in outcome among exposed and control groups. As our outcome (children's educational outcomes) was binary, the ATE calculated using the combined doubly robust estimate is equivalent to population proportion of children performing at or below the NMS on each NAPLAN domain.

All analyses were conducted using STATA/SE version 15. (StataCorp, College Station, TX, USA).

Results

Figure 1 shows the flow of the participants within this study, starting at singleton live births with vertex presentations at 39-42 weeks of gestation in South Australia from 1999 to 2008, and the proportion of children merged to school assessments data (NAPLAN) from 2008 to 2015. A total of 31,120 children—1,353 were delivered after elective induction of labour at 39 weeks and 29,767 were born following expectant management—who had at least one NAPLAN domain at eight years of age were merged. The total number of births differ according to each NAPLAN domain because some children did not sit all tests (e.g., if they were absent on a particular day due to illness). The timing of the assessments also affects the linkage to perinatal data. For instance, only children in the perinatal database who attended grade three during the period from 2008 to 2015 (when NAPLANs were collected) were merged and included in this study.

Perinatal and sociodemographic characteristics of the mothers and infants according to elective induction of labour at 39 weeks versus expectant management are shown in **Table 1**. The proportion of spontaneous vaginal delivery was higher among women who were electively induced at 39 weeks than women who were expectantly managed. Elective induction of labour at 39 weeks of gestation was more common in women who did not smoke during pregnancy, who attended antenatal care with private obstetrician, who had 8-12 antenatal visits, and who are Caucasian or living in a city. The proportion of elective induction of labour at 39 weeks was also higher in women who had partner. The average age of women who underwent elective induction of labour at 39 weeks was higher induction of labour at 39 weeks was also higher in women who had partner. The average age of women who underwent elective induction of labour at 39 weeks was higher compared with those who were expectantly managed (**Table 1**).

Table 2 shows the proportion of children's educational outcomes at or below the NMS

 according to elective induction of labour at 39 weeks versus expectant management as well as

 the results from the AIPW estimator analysis. For instance, the proportion of children

 performing at/below the NMS for each NAPLAN according to elective induction of labour at

 39 weeks versus expectant management were compared as follows: reading (17.61%) and

(18.93%), writing (11.61%) and (11.56%), spelling (18.61%) and (19.74%), grammar (18.30%) and (18.58%), and numeracy (21.46%) and (21.23%).

The ATEs (as calculated from the AIPW estimator analysis) comparing elective induction of labour at 39 weeks with expectant management for children scoring at or below the NMS on NAPLAN domain ranged from 0.01 (95% CI -0.02 to 0.03) for reading to 0.03 (95% CI 0.00 to 0.05) for numeracy (**Table 2**).

The pattern of the results is also the same when similar analyses were conducted for children scoring band 1 versus bands 2–6 (**Supplementary Table S1**).

Discussion

In the current study, using the whole-of-population perinatal data including information regarding infants in South Australia born from 1999 to 2008 and year 3 NAPLAN data, we compared births following elective induction of labour at 39 weeks of gestation with expectant management to examine children's educational outcome at around age eight years. Our estimates suggest that elective induction of labour at 39 weeks, compared with expectant management, had little to no effect on performing at or below the NMS for each NAPLAN domain.

Previous systematic reviews and meta-analyses, randomized control trials and observational studies of elective induction of labour at 39 weeks of gestation have exclusively reported on short-term perinatal outcomes, including caesarean delivery.¹¹⁻¹⁷ These previous studies have shown that elective induction of labour at 39 weeks, compared with expectant management, had lower rates of maternal and neonatal morbidity, lower rates of caesarean section, and fewer stillbirths. The underlying biological mechanism that provides the possible explanation for the improved perinatal outcomes following elective induction of labour at 39 weeks may be related to the prevention of the impact of placental insufficiency and excess foetal growth. For example, elective induction of labour at 39 weeks of gestation may lower rates of stillbirth as the placenta is able to perfuse the foetus before and during labour. It was hypothesised that placental insufficiency is the underlying mechanism for unexplained term stillbirth.²⁷ On the other hand, it has been demonstrated that foetal macrosomia (birth weight \geq 4500 g) is associated with prolonged labour, cephalo-pelvic disproportion and shoulder dystocia.²⁸ Elective induction of labour at 39 weeks may reduce the risk of adverse perinatal outcomes by preventing excess foetal growth as the gestational age increases.

Despite the large amount of evidence regarding the short-term outcome of elective induction of labour at 39 weeks, there is limited evidence regarding the longer-term neurodevelopmental effects of induction of labour at 39 weeks of gestation. A 2020 study conducted by Werner *et al*¹⁸ in Rhode Island, USA to evaluate educational outcomes among

This article is protected by copyright. All rights reserved.

d Articl Accepte children born by induction of labour at 39 or 40 weeks versus children born following expectant management found that induction of labour at term is not associated with poorer third-grade reading and math test scores. However, the study did not distinguish whether the induction of labour was elective or not. The current study was, however, the first to examine the effect of elective induction of labour at 39 weeks on children's educational outcomes. We found that elective induction of labour at 39 weeks had little to no effect on standardised NAPLAN tests at around age 8 years. The current study provides the first contribution to the ongoing scientific discourse on the effect of elective induction of labour at 39 weeks of gestation and longer-term neurodevelopmental outcomes.

The strength of the present study included the use of the whole-of-population perinatal data and a standardised NAPLAN test data. Our data comprised information regarding the reason of induction of labour, which helped us identify the elective and medically-indicated induction of labour. Moreover, we have improved the scientific rigour of the current study by using AIPW estimator in order to balance a broad range of sociodemographic and perinatal factors associated with both the exposure (elective induction of labour at 39 weeks) and the outcome (children's educational outcomes).

The study has some limitations. The first limitation could be regarding information on gestational age, which was only available in weeks in our perinatal dataset. There may be errors in the gestational age dating, resulting in misclassification of our exposure. However, the misclassification of the exposure would occur independent of the children's NAPLAN outcome but may bias estimates to the null. The other limitation of this study was that our data only included NAPLAN scores from public schools, which is representative of the student population from public schools in South Australia. Nevertheless, it has been emphasised by a number of authors that representativeness is not a prerequisite for causal inference.^{29,30} Moreover, as the majority (more than two-thirds) of year 3 students were enrolled in public schools in South Australia from 2008 to 2015,²² our estimate regarding the effect of elective induction of labour on children's educational outcomes is unlikely to be biased. The only way our estimate would be biased was if the association between elective

induction of labour and NAPLAN was vastly different among private school attenders, and led to a different estimate if the total target population had been available. However, we cannot see any reason for this to be the case.

In conclusion, elective induction of labour at 39 weeks of gestation compared with expectant management did not affect the proportion of children scoring at or below the NMS on standardised NAPLAN tests at eight years of age.
References

 ACOG Committee Opinion No. 765: Avoidance of Nonmedically Indicated Early-Term Deliveries and Associated Neonatal Morbidities. *Obstet Gynecol* 2019; **133**(2): e156e63.

2. ACOG. Practice bulletin No. 146: Management of late-term and postterm pregnancies. *Obstet Gynecol* 2014; **124**(2 Pt 1): 390-6.

3. World Health Organization. WHO recommendations: induction of labour at or beyond term. Geneva: World Health Organization, 2018.

4. Vrouenraets FP, Roumen FJ, Dehing CJ, van den Akker ES, Aarts MJ, Scheve EJ.
Bishop score and risk of cesarean delivery after induction of labor in nulliparous women. *Obstet Gynecol* 2005; **105**(4): 690-7.

5. Macer JA, Macer CL, Chan LS. Elective induction versus spontaneous labor: a retrospective study of complications and outcome. *Am J Obstet Gynecol* 1992; **166**(6 Pt 1): 1690-6; discussion 6-7.

 Vahratian A, Zhang J, Troendle JF, Sciscione AC, Hoffman MK. Labor progression and risk of cesarean delivery in electively induced nulliparas. *Obstet Gynecol* 2005; **105**(4): 698-704.

7. Maslow AS, Sweeny AL. Elective induction of labor as a risk factor for cesarean delivery among low-risk women at term. *Obstet Gynecol* 2000; **95**(6 Pt 1): 917-22.

8. Levine LD, Hirshberg A, Srinivas SK. Term induction of labor and risk of cesarean delivery by parity. *J Matern Fetal Neonatal Med* 2014; **27**(12): 1232-6.

9. Rattigan MI, Atkinson AL, Baum JD. Delivery route following elective induction of labor at term: analysis of 807 patients. *J Clin Med Res* 2013; **5**(4): 305-8.

10. Little SE, Caughey AB. Induction of Labor and Cesarean: What is the True Relationship? *Clin Obstet Gynecol* 2015; **58**(2): 269-81.

Sotiriadis A, Petousis S, Thilaganathan B, Figueras F, Martins WP, Odibo AO, Dinas K, Hyett J. Maternal and perinatal outcomes after elective induction of labor at 39 weeks in uncomplicated singleton pregnancy: a meta-analysis. *Ultrasound Obstet Gynecol* 2019; 53(1): 26-35.

Grobman WA, Caughey AB. Elective induction of labor at 39 weeks compared with expectant management: a meta-analysis of cohort studies. *Am J Obstet Gynecol* 2019; **221**(4): 304-10.

This article is protected by copyright. All rights reserved.

Grobman WA, Rice MM, Reddy UM, Tita ATN, Silver RM, Mallett G, Hill K, Thom EA, El-Sayed YY, Perez-Delboy A, Rouse DJ, Saade GR, Boggess KA, Chauhan SP, Iams JD, Chien EK, Casey BM, Gibbs RS, Srinivas SK, Swamy GK, Simhan HN, Macones GA. Labor Induction versus Expectant Management in Low-Risk Nulliparous Women. *N Engl J Med* 2018; **379**(6): 513-23.

14. Gibbs Pickens CM, Kramer MR, Howards PP, Badell ML, Caughey AB, Hogue CJ. Term Elective Induction of Labor and Pregnancy Outcomes Among Obese Women and Their Offspring. *Obstet Gynecol* 2018; **131**(1): 12-22.

 Sinkey RG, Blanchard CT, Szychowski JM, Ausbeck E, Subramaniam A, Neely CL, Casey BM, Tita AT. Elective Induction of Labor in the 39th Week of Gestation Compared With Expectant Management of Low-Risk Multiparous Women. *Obstet Gynecol* 2019;
 134(2): 282-7.

16. Sinkey RG, Lacevic J, Reljic T, Hozo I, Gibson KS, Odibo AO, Djulbegovic B, Lockwood CJ. Elective induction of labor at 39 weeks among nulliparous women: The impact on maternal and neonatal risk. *PLoS One* 2018; **13**(4): e0193169.

17. Souter V, Painter I, Sitcov K, Caughey AB. Maternal and newborn outcomes with elective induction of labor at term. *Am J Obstet Gynecol* 2019; **220**(3): 273 e1- e11.

Werner EF, Schlichting LE, Grobman WA, Viner-Brown S, Clark M, Vivier PM.
 Association of Term Labor Induction vs Expectant Management With Child Academic
 Outcomes. *JAMA Netw Open* 2020; **3**(4): e202503.

MacKay DF, Smith GC, Dobbie R, Pell JP. Gestational age at delivery and special educational need: retrospective cohort study of 407,503 schoolchildren. *PLoS Med* 2010; 7(6): e1000289.

20. Tribe RM, Taylor PD, Kelly NM, Rees D, Sandall J, Kennedy HP. Parturition and the perinatal period: can mode of delivery impact on the future health of the neonate? *J Physiol* 2018; **596**(23): 5709-22.

21. The Joint Commission. Specifications manual for Joint Commission National Quality Core Measures. Appendix A Table 11.07: Conditions possibly justifying elective delivery prior to 39 weeks gestation. 2012.

https://manual.jointcommission.org/releases/TJC2013A/AppendixATJC.html (accessed 27 February 2020).

22. Australian Bureau of Statistics. Schools, Australia, 2018, 'Table 42b Number of Fulltime and Part-time Students, 2006-2018', data cube: Excel spreadsheet, cat. no. 4221.0. 2018.

23. Dobbins TA, Sullivan EA, Roberts CL, Simpson JM. Australian national birthweight percentiles by sex and gestational age, 1998-2007. *Med J Aust* 2012; **197**(5): 291-4.

24. Glynn AN, Quinn KM. An Introduction to the Augmented Inverse Propensity Weighted Estimator. *Political Analysis* 2017; **18**(1): 36-56.

25. Funk MJ, Westreich D, Wiesen C, Sturmer T, Brookhart MA, Davidian M. Doubly robust estimation of causal effects. *Am J Epidemiol* 2011; **173**(7): 761-7.

26. Emsley R, Lunt M, Pickles A, Dunn G. Implementing Double-robust Estimators of Causal Effects. *Stata J* 2018; **8**(3): 334-53.

 Pasztor N, Kereszturi A, Kozinszky Z, Pal A. Identification of causes of stillbirth through autopsy and placental examination reports. *Fetal Pediatr Pathol* 2014; **33**(1): 49-54.
 Galal M, Symonds I, Murray H, Petraglia F, Smith R. Postterm pregnancy. *Facts Views Vis Obgyn* 2012; **4**(3): 175-87.

29. Rothman KJ, Gallacher JEJ, Hatch EE. Why representativeness should be avoided. *Int J Epidemiol* 2013; **42**(4): 1012-4.

30. Richiardi L, Pizzi C, Pearce N. Commentary: Representativeness is usually not necessary and often should be avoided. *Int J Epidemiol* 2013; **42**(4): 1018-22.

Legends

Figure 1. The flowchart of the study population. NAPLAN, National Assessment Program— Literacy and Numeracy.

Supplementary Material

Table S1. Proportion of children performing at Band 1 according to elective induction of

 labour at 39 weeks versus expectant management and

 ATEs comparing elective induction of labour at 39 weeks with expectant management for

 children performing at Band 1 for each NAPLAN

 domain

Table 1. Maternal, sociodemographic, and birth characteristics according to elective induction of labour at 39 weeks and expectant management at 40, 41 and 42 weeks: live singleton births in South Australia, 1999-2008 (N=31,120⁺)

		Expectant-
	Elective induction	Management
	of labour	(N= 29767)
	(N=1353)	
Method of delivery		
Normal spontaneous	1052 (77.8%)	21630 (72.7%)
Forceps	60 (4.4%)	1731 (5.8%)
Emergency caesarean section	158 (11.7%)	3970 (13.3%)
Ventouse	83 (6.1%)	2436 (8.2%)
Smoked second half of pregnancy (n= 30,349)		
No	1058 (80.0%)	22683 (78.1%)
Yes	265 (20.0%)	6343 (21.9%)
BWGA z-score, mean (SD)	0.2 (1.0)	0.0 (1.0)
Types of antenatal care (n= 31,068)		
Hospital-based	460 (34.0%)	12035 (40.5%)
Private obstetrician	650 (48.0%)	6592 (22.2%)
General practitioner	208 (15.4%)	7780 (26.2%)
Other (e.g. none, home birth, midwife)	35 (2.6%)	3308 (11.1%)
Number of antenatal visits		
<=7	88 (6.5%)	2170 (7.3%)
8-12	901 (66.6%)	18727 (62.9%)
>=13	364 (26.9%)	8870 (29.8%)
Maternal ethnicity (n= 31,119)		
Caucasian	1286 (95.0%)	27854 (93.6%)
Aboriginal or Torres Strait Islander	33 (2.4%)	651 (2.2%)
Others	34 (2.5%)	1261 (4.2%)
Maternal occupation (n= 30,305)		
Managers	99 (7.5%)	2014 (6.9%)
Professionals	143 (10.9%)	2706 (9.3%)
Para-professionals	76 (5.8%)	1373 (4.7%)
Tradespersons	43 (3.3%)	1057 (3.6%)
Clerks	201 (15.3%)	3899 (13.5%)
Salespersons	189 (14.4%)	5008 (17.3%)
Machine operators and labourers	40 (3.0%)	1591 (5.5%)
Students	19 (1.4%)	1026 (3.5%)
Home duties	448 (34.0%)	8192 (28.3%)
Unemployed, other	59 (4.5%)	2122 (7.3%)

This article is protected by copyright. All rights reserved.

Australian Remoteness Index for Areas		
(n=31,116)		
City	955 (70.6%)	19657 (66.0%)
Inner regional	107 (7.9%)	3166 (10.6%)
Outer regional	232 (17.1%)	5305 (17.8%)
Remote and very remote	59 (4.4%)	1635 (5.5%)
Mother had partner (n= 31,114)		
No	166 (12.3%)	4665 (15.7%)
Yes	1187 (87.7%)	25096 (84.3%)
Maternal age at birth (years), mean (SD)	29.6 (5.5)	28.1 (5.7)
IRSAD (n= 31,078)		
Most disadvantaged	197 (14.6%)	5249 (17.7%)
2	102 (7.5%)	2829 (9.5%)
3	142 (10.5%)	3057 (10.3%)
4	158 (11.7%)	3725 (12.5%)
5	141 (10.4%)	2949 (9.9%)
6	83 (6.1%)	2208 (7.4%)
7	121 (9.0%)	2583 (8.7%)
8	178 (13.2%)	2838 (9.5%)
9	153 (11.3%)	2568 (8.6%)
Most advantaged	76 (5.6%)	1721 (5.8%)
Infant sex		
Male	666 (49.2%)	15135 (50.8%)
Female	687 (50.8%)	14632 (49.2%)
		1 0

Abbreviations: BWGA, birthweight for gestational age; IRSD, Index of Relative Socioeconomic Disadvantage

[†]The total sample size included in this table is based on at least one NAPLAN domain recorded in the database.

Table 2. Proportion of children performing at or below the NMS according to elective induction of labour at 39 weeks versus expectant management and ATEs comparing elective induction of labour at 39 weeks with expectant management for children performing at or below the NMS for each NAPLAN domain

	Total sample	Elective labour induction Expectant management				
Domain	(N)	≤NMS/total (%)	≤NMS/total (%)	\mathbf{ATE}^{\dagger}	95% CI	Р
Reading	29,114	223/1,266 (17.61%)	5271/27,848 (18.93%)	0.01	-0.02, 0.03	0.524
Writing	29,022	140/1,264 (11.61%)	3209/27,758 (11.56%)	0.02	0.00, 0.04	0.101
Spelling	29,141	236/1,268 (18.61%)	5502/27,873 (19.74%)	0.01	-0.01, 0.04	0.308
Grammar	29,141	232/1,268 (18.30%)	5178/27,873 (18.58%)	0.02	-0.01, 0.04	0.178
Numeracy	29,031	273/1,272 (21.46%)	5892/27,759 (21.23%)	0.03	-0.00, 0.05	0.053

Abbreviations: ATE, Average Treatment Effect; CI, confidence interval; NMS, National Minimum Standard.

[†]ATE calculated from augmented inverse probability weighted analysis adjusted for maternal smoking during the second half of pregnancy, birth weight for gestational age z-score, mother's age at birth, index of relative socioeconomic disadvantage, types of antenatal care, number of antenatal visits, maternal occupation, maternal ethnicity, Australian remoteness index for areas, mother had a partner, and infant sex.



Chapter 9: Discussion and conclusions

The discussion of specific results of the five studies was detailed in previous chapters. In this chapter, I summarise the overall research and provide a general discussion. The main objective of the thesis was to examine the effect of obstetrical interventions during labour and birth on maternal and child outcomes, including breastfeeding, neonatal mortality and children's educational outcomes. The uniqueness of the current thesis is that it utilised high-quality data from low-, middle- and high-income countries, and applied advanced epidemiological data analysis methods that are designed to improve causal inference.¹⁹⁰⁻¹⁹⁴ In this final chapter, key findings and contributions of each study are presented, strengths and limitations are discussed, as well as implications and concluding remarks are provided.

9.1 Key findings and contributions

Caesarean section rates and sociodemographic characteristics in Ethiopia

In Study 1, it was found that the caesarean section rate in Ethiopia in 2016 was more than double the rate in 2000. As detailed in Chapter 4, Study 1 also highlighted that there were substantial regional and socioeconomic disparities in caesarean section rates in Ethiopia. For instance, among the administrative regions of Ethiopia, Somali state had the lowest rate and Addis Ababa had the highest in 2016. The rates among urban women were about twelve times higher than those among rural women in 2016. The caesarean section rate among births in the richest quintiles of household wealth was about 14 times more than for births among women in the poorest quintiles of household wealth in 2016. Furthermore, the risk of

caesarean delivery was about 78% higher for women who gave birth in private sector than women who gave birth in public health facilities.

The increase in the national caesarean section rates seen in Ethiopia was partly attributable to an increase in the proportion of deliveries occurring within health facilities. The proportion of health facility deliveries in Ethiopia in 2016 was more than five times the proportion of health facility births in 2000.¹¹⁶ Moreover, urban women, women from the richest quintiles of household wealth and private healthcare facilities were the major contributors to the national increase in caesarean section rates in Ethiopia. The large disparities in the caesarean section rate according to sociodemographic characteristics suggests unequal access to emergency caesarean section, particularly for poor and rural women in Ethiopia. This means that if there are obstetric complications during labour and birth, poorer and rural women and their newborns are at higher risk for adverse birth outcomes. Boerma *et al.*⁵ in their 2018 study defined the population level caesarean section use thresholds of less than 10% as an indicator of poor access (underuse), and reported that access to caesarean section remains a challenge in many low-income countries, particularly for poor women.

Caesarean section and neonatal mortality in Ethiopia

In Study 2, the temporal association between caesarean section and neonatal mortality was examined using both individual-and aggregate-level data from 2000 to 2016 in Ethiopia. Study 2 found that the association between caesarean birth and neonatal death had changed over time. For instance, the adjusted prevalence ratios (aPR) for neonatal death among neonates born via caesarean section versus vaginal birth increased from 0.95 (95% CI: 0.29 to 3.19) to 2.81 (95% CI: 1.11 to 7.13) from 2000 to 2016, respectively. These differences in associations were attributable to changes in patterns of the underlying indications for

caesarean intervention which may be aggravated by contextual factors such as unequal access, infrastructural, structural and health workforce constraints. The changing association between caesarean birth and neonatal death may reflect a maturing of the health system and a secular shift in the characteristics of Ethiopian women undergoing caesarean section after complicated labour or severe foetal compromise. Hence, caesarean birth in Ethiopia may not prevent neonatal deaths because the foetus has already experienced complications that mean neonatal death is difficult to prevent.

There have been a number of studies that reported on the association between caesarean section and neonatal mortality.^{35-44,49-51} However, a review of the literature revealed that these previous studies relied on reporting only the magnitude and direction of the association between caesarean section and neonatal mortality and did not provide an interpretation of the associations by considering contextual factors such as unequal access, infrastructural, structural and health workforce constraints that could play a role in the association between caesarean section and neonatal mortality. As detailed in Chapter 5, Study 2 was guided by the 'Three Delays Model' to facilitate the interpretation of the association between caesarean section and neonatal mortality by considering these contextual factors in Ethiopia. It is clear that in addition to reporting the magnitude of associations, the ability to provide an interpretation of what the associations mean using the 'Three Delays Model' constitutes a major advance in the literature. This is because Study 2 was the first study to examine the temporal association between caesarean birth and neonatal mortality as well as to interpret the association between caesarean birth and neonatal mortality both empirically and theoretically.

Caesarean section and breastfeeding in sub-Saharan Africa

In Study 3, it was found that the within-country aPR for early initiation of breastfeeding and exclusive breastfeeding under 6 months was variable across the 33 countries in sub-Saharan Africa, but not for children ever breastfed. Furthermore, as detailed in Chapter 6, the meta-analyses, which summarised the magnitude of these within-country effect estimates (aPR), showed that caesarean section was associated with a 46% reduction in the prevalence of early initiation of breastfeeding, while there was little difference in exclusive breastfeeding under 6 months, and for children ever breastfed.

Unlike other studies that investigated the relationship between caesarean section and breastfeeding, Study 3 is unique in a number of ways. First, it utilised standard definitions of breastfeeding outcomes recommended by agencies such as the WHO. Second, it utilised high-quality nationally representative data collected using standardised methodology and identical core questionnaire in large number of countries (33 countries) in sub-Saharan Africa over a long time period (2010-2017/2018). Third, it employed similar analytical methodology, in which adjustment for similar potential confounders for each of the 33 countries was made, for three different breastfeeding outcomes (early initiation of breastfeeding, exclusive breastfeeding under 6 months and children ever breastfed). Finally, the study summarised the magnitudes of the 33 within-country effects for the three breastfeeding outcomes in an overall estimate using random-effects meta-analysis. Given these, Study 3 provides high-quality public health evidence regarding the link between caesarean section and breastfeeding indicators from the perspective of LMICs in sub-Saharan Africa.

Apgar scores and children's educational outcomes

144

In Study 4, as each value of the Apgar score may indicate different physiological conditions, it was aimed at examining the associations between one- and five-minute Apgar scores of 0-5, 6, 7, 8 and 9 (compared with 10) and children's educational outcomes at eight years of age. As detailed in Chapter 7, Study 4 found that children with five-minute Apgar scores of 0-5 and 6, compared with children with Apgar score of 10, were at higher risk of scoring at/below the NMS on NAPLAN assessments at eight years of age. However, the potential mechanisms by which Apgar scores of below 6 could affect children's educational outcomes remain unclear. One possible explanation is that low Apgar scores may co-occur with markers of hypoxia and hypoxic–ischemic encephalopathy,¹⁷² and prenatal diseases, including the effects of gene polymorphisms and these may predispose to poor cognitive function in later life.¹⁷⁰ Another possibility is that mild prolonged partial hypoxic injury to the brain at birth may cause low Apgar scores that may be sufficient to cause neuronal damage and affect cognition at later age.¹⁷⁰

An important contribution of Study 4 to the ongoing scientific discourse is that it investigated the link between a range of Apgar score values (i.e., Apgar score of 0-5, 6, 7, 8 and 9 (compared with 10)) and children's educational outcomes at eight years of age. It has been suggested that Apgar score could perform better when it is used as a quasi-continuous measure rather than the use of Apgar scores by grouping at familiar cut-points (<7 or <4) for research purposes. For example, Bovbjerg *et al.* in their 2019 study conducted to determine the optimal cut-point for five-minute Apgar score concluded that 'a dichotomized [dichotomised] Apgar score is not an effective proxy outcome for research purposes.¹⁶³ This is because the false-positive and false-negative rates were found to be above the required levels of accuracy. Moreover, a 2020 commentary by Frey¹⁹⁷ suggests that an Apgar score of

7-10 may not be considered as 'normal' and suggested that anything less than a perfect Apgar score of 10 may be a cause for concern.

Labour induction versus expectant management and children's educational outcomes

Study 5 aimed at estimating the effect of elective induction of labour at 39 weeks of gestation on children's educational outcomes as measured by NAPLAN tests at age eight years, compared with expectant management. A good deal is already known about the beneficial effects of induction of labour at 39 weeks of gestation as compared with expectant management on short-term perinatal outcomes, including caesarean delivery.¹⁹⁻²⁵ However, a review of literature revealed no study regarding the effect of elective induction of labour at 39 weeks of gestation versus expectant management on longer-term neurodevelopmental outcomes in children's later life. Hence, Study 5 estimated the effect of elective induction of labour at 39 weeks of gestation as compared to expectant management on children's educational outcomes at age eight years. This is important because a 2010 study by MacKay and colleagues¹¹¹ found that the risk of special educational need, compared to children born at 40 weeks, increased among children born from 37–39 weeks gestation.

As detailed in Chapter 8, Study 5 showed that elective induction of labour at 39 weeks of gestation compared with expectant management was not associated with the proportion of children scoring at or below the NMS on each NAPLAN domain (reading, writing, spelling, grammar and numeracy) at eight years of age. The findings of this study provide the first presentation of the effect of induction of labour at 39 weeks of gestation as compared with expectant management on children's educational outcomes at eight years of age.

146

9.2 Strengths and limitations

Strengths

This thesis had several strengths. First, the thesis brings together a range of issues relevant to clinical obstetric and public health practice during labour and birth, breastfeeding, neonatal health and children's educational outcomes spanning low-, middle-, and high-income country contexts.

Second, the thesis utilised two different high-quality data sources, encompassing data from low-, middle-, and high-income country settings. The two data sources included the DHS from 33 LMICs in sub-Saharan Africa and the SAECDP from Australia (high-income country). The DHS data are among the most valid and reliable interviewer-led surveys that are representative at both local and national levels in LMICs. The DHS study design, sampling and methodology were standardised and replicable with quality assurance measures. The SAECDP is an established project that encompasses high quality whole-ofpopulation linked administrative data from state and federal sources in South Australia. The use of these two different data sources allowed this thesis to examine the five research questions across diverse health system resource settings.

Finally, this thesis has applied the best practice approach in designing and analysing observational data to enhance causal inference. For instance, a qualitative *a priori* assumptions regarding the underlying biologic mechanisms between exposures, outcomes and potential confounders were encoded using DAGs before performing any statistical analyses. This is consistent with the best practice to identify confounding and other biases in contemporary epidemiology. Moreover, advanced statistical methods such as AIPW was used, which is consistent with the best practice using a potential outcomes approach, focused

on achieving exchangeability between the exposed and unexposed, to analysing observational data.¹⁹¹⁻¹⁹³ Sensitivity analyses via negative control outcome were also conducted to examine the potential for residual confounding.¹⁹⁴

Limitations

In Chapters 4-8, the limitations of each study have been discussed. In this section, I discuss the overall limitations and potential areas for future research.

Even though two of the three studies that utilised DHS data examined the effect of caesarean section on neonatal mortality and breastfeeding, it was not possible to examine the link between caesarean section and longer-term outcomes such as neurodevelopmental outcomes. This is because the DHS did not collect data on longer-term outcomes, including children's school assessments. However, the effect of caesarean section on children's school achievement has already been conducted using the SAECDP data previously.¹⁵⁷ Furthermore, the effect of instrumental delivery, as a possible obstetrical intervention during childbirth, on longer-term neurodevelopmental outcome in children's later life using SAECDP data was not included in this thesis. However, this has already been examined using similar data (SAECDP) in South Australia by Hsieh *et al.*¹⁹⁸ in 2019.

Although this thesis has advantage of using data from two sources (DHS and SAECDP) to help examine the effect of obstetrical interventions during labour and birth on short-and longterm outcomes across diverse health system resource settings, there are some concerns regarding these data sources. For instance, the DHS data were cross-sectional surveys, which relied on face-to-face interviews by trained interviewers (data collectors). Therefore, information on obstetric interventions during labour and birth (exposures) and outcomes (e.g., breastfeeding practices) were entirely dependent up on retrospective maternal reports and this may be subject to response bias, incomplete and inconsistent reporting. However, in order to overcome such problems, 'The DHS Program' performs extensive data editing operations, including imputation of incomplete dates of events (e.g., date of birth of each child and date of birth of the mother).¹⁹⁹ Moreover, prior to the implementation of the DHS, the survey questionnaires were pretested in both urban and rural clusters in each country. The results of the pilot survey were used to modify the survey questionnaire by the DHS implementing agencies.

Unlike the DHS data, the SAECDP data, which were collected on the entire population in South Australia, contain information on longer-term outcomes (e.g., ~8-year-old children's school assessments data). However, as the SAECDP data were collected by government departments largely for government purposes, it lacks the nuances of self-report and may not contain data on topics that the DHS usually collect. The combined use of the householdbased surveys and linked administrative government database in this thesis, however, could be considered as a better approach to capture the available data in public health research because the limitations and advantages of these data sources may balance each other.

As it was highlighted in Study 1, caesarean section rate was defined as the number of caesarean births over the total number of live births expressed as a percentage, by adopting the definition used in the existing literature.⁴⁵ However, it has been suggested that a better measure of caesarean section rate is when it includes the proportion of all births, including stillbirths.⁵ As such data were not available for present research, it would be important to include this when defining caesarean section rate in future research. In Study 2, the interpretation of the association between caesarean birth and neonatal mortality was provided

in view of contextual factors in Ethiopia using the 'Three Delays Model'. However, this interpretation may not reflect the context of other LMICs because the base rate of caesarean delivery is very low in Ethiopia. Thus, it would be important to explore the association between caesarean birth and neonatal mortality within the context of other LMICs in Africa or elsewhere in future research. In Study 3, delayed initiation of breastfeeding following caesarean birth was found using DHS data from 33 countries in sub-Saharan Africa. However, as it is difficult to distinguish whether the caesarean section was medically indicated or not from the DHS data, it would be useful to study in future whether delayed initiation of breastfeeding occurs in non-medically indicated caesarean births. This is important because adverse health outcomes related to non-medically indicated caesarean births could be addressed in institutional settings. Studies 4 and 5 have examined the effects of obstetrical interventions (Apgar score and elective labour induction) using data from highincome country context. Further research is required to add to knowledge about the effect of obstetrical interventions, more specifically Apgar score and labour induction, on neurodevelopmental outcomes in children's later life in LMICs in sub-Saharan Africa, where there is no evidence.

9.3 Implications

This thesis has a number of implications. First, the findings of each study in this thesis add to a growing body of evidence regarding the effect of obstetrical interventions during labour and birth on short- and long-term outcomes. This is important from clinical medicine and public health perspectives because it has documented the effects of obstetrical interventions during labour and birth on breastfeeding, neonatal mortality and children's educational outcomes across diverse health system resource settings. Second, the application of the 'Three Delays Model' in Study 2 to understand and interpret the increased risk of neonatal mortality following caesarean birth within the context of Ethiopia may guide investments in health services and help provide context-specific interventions. For instance, in resource constrained areas, neonatal mortality can be reduced by increasing timely access to caesarean section and timely decisions for caesarean delivery via increasing health service coverage, improving infrastructure, increasing the number of skilled birth attendants, improving quality of care and increasing awareness about the importance of antenatal care and health facility delivery among women.

Third, given the increase in rate of caesarean section continues to be a major public health concern, evidence regarding the links between caesarean section and both short- and long-term outcomes would be relevant to inform public health policy for intervention. For instance, in Study 3, it was found that caesarean section has a negative influence on early initiation of breastfeeding in 33 sub-Saharan African countries. Interventions that help support and promote early initiation of breastfeeding such as immediate or early skin-to-skin contact, parent education, and hand expression of breast milk (to establish and maintain adequate milk supply) would be important.²⁰⁰ Moreover, in Study 4, it was found that children with five-minute Apgar scores of 0–5 and 6, compared with Apgar score of 10, are at higher risk of scoring at/below the NMS on the NAPLAN assessments at eight years. These findings could be translated into real-world impacts and could be applied in each nation's context. For example, a more immediate intensive follow-up of babies born with Apgar scores below 6, and educational support at school age would be important to improve children's health and neurodevelopment.

151

Finally, this thesis suggests some implications for future research. For instance, in Study 1, the disproportionately increased use of caesarean section among richest and urban women in Ethiopia were found. Thus, further work is needed to understand why caesarean section is increasingly used among wealthier and urban women in low-income countries, especially in sub-Saharan Africa to provide a more concrete basis for interventions towards appropriate use of caesarean section for the benefit of mothers and their babies.

9.4 Concluding remarks

The findings from this thesis present a comprehensive analyses of the effect of obstetrical interventions during labour and birth on breastfeeding, neonatal mortality and children's educational outcomes at eight years of age by utilising data from low-, middle-, and high-income countries. The findings of Study 1 highlighted that there were large disparities in caesarean section use in Ethiopia, demonstrating unequal access. The results from Studies 2, 3 and 4 suggest that obstetrical interventions during labour and birth (caesarean section and Apgar score) have an influence on neonatal mortality, breastfeeding and children's educational outcomes at later age. However, the findings of Study 5 suggest that elective induction of labour at 39 weeks of gestation as compared with expectant management did not affect children's educational outcomes at eight years of age.

152

Appendices

Appendix A: Online Supplementary Material for Study 1

Appendix A contains the published version of the online supplementary material for Study 1:

Yisma E, Smithers LG, Lynch JW, Mol BW. Cesarean section in Ethiopia: prevalence and sociodemographic characteristics. *J Matern Fetal Neonatal Med* 2019; **32**(7): 1130-1135.

Table 4a. Log-Poisson regression analysis showing prevalence ratios for caesarean section in

	Unadjusted PR (95% CI)	Adjusted PR ^a (95% CI)
	(N=1,571)	(N=1,571)
Types of residence	× ' '	
Urban	1	1
Rural	0.74 (0.47, 1.19)	0.97 (0.53, 1.77)
Region		
Tigray	1	1
Afar	1.19 (0.69, 2.07)	1.23 (0.68, 2.23)
Amhara	0.50 (0.23, 1.08)	0.49 (0.23, 1.04)
Oromia	0.26 (0.12, 0.56)	0.26 (0.12, 0.56)
Somali	0.36 (0.14, 0.91)	0.5 (0.20, 1.28)
Benishangul-Gumuz	0.57 (0.28, 1.16)	0.58 (0.28, 1.21)
SNNP	0.79 (0.44, 1.40)	0.75 (0.42, 1.33)
Gambela	1.27 (0.81, 1.99)	1.52 (0.91, 2.53)
Harari	0.87 (0.55, 1.36)	0.90 (0.56, 1.43)
Addis Ababa	1.06 (0.72, 1.55)	0.95 (0.61, 1.49)
Dire Dawa	0.62 (0.39, 0.99)	0.69 (0.42, 1.12)
Mother's age at birth		
<20	1	
20-34	1.11 (0.63, 1.95)	1.45 (0.80, 2.61)
35-49	1.49 (0.61, 3.64)	2.69 (1.07, 6.75)
Mother's education		
No education	1	1
Primary	1.83 (1.00, 3.34)	1.68 (0.88, 3.20)
Secondary	2.35 (1.27, 4.35)	1.91 (1.00, 3.63)
Higher	2.03 (1.03, 4.00)	1.81 (0.86, 3.79)
Place of delivery		
Public	1	1
Private	1.42 (0.90, 2.26)	1.18 (0.82, 1.71)
Birth order		
1	1	1
2-3	0.59 (0.39, 0.88)	0.59 (0.38, 0.90)
4+	0.57 (0.29, 1.13)	0.60 (0.28, 1.27)
Sex of child		
Male	1	1
Female	(0.52, 1.19)	(0.57, 1.29)
Wealth quantile		
Poorest	1	
Poorer	2.62(0.83, 8.29)	3.18(0.97, 10.41)

relation to sociodemographic characteristics using Ethiopia DHS 2011

Middle	3.17 (1.19, 8.46)	3.67 (1.44, 9.34)
Richer	1.12 (0.38, 3.28)	1.36 (0.47, 3.89)
Richest	2.47 (1.14, 5.35)	1.69 (0.73, 3.91)

Notes: In this table, it is assumed that women who did not give birth in health facility did not receive a caesarean section.

^aMutually adjusted.

Abbreviations: PR, prevalence ratio; CI, confidence interval.

Table 4b. Log-Poisson regression analysis showing prevalence ratios for caesarean section in

	Unadjusted PR (95% CI)	Adjusted PR ^a (95% CI)		
	(N=997)	(N=997)		
Types of residence				
Urban	1	1		
Rural	0.64 (0.40, 1.02)	0.73 (0.36, 1.48)		
Region				
Tigray	1	1		
Afar	1.09 (0.27, 4.34)	0.94 (0.25, 3.52)		
Amhara	1.30 (0.52, 3.25)	1.10 (0.44, 2.76)		
Oromia	1.20 (0.50, 2.87)	1.36 (0.55, 3.34)		
Somali	1.39 (0.45, 4.30)	1.16 (0.34, 3.96)		
Benishangul-Gumz	0.22 (0.03, 1.74)	0.25 (0.03, 1.88)		
SNNP	1.84 (0.82, 4.15)	2.49 (1.08, 5.74)		
Gambela	0.77 (0.30, 1.98)	1.15 (0.42, 3.15)		
Harari	0.74 (0.32, 1.72)	0.62 (0.27, 1.45)		
Addis Ababa	1.45 (0.68, 3.08)	1.05 (0.49, 2.25)		
Dire Dawa	0.95 (0.41, 2.24)	0.79 (0.33, 1.89)		
Mother's age at birth				
<20	1	1		
20-34	1.17 (0.68, 2.03)	1.31 (0.75, 2.26)		
35-49	0.89 (0.38, 2.11)	1.44 (0.56, 3.73)		
Mother's education				
No education	1	1		
Primary	0.64 (0.32, 1.28)	0.49 (0.24, 1.02)		
Secondary	1.48 (0.91, 2.39)	0.98 (0.52, 1.85)		
Higher	1.66 (0.86, 3.21)	1.24 (0.57, 2.70)		
Place of delivery				
Public	1	1		
Private	1.04 (0.54, 2.00)	1.13 (0.60, 2.12)		
Birth order				
1	1	1		

relation to sociodemographic characteristics using Ethiopia DHS 2005

2-3	0.77 (0.51, 1.17)	0.75 (0.50, 1.13)
4+	0.52 (0.27, 0.97)	0.43 (0.21, 0.86)
Sex of child		
Male	1	1
Female	1.04 (0.70, 1.56)	0.96 (0.66, 1.40)
Wealth quantile		
Poorest	1	1
Poorer	4.87 (0.64, 36.84)	4.85 (0.62, 38.11)
Middle	1.98 (0.24, 16.28)	1.38 (0.16, 12.10)
Richer	2.53 (0.36, 17.78)	2.41 (0.33, 17.58)
Richest	4.62 (0.74, 28.75)	3.04 (0.42, 22.23)

Notes: In this table, it is assumed that women who did not give birth in health facility did not receive a caesarean section.

^aMutually adjusted.

Abbreviations: PR, prevalence ratio; CI, confidence interval.

 Table 4c. Log-Poisson regression analysis showing prevalence ratios for caesarean section in

relation to sociodemographic characteristics using Ethiopia DHS 2000.

	Unadjusted PR (95% CI)	Adjusted PR ^a (95% CI)
	(n=1,041)	(n=1,041)
Types of residence		
Urban	1	1
Rural	0.59 (0.31, 1.14)	0.88 (0.40, 1.94)
Region		
Tigray	1	1
Afar	3.47 (1.07, 11.22)	2.63 (0.68, 10.18)
Amhara	0.31 (0.05, 1.73)	0.49 (0.07, 3.39)
Oromia	1.58 (0.48, 5.20)	1.41 (0.36, 5.45)
Somali	2.11 (0.58, 7.73)	2.54 (0.54, 11.80)
Benishangul-Gumz	2.40 (0.72, 8.01)	2.77 (0.68, 11.32)
SNNP	1.22 (0.35, 4.34)	1.53 (0.37, 6.33)
Gambela	0.69 (0.19, 2.53)	0.70 (0.16, 3.02)
Harari	1.11 (0.36, 3.39)	0.87 (0.23, 3.28)
Addis Ababa	1.02 (0.35, 2.98)	0.77 (0.21, 2.82)
Dire Dawa	0.65 (0.19, 2.18)	0.57 (0.14, 2.33)
Mother's age at birth		
<20	1	1
20-34	0.80 (0.41, 1.58)	1.14 (0.59, 2.22)

35-49	0.18 (0.07, 0.51)	1.28 (0.35, 4.71)
Mother's education		
No education	1	1
Primary	2.31 (1.00, 5.36)	1.91 (0.93, 3.92)
Secondary	3.76 (1.78, 7.95)	2.31 (1.12, 4.76)
Higher	4.27 (1.56, 11.71)	2.63 (0.75, 9.28)
Place of delivery		(*,*,**)
Public	1	1
Private	0.55 (0.21, 1.43)	0.56 (0.19, 1.64)
Birth order		
1	1	1
2-3	0.38 (0.20, 0.74)	0.41 (0.20, 0.84)
4+	0.09 (0.04, 0.22)	0.10 (0.04, 0.30)
Sex of child		
Male	1	1
Female	0.69 (0.42, 1.15)	0.74 (0.46, 1.19)
Wealth quantile		
Poorest	1	1
Poor	1.44 (0.15, 14.24)	2.30 (0.23, 22.54)
Middle	0.95 (0.10, 8.87)	0.61 (0.05, 7.45)
Richer	0.72 (0.08, 6.33)	0.49 (0.05, 5.05)
Richest	1.75 (0.26, 12.02)	0.95 (0.10, 8.69)

Notes: In this table, it is assumed that women who did not give birth in health facility did not receive a caesarean section.

^aMutually adjusted.

Abbreviations: PR, prevalence ratio; CI, confidence interval.

Appendix B: Online Supplementary Material for Study 2

Appendix B contains the published version (pdf) of the online supplementary material for Study 2:

Yisma E, Mol BW, Lynch JW, Smithers LG. The changing temporal association between caesarean birth and neonatal death in Ethiopia: secondary analysis of nationally representative surveys. *BMJ Open* 2019; 9(10):e027235.

SUPPLEMENTARY MATERIAL

This file includes supplementary analyses that complement the main findings and pictures that describe the 'Three Delays Model' and the Ethiopian health system structure cited in the full text of the article.

Table A1. Crude and multivariable-adjusted prevalence ratios for neonatal death

associated with 'timing of decision to conduct caesarean section' versus vaginal

delivery, Ethiopian DHS, 2016

	Prevalence ratio (95%CI)
	for neonatal death
Unadjusted (n= 10 641)	
Vaginal delivery	1 [<i>Ref.</i>]
Caesarean section decided before onset of labour	4.21 (1.34 to 13.19)
Caesarean section decided after onset of labour	2.31 (0.84 to 6.41)
Adjusted ^a (n=10 641)	
Vaginal delivery	1 [<i>Ref.</i>]
Caesarean section decided before onset of labour	3.79 (1.03 to 13.93)
Caesarean section decided after onset of labour	2.26 (0.75 to 6.82)

^aAdjusted for place of delivery, type of residence (urban/rural), sex of child, size of baby at birth, mother's age at birth, mother's education, birth order, household wealth.

NB: 'Timing of decision to conduct caesarean section'—caesarean section that was

planned before the onset of labour pains and caesarean section that was decided after the

onset of labour pains-was used as a proxy to 'types of caesarean section'.

	Survey year								
	2000		2005		2011		2016		Absolute
		Number of		Number of		Number of		Number of	change
	%	births	%	births	%	births	%	births	%
Residence									
Urban	4.4	1 277	4.4	815	3.9	1528	3.4	1 216	-1.0
Rural	4.8	10 983	3.9	10348	3.7	10344	2.8	9 807	-2.0
Region									
Tigray	5.3	788	2.7	698	4.1	753	2.7	716	-2.6
Affar	2.9	126	2.9	107	1.9	121	2.6	114	-0.3
Amhara	4.8	3 202	5.2	2621	4.5	2656	3.2	2 072	-1.6
Oromia	5.3	4 999	3.8	4411	3.4	5014	2.8	4 851	-2.5
Somali	3.8	142	3.0	477	2.9	364	4.1	508	+0.3
Benishangul- Gumuz	6.3	124	3.8	105	4.8	140	2.9	122	-3.4
SNNPR	4.0	2 602	3.4	2500	3.5	2494	2.5	2 296	-1.5
Gambela	5.3	29	2.4	31	3.6	40	2.8	27	-2.5
Harari	3.7	25	2.2	22	4.1	29	3.1	26	-0.6
Addis Ababa	2.8	182	2.9	153	2.2	222	2.1	244	-0.7
Dire Dawa	3.7	40	2.7	37	1.6	39	3.0	47	-0.7
Total	4.8	12 260	3.9	11 163	3.7	11 872	2.9	11 023	-2.0

Table A2. Trends in proportion of neonatal deaths in the 5 years before each of the surveys according to residence and region, Ethiopia DHS 2000, 2005, 2011 and 2016



Source: Soc Sci Med, 1994; 38(8): 1091-110.¹

Figure A1. The 'Three Delays Model'



Source: Soc Sci Med, 1994; **38**(8): 1091-110.¹



5



Source: Soc Sci Med, 1994; **38**(8): 1091-110.¹

Figure A3. Phase II delay, detail



Source: *Soc Sci Med*, 1994; **38**(8): 1091-110.¹

Figure A4. Phase III delay, detail

7





Source: Ethiopian Health Sector Transformation Plan, 2015,² and World Health Organization, 2017.³



References

Thaddeus S, Maine D. Too far to walk: maternal mortality in context. *Soc Sci Med* 1994;
 38(8): 1091-110.

2. Ministry of Health. Health Sector Transformation Plan. Addis Ababa: Federal Democratic Republic of Ethiopia; 2015.

3. WHO. Primary health care systems (PRIMASYS): case study from Ethiopia, abridged version. Geneva: World Health Organization; 2017. Licence: CC BY-NC-SA 3.0 IGO.
Appendix C. Online Supplementary Material for Study 3

Appendix C contains the published version (pdf) of the online supplementary material for Study 3:

Yisma E, Mol BW, Lynch JW, Smithers LG. Impact of caesarean section on breastfeeding indicators: within-country and meta-analyses of nationally representative data from 33 countries in sub-Saharan Africa. *BMJ Open* 2019; **9**(9): e027497.



Web-appendix 1. Proposed Directed Acyclic Graphs (DAGs) describing the association between caesarean birth and breastfeeding

Web-appendix 2. DAGs model code

The following **DAGs code** can used to generate the proposed DAGs in this study online at <u>http://www.dagitty.net/.</u>

Birth_Order 1 @0.558,0.687 Birth_weight 1 @0.490,0.099 Breastfeeding%20indicators O @0.677,0.369 C-%20section E @0.135,0.342 Distance_to_health_facility 1 @0.187,0.197 Gestational_age U @0.248,0.190 Mother%E2%80%99s education 1 @0.348,0.751 Mother's_age_at_birth 1 @0.426,0.076 Mother's_occupation 1 @0.521,0.720 Number_of_ANC_visit 1 @0.339,0.077 Place_of_delivery 1 @0.151,0.455 Pregnancy_complications%20(Diabetes%2C%20hypertension) U @0.235,0.113 Pregnancy_planning 1 @0.452,0.758 Region_of_residence 1 @0.590,0.618 Sex of the child 1 @0.224,0.657 Tobacco_use 1 @0.176,0.572 Types_of_place_of_residence_(urban%2Frural) 1 @0.257,0.696 Wealth_index 1 @0.544,0.141

Birth_Order Breastfeeding%20indicators @0.548,0.572 C-%20section

Birth_weight Breastfeeding%20indicators @0.511,0.205 C-%20section

C-%20section Breastfeeding%20indicators @0.667,0.361

Distance_to_health_facility Number_of_ANC_visit @0.251,0.161 Place_of_delivery @0.183,0.353 Pregnancy_complications%20(Diabetes%2C%20hypertension) @0.211,0.169 Pregnancy_planning

Gestational_age Birth_weight @0.389,0.174 C-%20section

Mother%E2%80%99s_education Breastfeeding%20indicators C-%20section Mother's_age_at_birth Mother's_occupation @0.475,0.620 Number_of_ANC_visit Place_of_delivery @0.177,0.528 Pregnancy_planning @0.404,0.732 Tobacco_use @0.208,0.612

Mother's_age_at_birth Breastfeeding%20indicators @0.581,0.298 Gestational_age @0.270,0.161 Pregnancy_complications%20(Diabetes%2C%20hypertension) @0.393,0.130 Pregnancy_planning

Mother's_occupation Breastfeeding%20indicators @0.595,0.409 C-%20section Number_of_ANC_visit Place_of_delivery @0.473,0.694 Pregnancy_planning @0.490,0.720 Types_of_place_of_residence_(urban%2Frural) @0.461,0.690 Wealth_index

Number_of_ANC_visit Breastfeeding%20indicators @0.584,0.316 C-%20section @0.269,0.235 Place_of_delivery @0.296,0.228

Place_of_delivery Breastfeeding%20indicators C-%20section @0.160,0.403

Pregnancy_complications%20(Diabetes%2C%20hypertension) Birth_weight @0.302,0.151 Breastfeeding%20indicators @0.315,0.182 C-%20section @0.229,0.168 Gestational_age Number_of_ANC_visit @0.303,0.103

Pregnancy_planning Breastfeeding%20indicators C-%20section Number_of_ANC_visit Place_of_delivery

Region_of_residence Breastfeeding%20indicators @0.612,0.533 C-%20section Distance_to_health_facility Mother%E2%80%99s_education @0.385,0.628 Place_of_delivery

Sex_of_the_child Breastfeeding%20indicators @0.567,0.461 C-%20section @0.198,0.372

Tobacco_use Birth_weight @0.220,0.603 Breastfeeding%20indicators @0.270,0.609 Gestational_age

Types_of_place_of_residence_(urban%2Frural) Breastfeeding%20indicators Distance_to_health_facility @0.257,0.485 Mother%E2%80%99s_education @0.300,0.718 Place_of_delivery @0.294,0.602 Wealth_index

Wealth_index Breastfeeding%20indicators @0.572,0.218 C-%20section Place_of_delivery @0.519,0.167 Pregnancy_planning Region_of_residence

Sex_of_the_child Breastfeeding @0.635,0.256 C-%20section @0.198,0.372

Tobacco_use Birth_weight @0.220,0.603 Breastfeeding Gestational_age

Wealth_index Breastfeeding C-%20section Place_of_delivery Pregnancy_planning Region_of_residence

Web-appendix 3. Description of variables in Directed Acyclic Graphs (DAGs)—a priori

Pregnancy planning---- Women who are planning to conceive often take antenatal care and more likely to undergo C-section if pregnancy complication occurs. Besides, they may take better care of themselves and their newborns, which in turn may affect breastfeeding practice.

Birth order--- C-section rates are higher among the first births and breastfeeding practice is low among first births. Breastfeeding practice is high among second or more births because of previous infant feeding experiences among mothers.

Tobacco use--- Previous studies found that smokers are less likely to begin or persist with breastfeeding compared to non-smokers. Moreover, due to the greater dose of nicotine that will be delivered to the newborn through the breastmilk, the infant will spent less time in active sleep.¹ Tobacco smoking will also affect caesarean section mediated through birth weight and gestational age.

Place of delivery--- Antenatal care and place of delivery may contribute to mother's breastfeeding practice. Interventions about breastfeeding through health education will promote breastfeeding practice. On the hand, caesarean birth rates are higher in private health institutions than other health facilities.²

Number of antenatal care visits---Previous studies have demonstrated that antenatal care has an effect on women's breastfeeding practices. Women who had the recommended antenatal care visits during pregnancy might know signs of pregnancy complication and this may, in turn, affect caesarean section.

Mother's education--- Educated mothers are more likely to practice breastfeeding³ and also to undergo caesarean section if obstetric complications occur. On the other hand, previous studies showed that caesarean births are higher among educated women.

Region of residence---Both exposure (caesarean section) and outcome (breastfeeding indicators) may display geographic variations mediated through sociodemographic, socioeconomic, behavioural, ethnic, cultural, and other factors.

Sex of child--- Due to sex preferences, women may more likely undergo caesarean section to save the life of the newborn when pregnancy complication occurs. It has been shown that sex of child impacts breastfeeding as a result of different factors including, sex preference, and cultural beliefs surrounding gender.

Household wealth --- Both exposure and outcome may display household wealth variations not only directly but also mediated through other factors such as place of childbirth.

Birth weight--- Birth weight affects caesarean section rates. It is highly recommended that low birth weight infants should be put to the breast as soon as possible after birth and should be exclusively breastfed until six months of life. Birth weight is also affected by gestational age, pregnancy complications and maternal tobacco use.

Mother's age at birth---is a risk factor for some caesarean births, maternal age may also affect breastfeeding practice. For example, despite inconsistent findings, some studies found that early cessation of breastfeeding was common among younger mothers and another study concluded primiparous mothers in late child-bearing aged 35 years or older are at the greatest risk of exclusive breastfeeding.⁴

Pregnancy complications (Gestational diabetes mellitus, all types of hypertension in pregnancy and others) ----Because of both exposure and outcome may display variations based on factors related to pregnancy complications directly or mediated through other factors like birth weight and gestational age, it can be assumed that these factors are important confounders of the association between caesarean section and breastfeeding.

Gestational age---It has been demonstrated that caesarean sections are usually performed at early term "37-39" and infants born at these period have lower odds of being breastfed compared to infants born after 40 weeks gestation.

Distance to health facility---Long distance to health facilities may delay access to emergency caesarean section and may also aggravate the obstetric complication encountered by mothers leading to bad outcome. On the other hand, severe obstetric complications may prevent mother from breastfeeding and healthcare providers should provide breastfeeding support (e.g., how to express breastmilk in order to initiate and maintain breastfeeding) to establish and maintain breastfeeding. Distance to health facility also affects number antenatal care visits, place of delivery, and pregnancy planning (because access to health services is limited for women who live far away from health facilities).

Web-appendix 4. Description of how variables in the current study are defined

Exposure

Delivery by caesarean section: Whether the singleton last born child was born by caesarean section.

Outcome

Breastfeeding indicators:-

- (1) Early initiation of breastfeeding: Proportion of children born in the past 2 years before the survey who were put to the breast within one hour of birth. "This indicator is based on historic recall and the numerator and denominator include both living and deceased children who were born within the past 2 years."⁵
- (2) Exclusive breastfeeding under 6 months: Proportion of infants 0–5 months of age who are fed exclusively with breast milk. DHS use this definition to calculate exclusive breastfeeding under 6 months and the numerator in the calculation of 'exclusive breastfeeding under 6 months' involves "infants 0–5 months of age who received only breast milk during the previous day while the denominator is total number of 'infants 0–5 months of age'. This indicator is based on 'recall of the previous day' and includes living infants who are residing with mothers.⁵
- (3) Children ever breastfed: Proportion of children born in the last two years who reported to have been breastfed. "This indicator is based on historic recall. The numerator and denominator include both living and deceased children who were born within the past 2 years before the survey."⁵

Variables used to eliminate or minimize confounding

Wealth quintiles: It was derived from an index (generated through principal component analysis) based on ownership of a range of assets (e.g., car, refrigerator and television), housing characteristics (e.g., material of dwelling floor and roof and main cooking fuel) and access to basic services (e.g., electricity supply, source of drinking water and sanitation facilities), etc. depending on the specific questions asked in each country. The index was standardized to a mean of 0 and a standard deviation of 1 and higher scores refer to greater wealth. DHS categorizes household wealth index into 'wealth quintiles'.

Number of antenatal visits: Women were asked whether they had received antenatal consultation from a health professional (doctor, nurse, or midwife) during the prenatal period. They were also asked the total number of antenatal visits during pregnancy. The variable used in this study reflects the number of visits reported by women who had received at least one antenatal consultation from a health professional. If women received antenatal care from a non-health professional, it was coded as zero in the 'number of antenatal visits' variable.

Birth order: The order in which a child was born.

Birth weight: All DHS in low-and middle-income countries collect information on baby's size at birth based on mother's perception by asking question, "was the newborn very large, larger than average, average, smaller than average or very small?" This is because the majority of births in these countries occur at home and birth weight is not available to be included. Previous studies conducted using DHS in low-and middle-income countries have shown that about 75% mothers are able to correctly report baby size at birth.⁶⁻⁹ In reference to these studies and due to lack of information on actual birthweight data, we used mother's recall of baby's size at birth as 'proxy to

birth weight'.¹⁰ We then computed a categorical variable "low birth weight" (very small or smaller than average) and 'normal birth weight' (average, larger than average, very large).

Mother's education: Mother's highest education level attended which is standardized and categorized as: No education, Primary, Secondary, and Higher for each country in low-and middle-income countries.

Mother's occupation: The standardized mother's occupation groups (with some country-specific variations) include: professional, technical and managerial; clerical; sales or services; skilled manual; unskilled manual; household domestic; agricultural— own land; agricultural—family land; agricultural— rented land; agricultural—other; or other.

Distance to health facility: DHS collect information regarding problems faced by women of reproductive age (15-49 years) in accessing health care to obtain medical advice or treatment for themselves when they are sick were gathered. It consisted of four questions: distance to health facility (big problem/not big problem); getting money for treatment (big problem/not big problem); getting permission to go for treatment (big problem/not big problem); getting problem); and not wanting to go alone (big problem/not big problem). Thus, we have used distance to health facility (big problem/not big problem) as a proxy to 'actual distance' in this study.

Other variables include: Pregnancy planning (yes/no), region of residence (categories include country-specific administrative areas/regions/provinces), sex of child (male/female), mother's age at birth (in years), maternal tobacco use (yes/no), place of delivery (public sector, private sector, home, and other), and types of residence (urban/rural).

Web-appendix 5. Sample STATA code from Kenya DHS 2014 to generate 'exclusive breastfeeding under 6 months' variable.

Sample: Live singleton last-born children in the past 2 years and living with their mothers

* age in months (age) gen age = v008-b3 * drop if too old or not alive keep if age<24 & b5==1

```
* recode age into groups
recode age (0/1=1 "0-1")(2/3=2 "2-3")(4/5=3 "4-5")(6/8=4 "6-8")(9/11=5 "9-11")(12/17=6 "12-17")(18/23=7
"18-23")(24/59=.), gen(child_age)
```

* tab all living children born in the last 2 years tab child_age tab child_age [iw=v005/1000000]

* keep only those children living with mother keep if b9==0

* ... and keep the last born of those drop if $_n > 0$ & caseid == caseid[_n-1]

* check the denominator tab child_age tab child_age [iw=v005/1000000]

* Breastfeeding status

gen water=0 gen liquids=0 gen milk=0 gen solids=0 gen breast=0

* Water replace water=1 if (v409>=1 & v409<=7)

* Other non-milk liquids

replace liquids=1 if v410>=1 & v410<=7 replace liquids=1 if v412c>=1 & v412c<=7 replace liquids=1 if v413>=1 & v413<=7

* **Powdered or tinned milk, formula, fresh milk** replace milk=1 if v411>=1 & v411<=7

replace milk=1 if v411a>=1 & v411a<=7

* Solid food

replace solids=1 if v412a>=1 & v412a<=7 replace solids=1 if v414e>=1 & v414e<=7 replace solids=1 if v414f>=1 & v414f<=7 replace solids=1 if v414g>=1 & v414g<=7 replace solids=1 if v414b>=1 & v414b<=7 replace solids=1 if v414i>=1 & v414i<=7 replace solids=1 if v414j>=1 & v414j<=7 replace solids=1 if v414b>=1 & v414b<=7 replace solids=1 if v414o>=1 & v414o<=7 replace solids=1 if v414p>=1 & v414p<=7 replace solids=1 if v414s>=1 & v414s<=7 replace solids=1 if v414v>=1 & v414v<=7

* Still breastfeeding

replace breast=1 if m4==95 tab1 water liquids milk solids breast

* Compute column variables used in DHS final reports

gen feeding=1 replace feeding=2 if water==1 replace feeding=3 if liquids==1 replace feeding=4 if milk==1 replace feeding=5 if solids==1 replace feeding=0 if breast==0 tab feeding,m

label define feeding 0 "Not breastfeeding" 1 "Exclusive breastfeeding" 2 "+Water" 3 "+Liquids" 4 "+Other Milk" 5 "+Solids" label val feeding feeding

tab child_age feeding [iweight=v005/1000000], row

*Compute 'exclusive breastfeeding under 6 months' (exfeeding) variable gen exfeeding=. replace exfeeding=1 if age<6 & feeding==1 replace exfeeding=0 if age<6 & feeding!=1 label define exfeeding 1 "Exclusive breastfeeding" 0 "no exclusive breastfeeding" label val exfeeding exfeeding lab var exfeeding "Exclusive breastfeeding under 6 months"

Web-appendix 6, Figures A-C. Subgroup analyses for the association between caesarean section and early initiation of breastfeeding in sub-Saharan Africa

We have conducted subgroup random-effects meta-analyses to explore potential sources of heterogeneity in the association between caesarean section and early initiation of breastfeeding in sub-Saharan Africa. For this purpose, we have defined the following subgroups *a priori*: (1) categorising the countries by region (according to United Nations geoscheme); (2) by rate of caesarean section categories (<5%, 5-15%, and >15%)¹¹; and (3) by prevalence of early initiation of breastfeeding categories ($\le50\%$ and >50%). Although we have conducted the subgroup analyses, the source/s of the heterogeneity remains unclear. The forest plots from our subgroup investigations are displayed in **Figures A-C**.

Country	Year	aPR (95% CI)	Weight (%)
EASTAFRICA	Ĭ		
Ethiopia	2016	0.46 (0.32 to 0.66)	3.00
Burundi	2016-17	0 40 (0 33 to 0 47)	3 85
Comoros	2012	- 0.52 (0.33 to 0.83)	2 29
Mozambique	2011	0.79 (0.70 to 0.90)	3.63
Rwanda	2014-15	0.50 (0.45 to 0.57)	3.91
Zambia	2014-15	0.56 (0.47 to 0.68)	3 59
Tanzania	2015-16	0.24 (0.17 to 0.33)	3.78
Zimbabwe	2015	0.35 (0.25 to 0.49)	3.46
Malawi	2015-16	0.67 (0.60 to 0.75)	3.82
Kenva	2014	0.49 (0.39 to 0.61)	3.55
Llaanda	2014	0.49 (0.42 to 0.57)	3.82
Subtotal (I-squar	ed = 90.7% p = 0.000)	0.50 (0.42 to 0.57)	38.69
			201000
SOUTHERN AFR	ICA		
Namibia	2013	0.71 (0.62 to 0.80)	3.71
Lesotho	2014	0.59 (0.46 to 0.76)	3.18
South Africa	2016	0.89 (0.78 to 1.00)	3.55
Subtotal (I-squar	ed = 82.4%, p = 0.003)	> 0.74 (0.58 to 0.89)	10.44
WEST AFRICA			
Burkina Faso	2010	0.77 (0.60 to 1.00)	2.72
Togo	2013-14	0.55 (0.45 to 0.68)	3.50
Benin	2017-18	0.47 (0.38 to 0.59)	3.59
Cote d'Ivoire	2011-12	0.48 (0.26 to 0.90)	1.78
Senegal	2010-11	0.57 (0.42 to 0.78)	2.90
Gambia	2013	0.77 (0.47 to 1.25)	1.40
Ghana	2014	0.46 (0.35 to 0.59)	3.46
Guinea	2012	0.47 (0.15 to 1.46)	0.64
Liberia	2013	0.45 (0.30 to 0.69)	2.76
Mali	2012-13	0.60 (0.47 to 0.77)	3 18
Niger	2012	0.58 (0.42 to 0.81)	2 76
Nigeria	2013	0.49 (0.38 to 0.64)	3 37
Sierra Leone	2013	- 0.68 (0.56 to 0.84)	3 28
Subtotal (I-squar	ed = 25.6%, p = 0.186)	0.55 (0.50 to 0.60)	35.35
	A	0.00/0.051 / 511	0.70
Cameroon	2011	0.82 (0.65 to 1.04)	2.76
Congo Bra	2011-12	- 0.44 (0.23 to 0.85)	1.85
Congo DR	2013-14	0.44 (0.34 to 0.57)	3.50
Gabon		0.29 (0.15 to 0.56)	2.67
Chad	2014-15	- 0.35 (0.15 to 0.82)	1.69
Angola	2015-16	0.47 (0.34 to 0.67)	3.04
Subtotal (I-squar	ed = 68.9%, p = 0.007)	0.48 (0.33 to 0.62)	15.52
Overall (I-square	d = 84.5%, p = 0.000)	0.54 (0.48 to 0.60)	100.00
NOTE: Weights a	re from random effects analysis		
	.1	1 1.5	

Figure A. Subgroup analyses for the association between caesarean section and early initiation of breastfeeding in sub-Saharan Africa stratified by geographic regions

Country	Year	aPR (95% CI)	Weight (%)
CS RATE LESS	THAN 5%		
Niger	2012	0.58 (0.42 to 0.81)	2.76
Cote d'Ivoire	2011-12	0.48 (0.26 to 0.90)	1.78
Zambia	2014-15	0.56 (0.47 to 0.68)	3.59
Burkina Faso	2010	0.77 (0.60 to 1.00)	2.72
Angola	2015-16	0.47 (0.34 to 0.67)	3.04
Guinea	2012	0.47 (0.15 to 1.46)	0.64
Cameroon	2011	0.82 (0.65 to 1.04)	2.76
Senegal	2010-11	0.57 (0.42 to 0.78)	2.90
Chad	2014-15	0.35 (0.15 to 0.82)	1.69
Sierra Leone	2013	• 0.68 (0.56 to 0.84)	3.28
Gambia	2013	0.77 (0.47 to 1.25)	1.40
Liberia	2013	0.45 (0.30 to 0.69)	2.76
Mali	2012-13	0.60 (0.47 to 0.77)	3.18
Nigeria	2013	0.49 (0.38 to 0.64)	3.37
Benin	2017-18	0.47 (0.38 to 0.59)	3.59
Ethiopia	2016	0.46 (0.32 to 0.66)	3.00
Mozambique	2011	- 0.79 (0.70 to 0.90)	3.63
Subtotal (I-squa	ed = 62.5%, p = 0.000)	0.58 (0.52 to 0.65)	46.10
landa da d	CALL SHELL IN THE REAL		
CS RATE 5-15%	2015 10	0.07 (0.00 to 0.75)	2.02
Comoroo	2010-10	0.57 (0.60 to 0.73)	3.02
Contoios	2012	0.52 (0.33 to 0.63)	2.29
Congo DR	2013-14	0.44 (0.34 to 0.37)	3.50
Burundi	2016-17	0.40 (0.33 to 0.47)	3.80
Rwanda	2014-15	0.50 (0.45 to 0.57)	3.91
Togo	2013-14	0.55 (0.45 to 0.68)	3.50
Tanzania	2015-16	0.24 (0.17 to 0.33)	3.78
Congo Bra	2011-12	- 0.44 (0.23 to 0.85)	1.85
Uganda	2016	0.49 (0.42 to 0.57)	3.82
kenya	2014	0.49 (0.39 to 0.61)	3.55
Lesotho	2014	0.59 (0.46 to 0.76)	3.18
Zimbabwe	2015	0.35 (0.25 to 0.49)	3.46
Ghana	2014	0.46 (0.35 to 0.59)	3.46
Gabon	2012	0.29 (0.15 to 0.56)	2.67
Subtotal (I-squa	ed = 82.9%, p = 0.000)	0.46 (0.39 to 0.53)	46.64
C <mark>S RA</mark> TE GREA	ER THAN 15%		
Namibia	2013	0.71 (0.62 to 0.80)	3.71
South Africa	2016	0.89 (0.78 to 1.00)	3.55
Subtotal (I-squa	ed = 83.8%, p = 0.013)	0.80 (0.62 to 0.97)	7.26
Overall (I-square	d = 84.5%, p = 0.000)	0.54 (0.48 to 0.60)	100.00
NOTE: Weights a	re from random effects analysis		
	.1	1 1.5	

Figure B. Subgroup analyses for the association between caesarean section and early initiation of breastfeeding in sub-Saharan Africa stratified by caesarean section rates

Country	Year	aPR (95% CI)	Weight (%)
PREVALENCE BF	WITHIN 1 HOUR> 50%		
Ethiopia	2016		3.00
Namibia	2013	0.71 (0.62 to 0.80)	3.71
Burundi	2016-17	0.40 (0.33 to 0.47)	3.85
Тодо	2013-14		<mark>3.50</mark>
Benin	2017-18	0.47 (0.38 to 0.59)	3.59
Congo DR	2013-14	0.44 (0.34 to 0.57)	3.50
Gambia	2013	0.77 (0.47 to 1.25)	1.40
Ghana	2014	0.46 (0.35 to 0.59)	3.46
Liberia	2013	0.45 (0.30 to 0.69)	2.76
Mali	2012-13	0.60 (0.47 to 0.77)	3.18
Mozambique	2011	0.79 (0.70 to 0.90)	3.63
Niger	2012	0.58 (0.42 to 0.81)	2.76
Rwanda	2014-15	• 0.50 (0.45 to 0.57)	3.91
Sierra Leone	2013	0.68 (0.56 to 0.84)	3.28
Zambia	2014-15	0.56 (0.47 to 0.68)	3.59
Tanzania	2015-16	• 0.24 (0.17 to 0.33)	3.78
Zimbabwe	2015		3.46
Malawi	2015-16	• 0.67 (0.60 to 0.75)	3.82
Lesotho	2014	0.59 (0.46 to 0.76)	3.18
Kenva	2014	- 49 (0.39 to 0.61)	3.55
Udanda	2016	0.49 (0.42 to 0.57)	3.82
South Africa	2016	0.89 (0.78 to 1.00)	3.55
Subtotal (I-square	ed = 88.6%, p = 0.000)	• 0.55 (0.48 to 0.61)	74.29
PREVALENCE BE		1 %	
Burkina Faso	2010	0 77 (0 60 to 1 00)	2 72
Cameroon	2010	0.82 (0.65 to 1.04)	2 76
Comoros	2012	0.52 (0.33 to 0.83)	2.29
Congo Bra	2011-12	0 44 (0 23 to 0 85)	1.85
Cote d'Ivoire	2011-12	0.48 (0.26 to 0.90)	1 78
Gabon	2012		2.67
Senegal	2010-11	0.57 (0.42 to 0.78)	2.90
Guinea	2010 11	0.47 (0.15 to 1.46)	0.64
Nigeria	2012	0.49 (0.16 to 1.45)	3 37
Chad	2010	0.35 (0.56 to 0.82)	1.69
Angola	2015-16	0.00 (0.10 to 0.02)	3.04
Subtotal (Lequare	2010-10 d = 54.3% $n = 0.016$	0.53 (0.43 to 0.63)	25 71
oubioiai (i-squaic	a = 04.070, p = 0.010)	0.00 (0.40 to 0.00)	20.11
Overall (I-squared	d = 84.5%, p = 0.000)	0 .54 (0.48 to 0.60)	100.00
NOTE: Weights ar	e from random effects ar	nalysis	
		1 1 15	

Figure C. Subgroup analyses for the association between caesarean section and early initiation of breastfeeding in sub-Saharan Africa stratified by prevalence of early initiation of breastfeeding

References

1. Mennella JA, Yourshaw LM, Morgan LK. Breastfeeding and smoking: short-term effects on infant feeding and sleep. *Pediatrics* 2007; **120**(3): 497-502.

2. Yisma E, Smithers LG, Lynch JW, Mol BW. Cesarean section in Ethiopia: prevalence and sociodemographic characteristics. *J Matern Fetal Neonatal Med* 2019; **32**(7): 1130-5.

3. Ekubay M, Berhe A, Yisma E. Initiation of breastfeeding within one hour of birth among mothers with infants younger than or equal to 6 months of age attending public health institutions in Addis Ababa, Ethiopia. *Int Breastfeed J* 2018; **13**: 4.

4. Kitano N, Nomura K, Kido M, et al. Combined effects of maternal age and parity on successful initiation of exclusive breastfeeding. *Prev Med Rep* 2016; **3**: 121-6.

5. World Health Organization. Indicators for assessing infant and young child feeding practices. Part I: Definitions. Conclusions of a consensus meeting held 6–8 November 2007 in Washington D.C., USA. Geneva: World Health Organization, 2008.

6. Rahman MS, Howlader T, Masud MS, Rahman ML. Association of Low-Birth Weight with Malnutrition in Children under Five Years in Bangladesh: Do Mother's Education, Socio-Economic Status, and Birth Interval Matter? *PLOS ONE* 2016; **11**(6): e0157814.

7. Sreeramareddy CT, Shidhaye RR, Sathiakumar N. Association between biomass fuel use and maternal report of child size at birth--an analysis of 2005-06 India Demographic Health Survey data. *BMC Public Health* 2011; **11**(1): 403.

8. Haque SR, Tisha S, Huq N. Poor Birth Size a Badge of Low Birth Weight Accompanying Less Antenatal Care in Bangladesh with Substantial Divisional Variation: Evidence from BDHS-2011. *Public Health Research* 2015; **5**(6): 184-91.

9. Khanal V, Sauer K, Karkee R, Zhao Y. Factors associated with small size at birth in Nepal: further analysis of Nepal Demographic and Health Survey 2011. *BMC Pregnancy Childbirth* 2014; **14**: 32.

10. Channon AA. Can mothers judge the size of their newborn? Assessing the determinants of a mother's perception of a baby's size at birth. *J Biosoc Sci* 2011; **43**(5): 555-73.

11. Kyu HH, Shannon HS, Georgiades K, Boyle MH. Caesarean delivery and neonatal mortality rates in 46 low- and middle-income countries: a propensity-score matching and meta-analysis of Demographic and Health Survey data. *Int J Epidemiol* 2013; **42**(3): 781-91.

Appendix D: Online Supplementary Material for Study 4

Appendix D includes supplementary information and analyses that complement Study 4.

Yisma E, Mol BW, Lynch JW, Murthy NM, Smithers LG. Associations between Apgar

scores and children's educational outcomes at eight years of age. Aust N Z J Obstet Gynaecol

2020; 1–7.

Figure S1. Proposed Directed Acyclic Graphs (DAGs) describing the associations between Apgar scores and children's educational outcomes

Table S1. Proportion of children performing at or below the NMS according to 5-minute Apgar scores and risk differences comparing Apgar scores of <7 with scores of 7-10 for children performing at or below the NMS for each NAPLAN domain

Table S2. Risk differences comparing a range of 5-minute Apgar scores (i.e., 0-5, 6, 7, 8 and 9 (compared with 10) for children having school card according to NAPLAN domains

Table S3. Proportion of children performing at or below the NMS according to 1-minute Apgar scores and risk differences comparing Apgar scores of <7 with scores of 7-10 for children performing at or below NMS for each domain

Table S4. Proportion of children having school card according to 1-minute Apgar score and risk differences comparing Apgar scores of <7 with scores of 7-10 for children having school card

Table S5. Risk differences comparing a range of 1-minute Apgar score values (i.e., 0-5, 6, 7,8 and 9 (compared with 10) for children performing at or below the NMS for each domain



Figure S1. Proposed Directed Acyclic Graphs (DAGs) describing the associations between Apgar scores and children's educational outcomes

Table S1. Proportion of children performing at or below the NMS according to 5-minute Apgar scores and risk differences comparing Apgar scores of <7 with scores of 7-10 for children performing at or below the NMS for each NAPLAN domain

		Apgar <7	Apgar 7-10			
Domain	Ν	≤NMS/total (%)	≤NMS/total (%)	RD [‡]	95% CI	Р
Reading	60,722	138/604 (22.85%)	11929/60118 (19.84%)	0.02	-0.01, 0.05	0.283
Writing	60,514	84/593 (14.17%)	7366/59921 (12.29%)	0.02	-0.01, 0.05	0.127
Spelling	60,830	149/601 (24.79%)	12431/60229 (20.64%)	0.03	0.00, 0.06	0.078
Grammar	60,830	135/601 (22.46%)	11652/60229 (19.35%)	0.01	-0.02, 0.05	0.351
Numeracy	60,581	159/601 (26.46%)	13353/59980 (22.26%)	0.03	-0.00, 0.07	0.080

Abbreviations: RD, risk difference; NMS, National Minimum Standard; CI, confidence interval.

[‡]RD calculated from augmented inverse probability weighted analysis (adjusted for a number of confounders listed in methods section) of the proportion of children performing \leq NMS on each assessment for children with low versus normal 5-minute Apgar score.

Table S2. Risk differences comparing a range of 5-minute Apgar scores (i.e., 0-5, 6, 7, 8 and 9 (compared with 10) for children having school

card according to NAPLAN domains

	School card—Reading	School card—Writing	School card—Spelling	School card—Grammar	School card—Numeracy
5-minute Apgar	<i>n</i> =60,722	n=60,514	n=60,830	n=60,830	n=60,581
scores	RD [‡] (95%CI)	RD (95%CI)	RD (95%CI)	RD (95%CI)	RD (95%CI)
0-5 vs 10	0.02 (-0.24, 0.27)	0.01 (-0.25, 0.27)	0.02 (-0.24, 0.27)	0.02 (-0.24, 0.27)	0.02 (-0.24, 0.28)
6 vs 10	0.3 (0.08, 0.52)	0.25 (0.03, 0.47)	0.25 (0.03, 0.46)	0.25 (0.03, 0.46)	0.28 (0.06, 0.49)
7 vs 10	0.08 (-0.05, 0.22)	0.06 (-0.07, 0.19)	0.05 (-0.08, 0.18)	0.05 (-0.08, 0.18)	0.07 (-0.06, 0.2)
8 vs 10	0.09 (0.02, 0.17)	0.09 (0.02, 0.16)	0.09 (0.02, 0.16)	0.09 (0.02, 0.16)	0.09 (0.02, 0.16)
9 vs 10	0.00 (-0.03, 0.03)	-0.00 (-0.03, 0.03)	0.00 (-0.03, 0.03)	0.00 (-0.03, 0.03)	0.00 (-0.03, 0.04)
10	(ref.)	(ref.)	(ref.)	(ref.)	(ref.)

Abbreviations: RD, Risk difference; NMS, National Minimum Standard; CI, confidence interval.

[‡]**RD** calculated from augmented inverse probability weighted analysis (adjusted for a number of confounders listed in methods section) showing the proportion of children performing \leq NMS on each assessment for children with Apgar scores at 5 minutes.

Associations between 1-minute Apgar scores and children's educational outcomes at 8 years of age

Table S3. Proportion of children performing at or below the NMS according to 1-minute Apgar scores and risk differences comparing Apgar scores of <7 with scores of 7-10 for children performing at or below the NMS for each domain

		Apgar <7	Apgar 7-10			
NAPLAN domain	Ν	≤NMS/total (%)	≤NMS/total (%)	RD [‡]	95% CI	Р
Reading	60,700	1508/7127 (21.16%)	10554/53573 (19.70%)	0.01	-0.00, 0.02	0.281
Writing	60,490	967/7099 (13.62%)	6481/53391 (12.14%)	0.01	0.00, 0.02	0.045
Spelling	60,806	1641/7157 (22.93%)	10934/53649 (20.38%)	0.02	0.01, 0.03	0.001
Grammar	60,806	1476/7157 (20.62%)	10304/53649 (19.21%)	0.01	-0.00, 0.02	0.216
Numeracy	60,557	1647/7112 (23.16%)	11851/53445 (22.17%)	0.00	-0.01, 0.02	0.357

Abbreviations: RD, risk difference; NMS, National Minimum Standard; CI, confidence interval.

* **RD** calculated from augmented inverse probability weighted analysis (adjusted for a number of confounders listed in methods section) of the percentage of children performing \leq NMS on each assessment for children with low compared with normal Apgar score at 1 minute.

Table S4. Proportion of children having school card according to 1-minute Apgar scores and risk differences comparing Apgar scores of <7 with scores of 7-10 for children having school card

		Apgar <7	Apgar 7-10			
School card [§]	Ν	School card/total (%)	School card/total (%)	RD‡	95% CI	Р
Reading—school card	60,700	1590/7127 (22.31%)	11344/53573 (21.17%)	0.01	0.00, 0.02	0.042
Writing—school card	60,490	1578/7099 (22.23%)	11283/53391 (21.13%)	0.01	0.00, 0.02	0.049
Spelling—school card	60,806	1599/7157 (23.34%)	11374/53649 (21.20%)	0.01	0.00, 0.02	0.047
Grammar—school card	60,806	1599/7157 (23.34%)	11374/53649 (21.20%)	0.01	0.00, 0.02	0.047
Numeracy—school card	60,557	1576/7112 (22.16%)	11351/53445 (21.17%)	0.01	-0.00, 0.02	0.069

Abbreviations: RD, risk difference; CI, confidence interval.

[§]School card possession variable was computed for each assessment for children who had complete data on each NAPLAN domain and all the confounders.

‡RD calculated from augmented inverse propensity weighted estimator analysis (adjusted for a number of confounders listed in methods section) of the percentage of children having school card at year 3 among children with low versus normal Apgar score at 1 minute.

Table S5. Risk differences comparing a range of 1-minute Apgar score values (i.e., 0-5, 6, 7, 8 and 9 (compared with 10) for children performing at or below the NMS for each domain

	Reading	Writing	Spelling	Grammar	Numeracy
1-minute Apgar scores	n=60,700	n=60,490	n=60,806	n=60,806	n=60,557
	RD [‡] (95%CI)	RD (95%CI)	RD (95%CI)	RD (95%CI)	RD (95%CI)
0-5 vs 10	0.15 (-0.03, 0.33)	0.15 (-0.09, 0.40)	0.18 (0.00, 0.36)	0.10 (-0.08, 0.27)	0.02 (-0.13, 0.16)
6 vs 10	0.15 (-0.03, 0.33)	0.17 (-0.08, 0.42)	0.18 (0.00, 0.36)	0.10 (-0.08, 0.28)	0.04 (-0.11, 0.19)
7 vs 10	0.15 (-0.02, 0.32)	0.15 (-0.09, 0.38)	0.14 (-0.03, 0.31)	0.10 (-0.07, 0.27)	0.03 (-0.11, 0.17)
8 vs 10	0.14 (-0.03, 0.30)	0.07 (-0.14, 0.29)	0.11 (-0.05, 0.27)	0.10 (-0.07, 0.27)	0.02 (-0.11, 0.16)
9 vs 10	0.11 (-0.05, 0.27)	0.08 (-0.14, 0.29)	0.07 (-0.08, 0.22)	0.05 (-0.11, 0.21)	0.00 (-0.13, 0.13)
10	(ref.)	(ref.)	(ref.)	(ref.)	(ref.)

Abbreviations: RD, calculated from augmented inverse probability weighted analysis showing the proportion of children performing $\leq NMS$ on each assessment for children with Apgar scores at one minutes after birth. NB: Multinomial logit model was used to predict Apgar score (treatment model), using all confounders specified for outcome model (logit). In order to aid the interpretation of RDs, we have expressed the RDs as proportions (percentages) of the Potential Outcome Mean (POM) for the control (Apgar score of 10). Therefore, RDs can be interpreted as proportion as in the binary-treatment model.

Appendix E: Online Supplementary Material for Study 5

Table S1. Proportion of children performing at Band 1 according to elective induction of labour at 39 weeks versus expectant management and ATEs comparing elective induction of labour at 39 weeks with expectant management for children performing at Band 1 for each NAPLAN domain

	Total sample	Elective labour induction	Expectant management			
Domain	(N)	Band1/total (%)	Band 1/total (%)	ATE [†]	95% CI	Р
Reading	29,114	75/1,266 (5.92%)	1718/27,848 (6.17%)	0.01	-0.01, 0.03	0.321
Writing	29,022	39/1,264 (3.09%)	1105/27,758 (3.98%)	0.00	-0.02, 0.01	0.676
Spelling	29,141	62/1,268 (4.89%)	1736/27,873 (6.23%)	0.00	-0.02, 0.02	0.963
Grammar	29,141	89/1,268 (7.02%)	2161/27,873 (7.75%)	0.01	-0.01, 0.03	0.244
Numeracy	29,031	72/1,272 (5.66%)	1616/27,759 (5.82%)	0.02	-0.00, 0.03	0.089

Abbreviations: ATE, Average Treatment Effect; CI, confidence interval; NMS, National Minimum Standard.

[†]ATE calculated from augmented inverse probability weighted analysis adjusted for maternal smoking during the second half of pregnancy, birth weight for gestational age z-score, mother's age at birth, index of relative socioeconomic disadvantage, types of antenatal care, number of antenatal visits, maternal occupation, maternal ethnicity, Australian remoteness index for areas, mother had a partner, and infant sex.

References

Marconi AM. Recent advances in the induction of labor. *F1000Res* 2019;
 8(F1000 Faculty Rev): 1829.

2. Humphrey T, Tucker JS. Rising rates of obstetric interventions: exploring the determinants of induction of labour. *J Public Health (Oxf)* 2009; **31**(1): 88-94.

3. Rayburn WF, Zhang J. Rising rates of labor induction: present concerns and future strategies. *Obstet Gynecol* 2002; **100**(1): 164-7.

4. MacDorman MF, Declercq E, Zhang J. Obstetrical intervention and the singleton preterm birth rate in the United States from 1991-2006. *Am J Public Health* 2010;
100(11): 2241-7.

5. Boerma T, Ronsmans C, Melesse DY, et al. Global epidemiology of use of and disparities in caesarean sections. *Lancet* 2018; **392**(10155): 1341-8.

 Gibbons L, Belizan JM, Lauer JA, Betran AP, Merialdi M, Althabe F. Inequities in the use of cesarean section deliveries in the world. *Am J Obstet Gynecol* 2012; **206**(4): 331 e1-19.

7. Yaya S, Uthman OA, Amouzou A, Bishwajit G. Disparities in caesarean section prevalence and determinants across sub-Saharan Africa countries. *Glob Health Res Policy* 2018; **3**: 19.

8. Vogel JP, Souza JP, Gulmezoglu AM. Patterns and Outcomes of Induction of Labour in Africa and Asia: a secondary analysis of the WHO Global Survey on Maternal and Neonatal Health. *PLoS One* 2013; **8**(6): e65612.

9. Ronsmans C, Holtz S, Stanton C. Socioeconomic differentials in caesarean rates in developing countries: a retrospective analysis. *Lancet* 2006; **368**(9546): 1516-23.

10. Seijmonsbergen-Schermers AE, van den Akker T, Rydahl E, et al. Variations in use of childbirth interventions in 13 high-income countries: A multinational cross-sectional study. *PLoS Med* 2020; **17**(5): e1003103.

11. Keag OE, Norman JE, Stock SJ. Long-term risks and benefits associated with cesarean delivery for mother, baby, and subsequent pregnancies: Systematic review and meta-analysis. *PLoS Med* 2018; **15**(1): e1002494.

12. Sandall J, Tribe RM, Avery L, et al. Short-term and long-term effects of caesarean section on the health of women and children. *Lancet* 2018; **392**(10155): 1349-57.

13. Cluver C, Novikova N, Koopmans CM, West HM. Planned early delivery versus expectant management for hypertensive disorders from 34 weeks gestation to term. *Cochrane Database Syst Rev* 2017; **1**(1): CD009273.

14. World Health Organization. WHO recommendations: induction of labour at or beyond term. Geneva: World Health Organization, 2018.

15. Shetty A, Burt R, Rice P, Templeton A. Women's perceptions, expectations and satisfaction with induced labour--a questionnaire-based study. *Eur J Obstet Gynecol Reprod Biol* 2005; **123**(1): 56-61.

16. Thavagnanam S, Fleming J, Bromley A, Shields MD, Cardwell CR. A metaanalysis of the association between Caesarean section and childhood asthma. *Clin Exp Allergy* 2008; **38**(4): 629-33.

17. Liu S, Liston RM, Joseph KS, et al. Maternal mortality and severe morbidity associated with low-risk planned cesarean delivery versus planned vaginal delivery at term. *Can Med Assoc J* 2007; **176**(4): 455-60.

18. Cho CE, Norman M. Cesarean section and development of the immune system in the offspring. *Am J Obstet Gynecol* 2013; **208**(4): 249-54.

19. Sotiriadis A, Petousis S, Thilaganathan B, et al. Maternal and perinatal outcomes after elective induction of labor at 39 weeks in uncomplicated singleton pregnancy: a meta-analysis. *Ultrasound Obstet Gynecol* 2019; **53**(1): 26-35.

20. Grobman WA, Caughey AB. Elective induction of labor at 39 weeks compared with expectant management: a meta-analysis of cohort studies. *Am J Obstet Gynecol* 2019; **221**(4): 304-10.

21. Grobman WA, Rice MM, Reddy UM, et al. Labor Induction versus Expectant Management in Low-Risk Nulliparous Women. *N Engl J Med* 2018; **379**(6): 513-23.

22. Gibbs Pickens CM, Kramer MR, Howards PP, Badell ML, Caughey AB, Hogue CJ. Term Elective Induction of Labor and Pregnancy Outcomes Among Obese Women and Their Offspring. *Obstet Gynecol* 2018; **131**(1): 12-22.

23. Sinkey RG, Blanchard CT, Szychowski JM, et al. Elective Induction of Labor in the 39th Week of Gestation Compared With Expectant Management of Low-Risk Multiparous Women. *Obstet Gynecol* 2019; **134**(2): 282-7.

24. Sinkey RG, Lacevic J, Reljic T, et al. Elective induction of labor at 39 weeks among nulliparous women: The impact on maternal and neonatal risk. *PLoS One* 2018; **13**(4): e0193169.

25. Souter V, Painter I, Sitcov K, Caughey AB. Maternal and newborn outcomes with elective induction of labor at term. *Am J Obstet Gynecol* 2019; **220**(3): 273 e1- e11.

26. Vrouenraets FP, Roumen FJ, Dehing CJ, van den Akker ES, Aarts MJ, Scheve EJ. Bishop score and risk of cesarean delivery after induction of labor in nulliparous women. *Obstet Gynecol* 2005; **105**(4): 690-7.

27. Macer JA, Macer CL, Chan LS. Elective induction versus spontaneous labor: a retrospective study of complications and outcome. *Am J Obstet Gynecol* 1992; **166**(6 Pt 1): 1690-6; discussion 6-7.

28. Vahratian A, Zhang J, Troendle JF, Sciscione AC, Hoffman MK. Labor progression and risk of cesarean delivery in electively induced nulliparas. *Obstet Gynecol* 2005; **105**(4): 698-704.

29. Maslow AS, Sweeny AL. Elective induction of labor as a risk factor for cesarean delivery among low-risk women at term. *Obstet Gynecol* 2000; **95**(6 Pt 1): 917-22.

30. Levine LD, Hirshberg A, Srinivas SK. Term induction of labor and risk of cesarean delivery by parity. *J Matern Fetal Neonatal Med* 2014; **27**(12): 1232-6.

31. Rattigan MI, Atkinson AL, Baum JD. Delivery route following elective induction of labor at term: analysis of 807 patients. *J Clin Med Res* 2013; **5**(4): 305-8.

32. Little SE, Caughey AB. Induction of Labor and Cesarean: What is the True Relationship? *Clin Obstet Gynecol* 2015; **58**(2): 269-81.

33. Prior E, Santhakumaran S, Gale C, Philipps LH, Modi N, Hyde MJ. Breastfeeding after cesarean delivery: a systematic review and meta-analysis of world literature. *Am J Clin Nutr* 2012; **95**(5): 1113-35.

34. Edmond KM, Zandoh C, Quigley MA, Amenga-Etego S, Owusu-Agyei S, Kirkwood BR. Delayed breastfeeding initiation increases risk of neonatal mortality. *Pediatrics* 2006; **117**(3): e380-6.

35. Molina G, Weiser TG, Lipsitz SR, et al. Relationship Between Cesarean Delivery Rate and Maternal and Neonatal Mortality. *JAMA* 2015; **314**(21): 2263-70.

36. Althabe F, Sosa C, Belizan JM, Gibbons L, Jacquerioz F, Bergel E. Cesarean section rates and maternal and neonatal mortality in low-, medium-, and high-income countries: an ecological study. *Birth* 2006; **33**(4): 270-7.

37. Lumbiganon P, Laopaiboon M, Gulmezoglu AM, et al. Method of delivery and pregnancy outcomes in Asia: the WHO global survey on maternal and perinatal health 2007-08. *Lancet* 2010; **375**(9713): 490-9.

38. Villar J, Carroli G, Zavaleta N, et al. Maternal and neonatal individual risks and benefits associated with caesarean delivery: multicentre prospective study. *BMJ* 2007; **335**(7628): 1025.

39. MacDorman MF, Declercq E, Menacker F, Malloy MH. Infant and neonatal mortality for primary cesarean and vaginal births to women with "no indicated risk," United States, 1998-2001 birth cohorts. *Birth* 2006; **33**(3): 175-82.

40. MacDorman MF, Declercq E, Menacker F, Malloy MH. Neonatal mortality for primary cesarean and vaginal births to low-risk women: application of an "intention-to-treat" model. *Birth* 2008; **35**(1): 3-8.

41. Kallen K, Olausson PO. Neonatal mortality for low-risk women by method of delivery. *Birth* 2007; **34**(1): 99-100; author reply 1-2.

42. Shah A, Fawole B, M'Imunya J, et al. Cesarean delivery outcomes from the WHO global survey on maternal and perinatal health in Africa. *Int J Gynaecol Obstet* 2009; **107**(3): 191-7.

43. Kyu HH, Shannon HS, Georgiades K, Boyle MH. Caesarean delivery and neonatal mortality rates in 46 low- and middle-income countries: a propensity-score matching and meta-analysis of Demographic and Health Survey data. *Int J Epidemiol* 2013; **42**(3): 781-91.

44. Ye J, Zhang J, Mikolajczyk R, Torloni MR, Gulmezoglu AM, Betran AP. Association between rates of caesarean section and maternal and neonatal mortality in the 21st century: a worldwide population-based ecological study with longitudinal data. *BJOG* 2016; **123**(5): 745-53.

45. Betran AP, Merialdi M, Lauer JA, et al. Rates of caesarean section: analysis of global, regional and national estimates. *Paediatr Perinat Epidemiol* 2007; **21**(2): 98-113.

46. Ye J, Betran AP, Guerrero Vela M, Souza JP, Zhang J. Searching for the optimal rate of medically necessary cesarean delivery. *Birth* 2014; **41**(3): 237-44.

47. Jurdi R, Khawaja M. Caesarean section rates in the Arab region: a cross-national study. *Health Policy Plan* 2004; **19**(2): 101-10.

48. McClure EM, Goldenberg RL, Bann CM. Maternal mortality, stillbirth and measures of obstetric care in developing and developed countries. *Int J Gynaecol Obstet* 2007; **96**(2): 139-46.

49. Volpe FM. Correlation of Cesarean rates to maternal and infant mortality rates:
an ecologic study of official international data. *Rev Panam Salud Publica* 2011; 29(5):
303-8.

50. Zizza A, Tinelli A, Malvasi A, et al. Caesarean section in the world: a new ecological approach. *J Prev Med Hyg* 2011; **52**(4): 161-73.

51. Belizan JM, Althabe F, Barros FC, Alexander S. Rates and implications of caesarean sections in Latin America: ecological study. *BMJ* 1999; **319**(7222): 1397-400.

52. Tribe RM, Taylor PD, Kelly NM, Rees D, Sandall J, Kennedy HP. Parturition and the perinatal period: can mode of delivery impact on the future health of the neonate? *J Physiol* 2018; **596**(23): 5709-22.

53. Walker WA. The importance of appropriate initial bacterial colonization of the intestine in newborn, child, and adult health. *Pediatr Res* 2017; **82**(3): 387-95.

54. Borre YE, O'Keeffe GW, Clarke G, Stanton C, Dinan TG, Cryan JF. Microbiota and neurodevelopmental windows: implications for brain disorders. *Trends Mol Med* 2014; **20**(9): 509-18.

55. Underwood MA, German JB, Lebrilla CB, Mills DA. Bifidobacterium longum subspecies infantis: champion colonizer of the infant gut. *Pediatr Res* 2015; 77(1-2): 229-35.

56. Lagercrantz H. The good stress of being born. *Acta Paediatr* 2016; 105(12): 1413-6.

57. Dahlen HG, Kennedy HP, Anderson CM, et al. The EPIIC hypothesis: intrapartum effects on the neonatal epigenome and consequent health outcomes. *Med Hypotheses* 2013; **80**(5): 656-62.

58. Welberg LA, Seckl JR. Prenatal stress, glucocorticoids and the programming of the brain. *J Neuroendocrinol* 2001; **13**(2): 113-28.

59. Murphy SK, Jirtle RL. Imprinting evolution and the price of silence. *Bioessays*2003; 25(6): 577-88.

60. Schlinzig T, Johansson S, Gunnar A, Ekstrom TJ, Norman M. Epigenetic modulation at birth - altered DNA-methylation in white blood cells after Caesarean section. *Acta Paediatr* 2009; **98**(7): 1096-9.

61. Almgren M, Schlinzig T, Gomez-Cabrero D, et al. Cesarean delivery and hematopoietic stem cell epigenetics in the newborn infant: implications for future health? *Am J Obstet Gynecol* 2014; **211**(5): 502 e1-8.

62. Wright ML, Starkweather AR, York TP. Mechanisms of the Maternal Exposome and Implications for Health Outcomes. *ANS Adv Nurs Sci* 2016; **39**(2): E17-30.

63. Hillman NH, Kallapur SG, Jobe AH. Physiology of transition from intrauterine to extrauterine life. *Clin Perinatol* 2012; **39**(4): 769-83.

64. Australian Institute of Health and Welfare. National Core Maternity Indicators. Canberra: AIHW, 2019.

65. Martin JA HB, Osterman MJK, Driscoll AK, Births: Final data for 2018. National Vital Statistics Reports; vol 68, no13. Hyattsville: MD: National Center for Health Statistics, 2019. 66. EURO-PERISTAT Project with SCPE and EUROCAT. European Perinatal Health Report. The health and care of pregnant women and babies in Europe in 2010. May 2013: Available <u>www.europeristat.com</u>.

67. Guerra GV, Cecatti JG, Souza JP, et al. Factors and outcomes associated with the induction of labour in Latin America. *BJOG* 2009; **116**(13): 1762-72.

68. Leduc D, Biringer A, Lee L, Dy J, et al. Induction of labour. *J Obstet Gynaecol Can* 2013; **35**(9): 840-57.

69. Queensland Clinical Guidelines. Induction of labour. Guideline No. MN17.22-V7-R22. [Internet]. Queensland Health. March 2017. Available from https://www.health.qld.gov.au/ data/assets/pdf_file/0020/641423/g-iol.pdf.

70. ACOG. ACOG Practice Bulletin No. 107: Induction of labor. *Obstet Gynecol* 2009; **114**(2 Pt 1): 386-97.

71. National Institute for Health and Clinical Excellence (NICE). Inducting labour. Clinical Guideline 70. [Internet]. 23 July 2008. Available from https://www.nice.org.uk/guidance/cg70/resources/inducing-labour-pdf-975621704389

72. ACOG. Practice bulletin No. 146: Management of late-term and postterm pregnancies. *Obstet Gynecol* 2014; **124**(2 Pt 1): 390-6.

73. ACOG. Macrosomia: ACOG Practice Bulletin Summary, Number 216. *Obstet Gynecol* 2020; **135**(1): 246-8.

74. ACOG. Practice Bulletin No. 188: Prelabor Rupture of Membranes. *Obstet Gynecol* 2018; **131**(1): e1-e14.

75. Vogel JP, Gulmezoglu AM, Hofmeyr GJ, Temmerman M. Global perspectives on elective induction of labor. *Clin Obstet Gynecol* 2014; **57**(2): 331-42.

76. Mozurkewich EL, Chilimigras JL, Berman DR, et al. Methods of induction of labour: a systematic review. *BMC Pregnancy Childbirth* 2011; **11**: 84.

77. Goetzl L. Methods of cervical ripening and labor induction: pharmacologic. *Clin Obstet Gynecol* 2014; **57**(2): 377-90.

Sciscione AC. Methods of cervical ripening and labor induction: mechanical.
 Clin Obstet Gynecol 2014; **57**(2): 369-76.

79. Kelly AJ, Kavanagh J, Thomas J. Castor oil, bath and/or enema for cervical priming and induction of labour. *Cochrane Database Syst Rev* 2013; (7): CD003099.

80. Alfirevic Z, Keeney E, Dowswell T, et al. Labour induction with prostaglandins: a systematic review and network meta-analysis. *BMJ* 2015; **350**: h217.

81. Alfirevic Z, Keeney E, Dowswell T, et al. Methods to induce labour: a systematic review, network meta-analysis and cost-effectiveness analysis. *BJOG* 2016; **123**(9): 1462-70.

82. Cromi A, Ghezzi F, Tomera S, Uccella S, Lischetti B, Bolis PF. Cervical ripening with the Foley catheter. *Int J Gynaecol Obstet* 2007; **97**(2): 105-9.

83. Schoen CN, Grant G, Berghella V, Hoffman MK, Sciscione A. Intracervical Foley Catheter With and Without Oxytocin for Labor Induction: A Randomized Controlled Trial. *Obstet Gynecol* 2017; **129**(6): 1046-53.

84. Mackeen AD, Durie DE, Lin M, et al. Foley Plus Oxytocin Compared With Oxytocin for Induction After Membrane Rupture: A Randomized Controlled Trial. *Obstet Gynecol* 2018; **131**(1): 4-11.

85. Bauer AM, Lappen JR, Gecsi KS, Hackney DN. Cervical ripening balloon with and without oxytocin in multiparas: a randomized controlled trial. *Am J Obstet Gynecol* 2018; **219**(3): 294 e1- e6.

Levine LD, Downes KL, Elovitz MA, Parry S, Sammel MD, Srinivas SK.
 Mechanical and Pharmacologic Methods of Labor Induction: A Randomized Controlled
 Trial. *Obstet Gynecol* 2016; **128**(6): 1357-64.

87. Carbone JF, Tuuli MG, Fogertey PJ, Roehl KA, Macones GA. Combination of Foley bulb and vaginal misoprostol compared with vaginal misoprostol alone for cervical ripening and labor induction: a randomized controlled trial. *Obstet Gynecol* 2013; **121**(2 Pt 1): 247-52.

88. Ornat L, Alonso-Ventura V, Bueno-Notivol J, et al. Misoprostol combined with cervical single or double balloon catheters versus misoprostol alone for labor induction of singleton pregnancies: a meta-analysis of randomized trials. *J Matern Fetal Neonatal Med* 2019: 1-16.

89. Finucane EM, Murphy DJ, Biesty LM, et al. Membrane sweeping for induction of labour. *Cochrane Database Syst Rev* 2020; **2**(2): CD000451.

90. Smith CA, Armour M, Dahlen HG. Acupuncture or acupressure for induction of labour. *Cochrane Database Syst Rev* 2017; **10**: CD002962.

91. DeMaria AL, Sundstrom B, Moxley GE, Banks K, Bishop A, Rathbun L. Castor oil as a natural alternative to labor induction: A retrospective descriptive study. *Women Birth* 2018; **31**(2): e99-e104.

92. Segal S, Gemer O, Zohav E, Siani M, Sassoon E. Evaluation of breast stimulation for induction of labor in women with a prior cesarean section and in grandmultiparas. *Acta Obstet Gynecol Scand* 1995; **74**(1): 40-1.

93. Smith CA. Homoeopathy for induction of labour. *Cochrane Database Syst Rev* 2001; (4): CD003399.

94. Nishi D, Shirakawa MN, Ota E, Hanada N, Mori R. Hypnosis for induction of labour. *Cochrane Database Syst Rev* 2014; (8): CD010852.

95. Kavanagh J, Kelly AJ, Thomas J. Sexual intercourse for cervical ripening and induction of labour. *Cochrane Database Syst Rev* 2001; (2): CD003093.

96. Heffner LJ, Elkin E, Fretts RC. Impact of labor induction, gestational age, and maternal age on cesarean delivery rates. *Obstet Gynecol* 2003; **102**(2): 287-93.

97. Ehrenthal DB, Jiang X, Strobino DM. Labor induction and the risk of a cesarean delivery among nulliparous women at term. *Obstet Gynecol* 2010; **116**(1): 35-42.

98. Seyb ST, Berka RJ, Socol ML, Dooley SL. Risk of cesarean delivery with elective induction of labor at term in nulliparous women. *Obstet Gynecol* 1999; **94**(4): 600-7.

99. Cammu H, Martens G, Ruyssinck G, Amy JJ. Outcome after elective labor induction in nulliparous women: a matched cohort study. *Am J Obstet Gynecol* 2002; **186**(2): 240-4.

100. Johnson DP, Davis NR, Brown AJ. Risk of cesarean delivery after induction at term in nulliparous women with an unfavorable cervix. *Am J Obstet Gynecol* 2003; **188**(6): 1565-9; discussion 9-72.

101. Caughey AB, Sundaram V, Kaimal AJ, et al. Systematic review: elective induction of labor versus expectant management of pregnancy. *Ann Intern Med* 2009; **151**(4): 252-63, W53-63.

102. Mishanina E, Rogozinska E, Thatthi T, Uddin-Khan R, Khan KS, Meads C. Use of labour induction and risk of cesarean delivery: a systematic review and meta-analysis. *CMAJ* 2014; **186**(9): 665-73.

103. Wood S, Cooper S, Ross S. Does induction of labour increase the risk of caesarean section? A systematic review and meta-analysis of trials in women with intact membranes. *BJOG* 2014; **121**(6): 674-85; discussion 85.

104. Middleton P, Shepherd E, Crowther CA. Induction of labour for improving birth outcomes for women at or beyond term. *Cochrane Database Syst Rev* 2018; **5**(5): CD004945.

105. Saccone G, Della Corte L, Maruotti GM, et al. Induction of labor at full-term in pregnant women with uncomplicated singleton pregnancy: A systematic review and meta-analysis of randomized trials. *Acta Obstet Gynecol Scand* 2019; **98**(8): 958-66.

106. Stock SJ, Ferguson E, Duffy A, Ford I, Chalmers J, Norman JE. Outcomes of elective induction of labour compared with expectant management: population based study. *BMJ* 2012; **344**: e2838.

107. Cheng YW, Kaimal AJ, Snowden JM, Nicholson JM, Caughey AB. Induction of labor compared to expectant management in low-risk women and associated perinatal outcomes. *Am J Obstet Gynecol* 2012; **207**(6): 502 e1-8.

108. Muglu J, Rather H, Arroyo-Manzano D, et al. Risks of stillbirth and neonatal death with advancing gestation at term: A systematic review and meta-analysis of cohort studies of 15 million pregnancies. *PLoS Med* 2019; **16**(7): e1002838.

109. ACOG Committee Opinion No. 765: Avoidance of Nonmedically Indicated
Early-Term Deliveries and Associated Neonatal Morbidities. *Obstet Gynecol* 2019;
133(2): e156-e63.

Werner EF, Schlichting LE, Grobman WA, Viner-Brown S, Clark M, Vivier PM.
Association of Term Labor Induction vs Expectant Management With Child Academic
Outcomes. *JAMA Netw Open* 2020; **3**(4): e202503.

111. MacKay DF, Smith GC, Dobbie R, Pell JP. Gestational age at delivery and special educational need: retrospective cohort study of 407,503 school children. *PLoS Med* 2010;
7(6): e1000289.

112. Betran AP, Ye J, Moller AB, Zhang J, Gulmezoglu AM, Torloni MR. The Increasing Trend in Caesarean Section Rates: Global, Regional and National Estimates: 1990-2014. *PLoS One* 2016; **11**(2): e0148343.

113. Betran AP, Torloni MR, Zhang JJ, Gulmezoglu AM, W. H. O. Working Group on Caesarean Section. WHO Statement on Caesarean Section Rates. *BJOG* 2016; **123**(5): 667-70.

114. World Health Organisation. Appropriate technology for birth. *Lancet* 1985;2(8452): 436-7.

115. Boatin AA, Schlotheuber A, Betran AP, et al. Within country inequalities in caesarean section rates: observational study of 72 low and middle income countries. *BMJ* 2018; **360**: k55.

116. Central Statistical Agency - CSA/Ethiopia, ICF. Ethiopia Demographic and Health Sruvey 2016. Addis Ababa, Ethiopia: CSA and ICF, 2017.

117. Hoxha I, Syrogiannouli L, Luta X, et al. Caesarean sections and for-profit status of hospitals: systematic review and meta-analysis. *BMJ Open* 2017; **7**(2): e013670.

118. Mylonas I, Friese K. Indications for and Risks of Elective Cesarean Section.*Dtsch Arztebl Int* 2015; **112**(29-30): 489-95.

119. Betran AP, Temmerman M, Kingdon C, et al. Interventions to reduce unnecessary caesarean sections in healthy women and babies. *Lancet* 2018; **392**(10155): 1358-68.

120. Buyukbayrak EE, Kaymaz O, Kars B, et al. Caesarean delivery or vaginal birth:
preference of Turkish pregnant women and influencing factors. *J Obstet Gynaecol* 2010; **30**(2): 155-8.

121. Schantz C, Sim KL, Petit V, Rany H, Goyet S. Factors associated with caesarean sections in Phnom Penh, Cambodia. *Reprod Health Matters* 2016; **24**(48): 111-21.

122. Jena AB, Schoemaker L, Bhattacharya J, Seabury SA. Physician spending and subsequent risk of malpractice claims: observational study. *BMJ* 2015; **351**: h5516.

123. Ramires de Jesus G, Ramires de Jesus N, Peixoto-Filho FM, Lobato G. Caesarean rates in Brazil: what is involved? *BJOG* 2015; **122**(5): 606-9.
124. Litorp H, Mgaya A, Mbekenga CK, Kidanto HL, Johnsdotter S, Essen B. Fear, blame and transparency: Obstetric caregivers' rationales for high caesarean section rates in a low-resource setting. *Soc Sci Med* 2015; **143**: 232-40.

125. Long Q, Kingdon C, Yang F, et al. Prevalence of and reasons for women's, family members', and health professionals' preferences for cesarean section in China: A mixed-methods systematic review. *PLoS Med* 2018; **15**(10): e1002672.

126. Sobhy S, Arroyo-Manzano D, Murugesu N, et al. Maternal and perinatal mortality and complications associated with caesarean section in low-income and middle-income countries: a systematic review and meta-analysis. *Lancet* 2019; **393**(10184): 1973-82.

127. Torjesen I. Caesarean section is highly risky for mothers and babies in low and middle income countries. *BMJ* 2019; **364**: 11499.

128. Bishop D, Dyer RA, Maswime S, et al. Maternal and neonatal outcomes after caesarean delivery in the African Surgical Outcomes Study: a 7-day prospective observational cohort study. *Lancet Glob Health* 2019; **7**(4): e513-e22.

129. Moroz LA, Wright JD, Ananth CV, Friedman AM. Hospital variation in maternal complications following caesarean delivery in the United States: 2006-2012. *BJOG* 2016; **123**(7): 1115-20.

130. Abbott TEF, Fowler AJ, Dobbs TD, Harrison EM, Gillies MA, Pearse RM. Frequency of surgical treatment and related hospital procedures in the UK: a national ecological study using hospital episode statistics. *Br J Anaesth* 2017; **119**(2): 249-57.

131. Souza JP, Gulmezoglu A, Lumbiganon P, et al. Caesarean section without medical indications is associated with an increased risk of adverse short-term maternal outcomes: the 2004-2008 WHO Global Survey on Maternal and Perinatal Health. *BMC Med* 2010; **8**: 71.

132. Miller S, Abalos E, Chamillard M, et al. Beyond too little, too late and too much, too soon: a pathway towards evidence-based, respectful maternity care worldwide. *Lancet* 2016; **388**(10056): 2176-92.

133. Winberg J. Mother and newborn baby: mutual regulation of physiology and behavior--a selective review. *Dev Psychobiol* 2005; **47**(3): 217-29.

134. Righard L, Alade MO. Effect of delivery room routines on success of first breastfeed. *Lancet* 1990; **336**(8723): 1105-7.

135. Neovita Study Group. Timing of initiation, patterns of breastfeeding, and infant survival: prospective analysis of pooled data from three randomised trials. *Lancet Glob Health* 2016; **4**(4): e266-75.

136. Brandtzaeg PE. Current understanding of gastrointestinal immunoregulation and its relation to food allergy. *Ann N Y Acad Sci* 2002; **964**: 13-45.

137. Le Huerou-Luron I, Blat S, Boudry G. Breast- v. formula-feeding: impacts on the digestive tract and immediate and long-term health effects. *Nutr Res Rev* 2010; 23(1): 23-36.

Ballard O, Morrow AL. Human milk composition: nutrients and bioactive factors.*Pediatr Clin North Am* 2013; **60**(1): 49-74.

139. Tully KP, Ball HL. Maternal accounts of their breast-feeding intent and early challenges after caesarean childbirth. *Midwifery* 2014; **30**(6): 712-9.

140. Karlstrom A, Engstrom-Olofsson R, Norbergh KG, Sjoling M, Hildingsson I. Postoperative pain after cesarean birth affects breastfeeding and infant care. *J Obstet Gynecol Neonatal Nurs* 2007; **36**(5): 430-40.

141. Kolas T, Saugstad OD, Daltveit AK, Nilsen ST, Oian P. Planned cesarean versus planned vaginal delivery at term: comparison of newborn infant outcomes. *Am J Obstet Gynecol* 2006; **195**(6): 1538-43.

142. Zanardo V, Svegliado G, Cavallin F, et al. Elective cesarean delivery: does it have a negative effect on breastfeeding? *Birth* 2010; **37**(4): 275-9.

143. Evans KC, Evans RG, Royal R, Esterman AJ, James SL. Effect of caesarean section on breast milk transfer to the normal term newborn over the first week of life. *Arch Dis Child Fetal Neonatal Ed* 2003; **88**(5): F380-2.

144. Hyde MJ, Mostyn A, Modi N, Kemp PR. The health implications of birth by Caesarean section. *Biol Rev Camb Philos Soc* 2012; **87**(1): 229-43.

145. Dewey KG. Maternal and fetal stress are associated with impaired lactogenesis in humans. *J Nutr* 2001; **131**(11): 3012S-5S.

146. Liston FA, Allen VM, O'Connell CM, Jangaard KA. Neonatal outcomes with caesarean delivery at term. *Arch Dis Child Fetal Neonatal Ed* 2008; **93**(3): F176-82.

147. Hobbs AJ, Mannion CA, McDonald SW, Brockway M, Tough SC. The impact of caesarean section on breastfeeding initiation, duration and difficulties in the first four months postpartum. *BMC Pregnancy Childbirth* 2016; **16**(1): 90.

148. McDonald SD, Pullenayegum E, Chapman B, et al. Prevalence and predictors of exclusive breastfeeding at hospital discharge. *Obstet Gynecol* 2012; **119**(6): 1171-9.

149. Perez-Escamilla R, Maulen-Radovan I, Dewey KG. The association between cesarean delivery and breast-feeding outcomes among Mexican women. *Am J Public Health* 1996; **86**(6): 832-6.

150. Ahluwalia IB, Li R, Morrow B. Breastfeeding practices: does method of delivery matter? *Matern Child Health J* 2012; **16 Suppl 2**(2): 231-7.

151. Watt S, Sword W, Sheehan D, et al. The effect of delivery method on breastfeeding initiation from the The Ontario Mother and Infant Study (TOMIS) III. *J Obstet Gynecol Neonatal Nurs* 2012; **41**(6): 728-37.

152. Kapellou O. Effect of caesarean section on brain maturation. *Acta Paediatr* 2011;100(11): 1416-22.

153. Barrett E, Kerr C, Murphy K, et al. The individual-specific and diverse nature of the preterm infant microbiota. *Arch Dis Child Fetal Neonatal Ed* 2013; **98**(4): F334-40.

154. Curran EA, Kenny LC, Dalman C, et al. Birth by caesarean section and school performance in Swedish adolescents- a population-based study. *BMC Pregnancy Childbirth* 2017; **17**(1): 121.

155. Bentley JP, Roberts CL, Bowen JR, Martin AJ, Morris JM, Nassar N. Planned Birth Before 39 Weeks and Child Development: A Population-Based Study. *Pediatrics* 2016; **138**(6).

156. Polidano C, Zhu A, Bornstein JC. The relation between cesarean birth and child cognitive development. *Sci Rep* 2017; **7**(1): 11483.

157. Smithers LG, Mol BW, Wilkinson C, Lynch JW. Implications of caesarean section for children's school achievement: A population-based study. *Aust N Z J Obstet Gynaecol* 2016; **56**(4): 374-80.

158. Apgar V. A proposal for a new method of evaluation of the newborn infant. *Curr Res Anesth Analg* 1953; **32**(4): 260-7.

159. Drage JS, Kennedy C, Schwarz BK. The Apgar Score as an Index of Neonatal Mortality. A Report from the Collaborative Study of Cerebral Palsy. *Obstet Gynecol* 1964; **24**: 222-30.

160. American Academy of Paediatrics Committee on Fetus and Newborn; American College of Obstetricians and Gynecologists Committee on Obstetric Practice. The Apgar Score. *Pediatrics* 2015; **136**(4): 819-22.

161. Apgar V, Holaday DA, James LS, Weisbrot IM, Berrien C. Evaluation of the newborn infant; second report. *J Am Med Assoc* 1958; **168**(15): 1985-8.

162. Torday JS, Nielsen HC. The Molecular Apgar Score: A Key to Unlocking Evolutionary Principles. *Front Pediatr* 2017; **5**: 45.

Bovbjerg ML, Dissanayake MV, Cheyney M, Brown J, Snowden JM. Utility of the 5-Minute Apgar Score as a Research Endpoint. *Am J Epidemiol* 2019; **188**(9): 1695-704.

164. Razaz N, Cnattingius S, Joseph KS. Association between Apgar scores of 7 to 9 and neonatal mortality and morbidity: population based cohort study of term infants in Sweden. *BMJ* 2019; **365**: 11656.

165. Iliodromiti S, Mackay DF, Smith GC, Pell JP, Nelson SM. Apgar score and the risk of cause-specific infant mortality: a population-based cohort study. *Lancet* 2014;
384(9956): 1749-55.

166. Thorngren-Jerneck K, Herbst A. Low 5-minute Apgar score: a population-based register study of 1 million term births. *Obstet Gynecol* 2001; **98**(1): 65-70.

167. Salustiano EM, Campos JA, Ibidi SM, Ruano R, Zugaib M. Low Apgar scores at 5 minutes in a low risk population: maternal and obstetrical factors and postnatal outcome. *Rev Assoc Med Bras (1992)* 2012; **58**(5): 587-93.

168. Razaz N, Boyce WT, Brownell M, et al. Five-minute Apgar score as a marker for developmental vulnerability at 5 years of age. *Arch Dis Child Fetal Neonatal Ed* 2016;
101(2): F114-20.

169. Siddiqui A, Cuttini M, Wood R, et al. Can the Apgar Score be Used for
International Comparisons of Newborn Health? *Paediatr Perinat Epidemiol* 2017; **31**(4):
338-45.

170. Odd DE, Rasmussen F, Gunnell D, Lewis G, Whitelaw A. A cohort study of low
Apgar scores and cognitive outcomes. *Arch Dis Child Fetal Neonatal Ed* 2008; 93(2):
F115-20.

171. Stuart A, Otterblad Olausson P, Kallen K. Apgar scores at 5 minutes after birth in relation to school performance at 16 years of age. *Obstet Gynecol* 2011; **118**(2 Pt 1): 201-8.

172. Ehrenstein V. Association of Apgar scores with death and neurologic disability. *Clin Epidemiol* 2009; **1**: 45-53.

173. Ehrenstein V, Pedersen L, Grijota M, Nielsen GL, Rothman KJ, Sorensen HT. Association of Apgar score at five minutes with long-term neurologic disability and cognitive function in a prevalence study of Danish conscripts. *BMC Pregnancy Childbirth* 2009; **9**: 14.

174. Seidman DS, Paz I, Laor A, Gale R, Stevenson DK, Danon YL. Apgar scores and cognitive performance at 17 years of age. *Obstet Gynecol* 1991; **77**(6): 875-8.

175. Croft, Trevor N., Aileen M. J. Marshall, Courtney K. Allen, et al. Guide to DHS Statistics. Rockville, Maryland, USA: ICF, 2018.

176. Nuske T, Pilkington R, Gialamas A, Chittleborough C, Smithers L, Lynch J. The Early Childhood Data Project. Research Series 2016. Adelaide: School of Public Health, The University of Adelaide, 2016.

177. Stanton CK, Dubourg D, De Brouwere V, Pujades M, Ronsmans C. Reliability of data on caesarean sections in developing countries. *Bull World Health Organ* 2005;
83(6): 449-55.

178. WHO, UNICEF, USAID, AED, UCDAVIS, IFPRI. Indicators for assessing infant and young child feeding practices. Part I: Definitions. Conclusions of a consensus meeting held 6–8 November 2007 in Washington D.C., USA. Geneva: World Health Organization, 2008.

Australian Bureau of Statistics. Schools, Australia, 2018, 'Table 42b Number of
Full-time and Part-time Students, 2006-2018', data cube: Excel spreadsheet, cat. no.
4221.0. 2018.

180. Greenland S. Modeling and variable selection in epidemiologic analysis. *Am J Public Health* 1989; **79**(3): 340-9.

181. Robins JM. Data, design, and background knowledge in etiologic inference.*Epidemiology* 2001; **12**(3): 313-20.

182. Robins JM, Greenland S. The role of model selection in causal inference from nonexperimental data. *Am J Epidemiol* 1986; **123**(3): 392-402.

183. Greenland S, Neutra R. Control of confounding in the assessment of medical technology. *Int J Epidemiol* 1980; **9**(4): 361-7.

184. Glymour MM, Greenland S. Causal diagrams. In: Rothman KJ, Greenland S, Lash TL, eds. Modern Epidemiology, 3rd ed. Philadelphia: Wolters Kluwer Health/Lippincott Williams & Wilkins; 2008: 183-209.

185. Janani L, Mansournia MA, Nourijeylani K, Mahmoodi M, Mohammad K. Statistical Issues in Estimation of Adjusted Risk Ratio in Prospective Studies. *Arch Iran Med* 2015; **18**(10): 713-9.

186. Greenland S. Interpretation and choice of effect measures in epidemiologic analyses. *Am J Epidemiol* 1987; **125**(5): 761-8.

187. Sinclair JC, Bracken MB. Clinically useful measures of effect in binary analyses of randomized trials. *J Clin Epidemiol* 1994; **47**(8): 881-9.

188. Nurminen M. To use or not to use the odds ratio in epidemiologic analyses? *Eur J Epidemiol* 1995; **11**(4): 365-71.

189. Tamhane AR, Westfall AO, Burkholder GA, Cutter GR. Prevalence odds ratio versus prevalence ratio: choice comes with consequences. *Stat Med* 2016; **35**(30): 5730-5.

190. Zou G. A modified poisson regression approach to prospective studies with binary data. *Am J Epidemiol* 2004; **159**(7): 702-6.

191. Glynn AN, Quinn KM. An Introduction to the Augmented Inverse Propensity Weighted Estimator. *Political Analysis* 2017; 18(1): 36-56.

192. Funk MJ, Westreich D, Wiesen C, Sturmer T, Brookhart MA, Davidian M. Doubly robust estimation of causal effects. *Am J Epidemiol* 2011; **173**(7): 761-7.

193. Emsley R, Lunt M, Pickles A, Dunn G. Implementing Double-robust Estimators of Causal Effects. *Stata J* 2018; **8**(3): 334-53.

194. Lipsitch M, Tchetgen Tchetgen E, Cohen T. Negative controls: a tool for detecting confounding and bias in observational studies. *Epidemiology* 2010; 21(3): 383-8.

195. Thaddeus S, Maine D. Too far to walk: maternal mortality in context. *Soc Sci Med* 1994; **38**(8): 1091-110.

196. Ramachandrappa A, Jain L. Elective cesarean section: its impact on neonatal respiratory outcome. *Clin Perinatol* 2008; **35**(2): 373-93, vii.

197. Frey HA. Apgar scores: Is anything less than perfect a cause for concern? *Paediatr Perinat Epidemiol* 2020; **34**(5): 581-2.

198. Hsieh DC, Smithers LG, Black M, et al. Implications of vaginal instrumental delivery for children's school achievement: A population-based linked administrative data study. *Aust N Z J Obstet Gynaecol* 2019; **59**(5): 677-83.

199. Croft T. "Date Editing and Imputation". In Demographic and Health Surveys World Conference Proceedings, II: 1337-1356, Columbia, Maryland: IRD/ORC

Macro.<u>https://dhsprogram.com/publications/publication-DHSG3-DHS-Questionnaires-</u> and-Manuals.cfm. 1991.

200. Beake S, Bick D, Narracott C, Chang YS. Interventions for women who have a caesarean birth to increase uptake and duration of breastfeeding: A systematic review. *Matern Child Nutr* 2017; **13**(4).